Before the FEDERAL COMMUNICATIONS COMMISSION Washington, D.C. 20554

In the matter of		File No. SAT-MOD-20061109-00137 (S2651)
New ICO Satellite Services G.P.)	
		FILED/ACCEPTED
		DEC 18 2006

Federal Communications Commission Office of the Secretary

PETITION TO DENY OF INMARSAT GLOBAL LIMITED

Diane J. Cornell Vice President, Government Affairs INMARSAT, INC. 1100 Wilson Boulevard, Suite 1425 Arlington, VA 22209 Telephone: (703) 647-4767 John P. Janka Jeffrey A. Marks LATHAM & WATKINS LLP 555 Eleventh Street, N.W. Suite 1000 Washington, D.C. 20004 Telephone: (202) 637-2200

Counsel for Inmarsat Global Limited

December 18, 2006

TABLE OF CONTENTS

SUM	MARY	i
I.	ICO'S LICENSING HISTORY	2
II.	ICO'S DELAY WAS ENTIRELY WITHIN ITS CONTROL	5
III.	REQUESTED RELIEF	9
IV.	CONCLUSION	12
EXHI	IBIT A: November 29, 2005 Space Segment Contract Between ICO Satellite Mana LLC and Space Systems/Loral, Inc.	.gement

SUMMARY

In its application for milestone extension, New ICO Satellite Services G.P. ("ICO") asks the Commission to sanction satellite construction and launch delays that are caused by ICO's voluntary decision to change its satellite system design twice in as many years. In 2005, ICO sought Commission consent to abandon its licensed thirteen-satellite, non-geostationary orbit (NGSO) satellite system in favor of a single geostationary orbit (GSO) spacecraft. Even though the license milestone schedule in the Commission's rules called for such a GSO spacecraft to be launched by July 2006, the Commission gave ICO an extra year, and allowed ICO to deploy that single spacecraft on the same time schedule that applied to ICO's thirteen-satellite NGSO system. That final deadline was July 2007.

In granting ICO's request for a last-minute, fundamental change to its system architecture, the Commission relied on (i) ICO's representation that ICO was already building the spacecraft described in its application, and (ii) ICO's assurances that the simple "legacy" GSO spacecraft under construction could be completed on an accelerated schedule, in time for a July 2007 launch. To be safe, the Commission subjected ICO to a total of *fourteen* milestones (including the two that the Commission found ICO had already met), and imposed a series of reporting obligations, all to ensure that the Commission would know in advance about any further problems, and that service to the public would not be further delayed.

Before the ink was dry on that Commission order, ICO amended its spacecraft construction contract. In fact, ICO amended that contract four times in the next six months. In doing so, ICO abandoned the simple spacecraft design on which the Commission had relied in favor of a design that ICO now describes as "first of its kind," and that includes many new technologies, such as a ground-based beam forming ("GBBF") capability. Today, ICO blames technical problems on the spacecraft related to GBBF capabilities for the delays that its

manufacturer is experiencing, and ICO seeks to use those GBBF-related delays as the basis for this extension request.

Commission policy and precedent are clear that ICO's voluntary incorporation of GBBF capabilities and other "late-breaking" design changes does not justify a milestone extension. Commission policy is that milestone extensions must be based on "unforeseeable" circumstances "beyond the licensee's control." If that were not the case, licensees could constantly ask their manufacturers to change the spacecraft design, and use the resulting complications of those changes as the basis for continued milestone extensions. In the meantime, valuable spectrum would remain unused, and service to the public would be delayed.

Inmarsat fully supports the development and deployment of innovative new technologies, as demonstrated by the many sophisticated technological advances that are included in Inmarsat's current fleet of satellites, the Inmarsat-4s. However, in this circumstance, ICO leveraged the simplicity of its original "bent-pipe" design and its promises of an accelerated construction schedule as a means of convincing the Commission to allow ICO to deploy a GSO system on the longer milestone schedule that applied to its NGSO system, and to grant ICO half of the entire 2 GHz band—only to then fundamentally modify its system design once the Commission granted ICO its requested relief. ICO should not be rewarded for its manipulation of the Commission's processes by being granted the regulatory relief it now seeks.

For these reasons, the Commission should deny ICO's milestone extension request. If the Commission nonetheless grants ICO's request, the Commission should clarify that this is the last extension it will grant to ICO. The Commission should also require, prior to granting ICO any ATC authority, that ICO strictly adhere to the ATC gating criteria. In particular, the Commission should make clear that any past or future delays in constructing or

launching ICO's two required spacecraft (an in-orbit spacecraft and a ground spare) will not provide the basis for waiving any ATC gating criteria, including the requirements that ICO construct and have a ground spare available for launch within one year of commencing ATC service, and that ICO actually launch an MSS spacecraft that provides MSS service to the entire United States before providing ATC service. Given ICO's history of delay in constructing and launching its MSS system, the Commission must be especially vigilant in ensuring that MSS service to the public is not delayed any further.

Before the FEDERAL COMMUNICATIONS COMMISSION Washington, D.C. 20554

In the matter of)	File No. SAT-MOD-20061109-00137 (S2651)
New ICO Satellite Services G.P.)	

PETITION TO DENY OF INMARSAT GLOBAL LIMITED

Inmarsat Global Limited ("Inmarsat") petitions the Commission to deny the application of New ICO Satellite Services G.P. ("ICO") for an extension of ICO's milestones to construct and launch its licensed 2 GHz mobile satellite service ("MSS") spacecraft. ICO has failed to demonstrate that the delays ICO faces are due to circumstances beyond its control. Thus, under longstanding Commission policy and precedent, the Commission should deny ICO's extension request. If the Commission nevertheless grants the requested extension, the Commission should condition the grant to ensure that ICO does not (i) avail itself of any further extensions, or (ii) use delay in building or launching its spacecraft as the basis for to seeking relief from the ATC gating criteria.

Inmarsat has standing as an MSS operator with which ICO seeks to compete, as a participant in the recent proceedings in which the Commission increased ICO's 2 GHz spectrum

New ICO Satellite Services G.P., Expedited Request for Milestone Extension, IB File No. SAT-MOD-20061109-00137 (filed Nov. 9, 2006) ("ICO Application"). See Public Notice, Policy Branch Information, Space Station Applications Accepted for Filing, Rep. No. SAT-00402 (Nov. 17, 2006).

assignment by 250 percent,² and as a competing applicant for a portion of the 2 GHz band currently licensed to ICO.³

I. ICO'S LICENSING HISTORY

ICO's current request for a milestone extension represents the latest in a long line of delays in ICO's 2 GHz MSS system implementation. ICO originally promised the Commission a global network of thirteen, non-geostationary orbit (NGSO) spacecraft, with 2 GHz MSS service beginning in 2000.⁴ ICO then went through bankruptcy reorganization, changed ownership, renegotiated with its spacecraft contractor, modified its spacecraft design, and told the Commission in November 2000 that it would need only a few more years to implement its system, until the second quarter of 2003.⁵ 2003 came and went without ICO providing MSS service.

In 2005, ICO was back before the Commission, seeking to modify its licensed MSS system. This time, ICO had abandoned its NGSO network, and sought authority to deploy a much-scaled-back system consisting of a single geostationary orbit (GSO) spacecraft.⁶ Similar

² Use of Returned Spectrum in the 2 GHz Mobile Satellite Service Frequency Bands, 20 FCC Rcd 19696 (2005) ("2 GHz Order").

Inmarsat Global Limited, Petition for Declaratory Ruling to Provide MSS to the United States Using the 2 GHz and Extended Ku Bands, SAT-PPL-20050926-00184 (filed Sept. 26, 2005). When the Commission increased ICO's and TMI/TerreStar's 2 GHz assignments instead of licensing additional operators, the Commission dismissed Inmarsat's request. Inmarsat Global Limited, 20 FCC Rcd 19409 (2005). Inmarsat's petition for reconsideration remains pending. See Consolidated Petition for Reconsideration, Inmarsat Ventures Limited and Inmarsat Global Limited, IB Docket Nos. 05-220, 05-221, et al. (filed Jan. 9, 2006).

⁴ ICO Letter of Intent to Access 2 GHz MSS Frequency Bands at 1990-2025/2165-2200 MHz, SAT-LOI-19970926-00163 (Sept. 26, 1997).

⁵ See Second Amendment to ICO's Letter of Intent to Access 2 GHz MSS Frequency Bands at 1990-2025/2165-2000 MHz, SAT-AMD-20001103-00155, at 7 (Nov. 3, 2000).

⁶ ICO Satellite G.P., 20 FCC Rcd 9796 (2005) ("ICO 2005 Modification Order").

to the facts at hand now, ICO had voluntarily changed its MSS system design, began implementing the new system design, and then pointed to the money it had expended (at its own risk) as justification for Commission relief. Although the license milestone schedule in the Commission's rules called for such a GSO spacecraft to be launched by July 2006, the Commission gave ICO an extra year, until July 2007, to deploy (this was the same final deadline that applied to ICO's abandoned thirteen-satellite NGSO system).

In granting ICO's request for a last-minute, fundamental change to its system architecture, the Commission relied on ICO's representation regarding ICO's modest GSO system design and ICO's progress in building that spacecraft, and determined that a July 2007 launch date was achievable because "the proposed geostationary satellite is of relatively simple bent-pipe design, and most of the components are 'legacy' equipment that has been used or developed for previous projects." ¹⁰

In order to ensure that ICO actually deployed on time, the Commission adopted a substantially more rigorous license milestone schedule than it ever had applied before. ICO was subjected to a total of fourteen milestones: the two it had already met, and twelve more, spaced an average of two months apart over the next two years. This stands in stark contrast to the milestones that the Commission had adopted for GSO spacecraft just two years earlier (four

⁷ ICO 2005 Modification Order, 20 FCC Rcd at 9804, n.39. The Commission rejected ICO's claims that it could not go forward with its NGSO system due to alleged "unforeseeable contractual difficulties," finding that ICO had failed to demonstrate "that the problem was either unforeseeable or beyond ICO's control." Id.

See Application to Modify Letter of Intent Authorization of ICO Satellite Services G.P., File No. Sat-MOD-20050110-0004, at 4-7 (filed Jan. 10, 2005).

 $^{^9}$ $\,$ ICO 2005 Modification Order, 20 FCC Rcd at 9803-9804, \P 25.

¹⁰ *Id.* at 9803, ¶¶ 24, 26.

¹¹ *Id.* at 9804, 9808, ¶¶ 27, 38.

milestones spread over five years).¹² At that time, the Commission rejected as unnecessary proposals for milestones spaced at six-month intervals, finding that under the new, standard milestones, "we will have sufficient assurance throughout the construction stage that the licensee is building its system."¹³

ICO's case, however, warranted a much more rigorous monitoring and milestone enforcement program. In addition to the *dozen* remaining milestones, the Commission required ICO to provide notification if other tasks delineated on Appendix F of its manufacturing agreement were completed more than 14 days late. ¹⁴ ICO not only will miss its final milestone; ICO also has not been able to achieve many of those other scheduled dates. ¹⁵

Seven months later, when the Commission increased ICO's (and TMI/TerreStar's) 2 GHz spectrum assignment by 250 percent, rather than license additional competitors in the band, the Commission did so based on the expectation that ICO would still be

Amendment of the Commission's Space Station Licensing Rules and Policies; Mitigation of Orbital Debris, 18 FCC Rcd 10760, 10827-10829, ¶¶ 173-178 (2003) ("Space Station Licensing Reform"); 25 C.F.R. § 25.164(a) ((1) contract execution; (2) critical design review ("CDR"); (3) begin construction; and (4) launch).

¹³ *Id.* at 10834, ¶¶ 195-196.

Appendix F is not attached to the publicly available version of ICO's January 10, 2005 contract on file with the Commission.

¹⁵ ICO 2005 Modification Order, 20 FCC Rcd at 9804, 08, ¶¶ 27, 39. ICO notified the FCC that at least 5 of these contractually scheduled dates were not achieved. Public flings report "Milestone 26" as 60 days late, "Milestone 28" as 21 days late, "Milestone 29" as 27 days late, the contract milestone to "complete propulsion integration" as 19 days after the contract deadline, and "Milestone 31" as 20 days late. See Letter from Suzanne Hutchings Malloy, ICO, to Marlene H. Dortch, FCC, File No. SAT-MOD-20050110-00004 (filed Nov. 22, 2006) (regarding Milestone 31); Letter from Cheryl A. Tritt, Morrison & Foerster LLP, to Marlene H. Dortch, FCC, File No. SAT-MOD-20050110-00004 (filed Oct. 3, 2006) (regarding Milestone 29); Letter from Suzanne Hutchings Malloy, ICO, to Marlene H. Dortch, FCC, File No. SAT-MOD-20050110-00004 (filed Aug. 23, 2006) (regarding Milestones 26 and 28); Letter from Cheryl A. Tritt, Morrison & Foerster LLP, to Marlene H. Dortch, FCC, File No. SAT-MOD-20050110-00004 (filed Apr. 27, 2006) (regarding completion of propulsion integration).

able to launch in accordance with the existing milestones in ICO's license. ¹⁶ In advocating that ICO should receive the additional spectrum, ICO emphasized that "No other option . . . will ensure the launch of service by 2007." To that end, the Commission cited its understanding that ICO would be able to "offer public safety services more quickly than would be possible if the spectrum were assigned to another party" as one of the public interest benefits that should flow from awarding ICO additional 2 GHz spectrum. ¹⁸ The Commission also reaffirmed the strict application of its milestone policy, emphasizing that "Both ICO's and TMI's spectrum reservations include milestone requirements, under which they will lose their rights to provide service in the United States if they do not construct their satellite systems in a timely manner." ¹⁹

As detailed in the next section, during this same time period, ICO was busy amending its spacecraft construction contract a total of four times, and making technology choices that have given rise to the very delays that now form the basis for its milestone extension request.

II. ICO'S DELAY WAS ENTIRELY WITHIN ITS CONTROL

Commission policy is clear that it grants milestone extensions only in cases of unforeseeable circumstances beyond the licensee's control.²⁰ In particular, the Commission has

¹⁶ 2 GHz Order, 20 FCC Rcd at 19709, 19722, ¶¶ 28, 57.

¹⁷ Comments of ICO Satellite Services G.P., IB Docket No. 05-221, at 3 (filed July 29, 2005).

¹⁸ 2 GHz Order, 20 FCC Rcd at 19709, ¶ 28.

¹⁹ *Id.* at 19722, ¶ 57.

See, e.g., Space Station Licensing Reform, 18 FCC Rcd 10760, 10826, ¶ 170 ("By making the bond payable upon failure to meet any milestone based on circumstances within the licensee's control, we require licensees to commit to construct and launch its satellite system, and so we further strengthen our protections against speculation and warehousing."); id. at 10882, ¶ 333 ("In the First Report and Order in this proceeding, the Commission required satellite licensees to post a bond, payable upon failure to meet a milestone, and without facing circumstances outside the licensee's control that warrant extension of the milestone.").

repeatedly found that technological choices made by a licensee after issuance of its license are within the licensee's control and do not justify a milestone extension.²¹ Moreover, the Commission has *denied* milestone extension requests where the resulting delay is due to such choices.²² The policy rationale underlying those decisions is clear:

Extending milestones on this basis would allow licensees to extend indefinitely their nonperformance by repeated modification of their proposals which could in turn facilitate warehousing of scarce orbital resources or, at a minimum, delay service to the public.²³

In this case, both public harms already exist. While ICO's 20 MHz of spectrum goes unused, other ready, willing and able operators are foreclosed from proceeding with their own 2 GHz MSS systems, the 2 GHz band remains unused for MSS, and service to the public continues to be delayed.

The public record demonstrates that ICO's current need for an extension of time is neither "unforeseeable" nor "due to circumstances beyond ICO's control." As an initial matter, industry participants recognize the current delays for what they always were—inevitable. In recent interview regarding ICO's current request for milestone extension, one leading analyst remarked that he:

wonders whether the delays were anticipated from the start. "When the first contract was written, it was obvious they could never meet that schedule."... "ICO took a while to sign the contract, then did so probably assuming they could probably get an extension. I think they

NetSat 28 Company L.L.C., 19 FCC Rcd 17722, 17726, ¶10 (2004); PanAmSat Licensee Corp., 16 FCC Rcd 11534, 11541, ¶21 (2001); Loral Space & Communications Corporation, 16 FCC Rcd 11044, 11047 ¶7 (2001).

²² See cases cited, supra, note 21.

NetSat 28 Company L.L.C., 19 FCC Rcd at 17726 ¶ 10; see also PanAmSat Licensee Corp., 16 FCC Rcd 11534, 11541, ¶ 21 (2001) ("We do not extend milestones merely to allow a licensee to incorporate a new technology into its satellite design. Otherwise, we would create a loophole in our milestone policy, allowing licensees to extend their milestones indefinitely by filing modification applications").

knew themselves all along and that the FCC would be hard-pressed to cancel the schedule."²⁴

ICO now admits that the delay in construction is caused by a voluntary technology choice—use of ground-based beam forming ("GBBF") and other new technology in what ICO now characterizes as a "first of its kind" satellite. In fact, far from the relatively simple satellite described in ICO's 2005 GSO application, ICO now describes its satellite as "the largest satellite that SS/L has constructed to date" and "the first two-way implementation of the innovative [GBBF] technology by any manufacturer worldwide," "using designs never before utilized in the satellite industry." Moreover, ICO's application highlights the schedule impact of making this late decision to incorporate GBBF technology:

The sophistication of this GBBF technology demands that the component parts of the ICO G1 satellite, particularly the oscillator and waveguide components, meet more demanding technical specifications than on previous satellite systems.²⁶

ICO's satellite construction contract is illuminating, even though the publicly available version is heavily redacted. A comparison of (i) ICO's satellite construction contract submitted to the Commission on May 9, 2005 (two weeks before the Commission granted ICO relief), with (ii) the November 29, 2005 version submitted to the SEC,²⁷ demonstrates that ICO chose to introduce substantial new complexities into its system design (including GBBF) *right* after the Commission granted ICO authority to implement a simple GSO design. For example,

²⁴ ICO Satellite Delay Raises Questions About ATC Market, SATELLITENEWS, Nov. 20, 2006 at 1-2 (quoting Roger Rusch, President, Telastra Inc.).

²⁵ ICO Extension Request, Exhibit 1 at 1-2, 6.

²⁶ Id. at Exhibit 1 at 1.

See Space Segment Contract Between ICO Satellite Management LLC and Space Systems/Loral, Inc., as amended and restated Nov. 29, 2005 (attached as Exhibit 10.1 to Registration of Securities Pursuant to Section 12(g) on Form 10, Securities and Exchange Commission File No. 000-52006 (filed May 15, 2006)). This later version is attached as Exhibit A hereto.

GBBF technology is not even mentioned once in the terms and conditions of the contract ICO filed with the Commission two weeks before receiving its GSO license, but GBBF references now permeate the amended construction contract. Indeed, the version of ICO's construction contract dated November 29, 2005 uses the term "GBBF" one hundred eleven times in the publicly available terms and conditions alone. Moreover, GBBF is such a significant addition that the current contract contains a separate Attachment C entitled "GBBF-Unique Components and Functions on Board the Satellite." 29

Curiously, although ICO amended its satellite construction contract at least *four times* in the six months following Commission's May 24, 2005 order authorizing ICO's GSO modification, there is no indication that any of those contract amendments was filed with the Commission. ICO clearly had an obligation to do so.

As detailed above, the terms of ICO's contract were material to the Commission's May 24, 2005 decision. Section 1.65 of the Commission's rules requires all applicants to amend their pending applications when the information previously furnished to the Commission is no longer accurate *or complete*. ICO's application was "pending" until the Bureau grant no longer was subject to reconsideration by the full Commission or review by a court³⁰ (*i.e.*, 40 days from release of the order).³¹ Thus, the Commission's rules required that ICO make appropriate filings to ensure the accuracy and completeness of the information on file through that 40-day period, or at least until early July 2005. Such filings should include any amendments to the construction

²⁸ See id.

²⁹ See id. at iv. Unfortunately, that attachment is not publicly available at the SEC.

³⁰ 47 C.F.R. § 1.65(a).

³¹ Id. § 1.117(a) (permitting Commission review on its own motion and within 40 days after public notice any action taken on delegated authority).

contract on which the Commission relied. ICO amended its construction contract during that 40-day time period, on June 23, 2005, and a mere 30 days after the Commission grant.

There are additional reasons ICO should have informed the Commission of its changed plans. ICO certainly was engaged in negotiations with its manufacturer well before then, even possibly before the Commission approved the GSO configuration. Moreover, ICO amended its contract a total of four times while the Commission was considering ICO's request for a license modification to increase its 2 GHz spectrum assignment.

Inmarsat fully supports the development and deployment of innovative new technologies, as demonstrated by the many sophisticated technological advances that are included in Inmarsat's current fleet of satellites, the Inmarsat-4s. However, in this circumstance, ICO leveraged the simplicity of its original "bent-pipe" design and its promises of an accelerated construction schedule as a means of convincing the Commission to allow ICO to deploy a GSO system on the longer milestone schedule that applied to its NGSO system, and to grant ICO half of the entire 2 GHz band—only to then fundamentally modify its system design once the Commission granted ICO its requested relief. ICO should not be rewarded for its manipulation of the Commission's processes by being granted the regulatory relief it now seeks.

III. REQUESTED RELIEF

ICO's history of delay, the facts underlying ICO's decision to fundamentally modify its spacecraft design immediately after the Commission authorized its current system design, the expectations in last year's 2 GHz proceeding that ICO would launch on time, and other questions surrounding ICO's current extension, all warrant that the Commission take a "hard look" at ICO's milestone extension request. Particularly in light of the scrutiny the Commission previously has given ICO's milestone compliance and the other scheduled dates in ICO's manufacturing agreement, the Commission should require that ICO: (i) file complete

copies of each and every amendment to its spacecraft construction contract; (ii) explain when ICO began to plan to change its spacecraft (which appears to have been while ICO's GSO modification application still was pending at the Commission); and (iii) demonstrate that a technology choice, or another circumstance within ICO's control, is not the cause of ICO's missing its upcoming milestone. A review of each ICO amendment to its construction contract executed after January 10, 2005 is essential to making that determination.

In the absence of a clear demonstration that ICO's milestone non-compliance is due to circumstances beyond its control, the Commission should deny ICO's request for a milestone extension. If the Commission nonetheless grants ICO's request, the Commission should take steps to ensure the continued effectiveness of the Commission's milestone and ATC rules and policies:

First, the Commission should clarify that ICO will receive no further extensions of its license milestones if ICO fails to launch its spacecraft by November 30, 2007 and have it operational by December 31, 2007. As detailed above, ICO is far behind the schedule that it contemplated, or that Commission rules provide for other similarly-situated licensees. The 2 GHz MSS band continues to lie fallow, and service to the public continues to be delayed in the meantime.

Second, the Commission should make clear that neither ICO's previous delays nor any future delays in building or launching its spacecraft would provide the basis for waiving or modifying any ATC gating criteria in ICO's case. With respect to ATC, Commission rules provide that ATC applications will not be granted prior to an applicant satisfying each of the ATC gating criteria.³² Those gating criteria are important as a policy matter, because they help

³² See id. § 25.149(b).

to ensure that ATC remains ancillary to MSS. One of the gating criteria requires that ICO construct and have a ground spare at the ready (available for launch within one year of commencing operations). If ICO is to become operational by the December 31, 2007 milestone it proposes, and if it is to start providing ATC shortly thereafter, ICO will need a ground spare at the ready by the end of 2008. There is no indication that such a spacecraft can be built in the next two years, or that ICO has a contract in place to do so. Given the delays to date in implementing ICO's primary spacecraft, it is especially important that the Commission be vigilant in enforcing this gating criterion. As a condition of any ATC authority that ICO might apply for in the future, ICO should be required to demonstrate that it will have its ground spare constructed, and available for launch, within one year of commencing service, before being allowed to offer ATC service.

Another relevant gating criterion requires that ICO actually be capable of providing MSS to the entire United States prior to offering ATC.³⁴ ICO has provided no firm assurances that it actually will be able to launch its current spacecraft at any time in the near future.³⁵ The Commission thus should be vigilant in enforcing this requirement. Continued progress on building this spacecraft, alone, would do nothing to ensure that MSS service actually will be provided. Certainly, simply building a satellite could not justify a waiver of this "MSS service" gating criterion.

³³ *Id.* § 25.149(b)(2)(ii).

³⁴ *Id.* § 25.149(b)(1).

For example, the letter provided by Lockheed Martin, Commercial Launch Services provides no indication of when it may be able to launch the ICO spacecraft, assuming ICO is able to have the satellite completed by fall of 2007. See ICO Application Narrative at Attachment B.

IV. CONCLUSION

For the foregoing reasons the Commission should deny ICO's request for a milestone extension. If the Commission nevertheless grants ICO's request, the Commission should condition the grant as provided above, to ensure that 2 GHz MSS service to the public will not be further delayed.

Diane J. Cornell Vice President, Government Affairs INMARSAT, INC. 1100 Wilson Boulevard, Suite 1425 Arlington, VA 22209 Telephone: (703) 647-4767 Respectfully submitted,

John PV Janka Jeffrey A. Marks

LATHAM & WATKINS LLP 555 Eleventh Street, N.W.

Suite 1000

Washington, D.C. 20004 Telephone: (202) 637-2200

Counsel for Inmarsat Global Limited

December 18, 2006

VERIFICATION

I, Diane J. Cornell, hereby certify under penalty of perjury that I am Vice President, Government Affairs of Immarsat, Inc., and that the facts presented in the foregoing Petition to Deny that are not a matter of public record are true and correct to the best of my knowledge, information and belief.

Rν

Diane J. Cornell

Vice President, Government Affairs

INMARSAT, INC.

1100 Wilson Blvd, Suite 1425

Arlington, VA 22209

Telephone: (703) 647-4767

Executed on December 18, 2006.

EXHIBIT A

November 29, 2005 Space Segment Contract Between ICO Satellite Management LLC and Space Systems/Loral, Inc.

ICO Global Communications (Holdings) LTD

PLAZA AMERICA TOWER 11700 PLAZA AMERICA DRIVE, SUITE 1010 RESTON, VA 20190 (703) 964-14

EX-10.1

MATERIAL CONTRACTS 10-12G Filed on 05/15/2006 File Number 000-52006



CONFIDENTIAL TREATMENT OF CERTAIN DESIGNATED PORTIONS OF THIS EXHIBIT HAS BEEN REQUESTED BY ICO GLOBAL COMMUNICATIONS (HOLDINGS) LIMITED. SUCH CONFIDENTIAL PORTIONS HAVE BEEN OMITTED, AS INDICATED BY A[*] IN THE TEXT, AND SUBMITTED TO THE COMMISSION.

EXECUTION ORIGINAL

SPACE SEGMENT CONTRACT

Between

ICO Satellite Management LLC

And

Space Systems/Loral, Inc.

The attached Contract and the information contained therein are confidential and proprietary to ICO Satellite Management LLC and Space Systems/Loral, Inc. and shall not be published or disclosed to any third party except as permitted by the terms and conditions of this Contract.

Use or disclosure of the data contained on this page is subject to the restrictions set forth in this Contract,

TABLE OF CONTENTS

ADCE	Page
Table of Contents	 .
Preamble	
Recitals	2
Article 1- Definitions	. 3
Article 2- Scope of Work	20
Article 3- Deliverable Items and Delivery Schedule	21
Article 4 - Price	23
Article 5 - Payments	27
Article 6 - Purchaser-Furnished Items	37
Article 7 - Compliance with U.S. Laws and Directives	40
Article 8 - Access to Work-in-Process	43
Article 9 - Satellite Pre-Shipment Review (SPSR) and Spacecraft Launch Readiness Review (SLRR)	46
Article 10 - Satellite and Space Segment Acceptance	51
Article 11 - Acceptance Inspection for Deliverable Items other than Satellite or Space Segment	59
Article 12 - Transfer of Title and Risk of Loss	65
Article 13 - Satellite Orbital Performance Incentives and Space Segment Orbital Performance Incentives	67
Article 14 - Corrective Measures in Satellite	79
Article 15 - Warranty	80
Article 16 - Changes and Stop Work	92
ii	72

							100		100		96
	Article 18 - Purchaser Delay of \	Vork.						÷.			98
	Article 19 - Intellectual Property	Indemnity					* a	* .			99
	Article 20 - Indemnity for Person	al Injury and Property Dan	nage	·							101
	Article 21 - Termination for Con					-					
	Article 22 - Liquidated Damages	,									105
	Article 23 – Termination for Defa								*.		108
	Article 24 – Options	mit and Excessive Poice M	njeure	•		-		. •			112
				F							121
	Article 25 - Dispute Resolution	1111111			, J						130
٠	Article 26 - Inter-Party Waiver o	f Liability for a Launch	**	•							134
	Article 27 - Major Subcontracts		111 111								136
	Article 28 - Contractor Insurance	Requirements									137
	Article 29 - Personnel and Key Pe	ersonnel					•				141
	Article 30 - Limitation of Liability	y									143
	Article 31 - Disclosure and Handi	ing of Proprietary Informat	ion								145
	Article 32- Intellectual Property R		٠								
	Article 33- Public Release of Info	rmation									148
	Article 34 – Notices					• •					157
	Article 35- Risk Management Ser	uina.	•							٠.	158
	Article 36- Order of Precedence	vices									160
	•										162
	Article 37- General						•				163
	Attachment A	Form of Invoice									
	nnex 1 to Attachment A				•						
S	chedule I to Annex I to Attachment										
	Attachment B	Key Personnel									

[*] CONFIDENTIAL TREATMENT REQUESTED BY ICO GLOBAL COMMUNICATIONS (HOLDINGS) LIMITED.

Attachment C

GBBF-Unique Components and Functions on Board the Satellite

Attachment D

[*]

Attachment E

Major Subcontracts List

iv

PREAMBLE

This Space Segment Contract (the "Contract") is executed as of November 29, 2005 ("Execution Date") and deemed entered into as of January 10, 2005 (the "Effective Date of Contract" or "EDC") between ICO Satellite Management LLC, a limited liability company organized and existing under the laws of the State of Delaware, having an office and place of business at 2300 Carillon Point, Kirkland, Washington, 98033 (hereinafter referred to as "Purchaser") and Space Systems/Loral, Inc., a corporation organized and existing under the laws of the State of Delaware, having an office and place of business at 3825 Fabian Way, Palo Alto, California 94303 (hereinafter referred to as "Contractor"). This Contract restates and amends the Satellite Contract entered into by Purchaser and Contractor as of January 10, 2005, as amended by Amendment One (1) entered into as of June 23, 2005, Amendment Two (2) entered into as of August 2, 2005, and Amendment Three (3) entered into as of September 23, 2005 (the "Original Satellite Contract").

RECITALS

WHEREAS, Purchaser desires to procure one (1) communication Satellite, and an option for an additional one (1) communication satellite, to be accepted on—orbit and integrated with a Ground—Based Beam Forming Subsystem to form a Space Segment, launch support services, risk management insurance procurement support, training services and other items and services to the extent and subject to the terms and conditions set forth herein, and

WHEREAS, Contractor is willing to furnish such Satellite and such option satellite, integrated with a Ground-Based Beam Forming Subsystem to form a Space Segment, launch support services, risk management insurance procurement support, training services and other items and services, to the extent of and subject to the terms and conditions set forth herein, in consideration of the Fixed Firm Price and other valid consideration,

NOW, THEREFORE, in consideration of the mutual covenants and agreements contained herein and intending to be legally bound, the Parties agree as follows:

ARTICLE 1- DEFINITIONS

Capitalized terms used and not otherwise defined herein shall have the following n	eanings:
--	----------

- 1.1 "Acceptance" or "Accepted" (i) with respect to the Satellite and the Space Segment shall be as provided for in Article 10, and (ii) with respect to any Deliverable Item other than the Satellite and the Space Segment (including, without limitation, the GBBF Subsystem and the DSS) shall be as provided for in Article 11.
- 1.2 "Actual Costs" shall mean, without duplication, Contractor's direct labor (and allocated fringe, overhead and general and administrative costs), materials (including subcontracts), and other direct costs (and allocated general and administrative costs), all in accordance with Contractor's generally accepted accounting practices.
- 1.3 "Affiliate" means, with respect to an entity, any other entity Controlling or Controlled by or under common Control with such entity.
- 1.4 "Applicable Space Segment Rate" has the meaning set forth in Article 13.1 hereof.
- 1.5 "Asserting Party" has the meaning set forth in Article 32.2.3.
- 1.6 "Reserved"
- 1.7 "Attachment(s)" means any and all attachment(s) that are attached hereto or to any Exhibit and incorporated herein or therein, as may be amended from time to time in accordance with the terms hereof.
- 1.8 "Business Day" means any day other than the following: a Saturday, Sunday, and any other day on which national banks are authorized to be closed in New York City, New York.
- 1.9 "Change Request" has the meaning set forth in Article 16.1.1.

	CO GLOBAL COMMUNICATIONS (HO	

- 1.10 "Competitor of Contractor" or words to that affect means any Person or any Affiliate thereof engaged in the manufacture of commercial telecommunication satellites.
- 1.11 "Conforming Space Segment" [*]
- 1.12 "Contract" means the terms and conditions (Preamble, Recitals and Articles) of this executed contract, its Exhibits and its Attachment(s) as set forth in Articles 2.1 and 36, as the same may be amended from time to time in accordance with the terms
- 1.13 "Contractor" has the meaning set forth in the Preamble and any successor or assignee permitted hereunder.
- 1.14 "Contractor Background Intellectual Property" means Intellectual Property first made, developed, or created by Contractor (or by others acting on behalf of Contractor including any subcontractor) prior to the negotiation or performance of the Original Satellite Contract or otherwise first made, developed, or created by Contractor (or by others acting on behalf of Contractor including any subcontractor) other than in performance of or pursuant to this Contract or the Original Satellite Contract and necessary to the use of any Deliverable Item.
- 1.15 "Contractor Notice of Default" has the meaning set forth in Article 23.2.1.
- 1.16 "Contractor Space Segment Background Intellectual Property" means Intellectual Property directly relating to (i) ground-based beam forming subsystems, (ii) the interaction between a ground-based beam forming subsystem and a satellite, or (iii) space segment-unique components of a satellite, first made, developed, or created by Contractor (or by others acting on

behalf of Contractor including any subcontractor) prior to the negotiation or performance of the Original Satellite Contract or otherwise first made, developed, or created by Contractor (or by others acting on behalf of Contractor including any subcontractor) other than in performance of or pursuant to this Contract or the Original Satellite Contract and necessary to the use of any Deliverable Item relating to the Space Segment (other than the Satellite) to be Delivered by Contractor hereunder. Anything in the foregoing to the contrary notwithstanding, "Contractor Space Segment Background Intellectual Property" shall not include any Intellectual Property in or relating specifically to a satellite other than space—segment unique components of a satellite.

- 1.17 "Contractor Space Segment Foreground Intellectual Property" means Intellectual Property first made, developed or created by Contractor (or by others acting on behalf of Contractor including any subcontractor) in the performance of Contractor's obligations with respect to the GBBF Subsystem under the Original Satellite Contract and/or the Space Segment under this Contract, including with respect to the Space Segment—unique components of the Satellite required for Satellite operation with the GBBF Subsystem, as set forth in Attachment C. "Contractor Space Segment Foreground Intellectual Property" shall not include any Intellectual Property in or relating specifically to the Satellite except as set forth in the immediately preceding sentence.
- 1.18 "Contractor Space Segment Intellectual Property" means, collectively, the Contractor Space Segment Background Intellectual Property and the Contractor Space Segment Foreground Intellectual Property.
- 1.19 "Control" and its derivatives mean, with respect to a Person, the legal, beneficial, or equitable ownership, directly or indirectly of more than fifty percent (50%) of the capital stock (or other ownership interest if not a corporation) of such Person ordinarily having voting rights or the power to direct the management policies of such Person, whether through the ownership of voting stock, by contract, or otherwise.

1.20	"Deadline Date" shall have the applicable meaning	set forth in Arti	cle 5.2.3A.				
1.01	(F) Brought Patell						
1.21 Statem	"Deliverable Data" means the data and documentated ent of Work and Exhibit G. Space Segment Statement		be delivered	to Purcha	iser as spec	ified in Exhi	bit A,
			•		•		
							•
1.22	"Deliverable Item" means any of the items or service			ay be ame	ended from	time to time	in
accord	ance with the terms hereof, and, collectively, the "De	liverable items."				-	
1.23	"Delivery" has the meaning set forth it in Article 3.2	! .					
		•				•	•
						· · · · · ·	
1.24	"Dispute" has the meaning set forth in Article 26.						
			•				
1.25	"DSS" means Contractor's Dynamic Satellite Simul				sary to the _l	proper funct	ion
uiereoi	, as more specifically described in Exhibit J, ICO DSS	SOW and Fund	tional Requ	irements.			
		•					
1.26	"Effective Date of Contract" or "EDC" means the ef	fective date of th	nis Contract	as specifi	ied in the Pr	eamble, beli	ng Januar
10, 200	5.						
1.27	"Equitably Adjusted" or "Equitable Adjustment" me	ans, with resne	ct to a partic	rular even	t nocecitat	ing an adiwe	tment in
the Cor	itract where an Equitable Adjustment is provided for.	a reasonable o	ood faith ad	inetment t	o the terms	of this Cont	ract
SUCH D	ng price, schedules, payment schedules, delivery dat articular event. The adjustments shall either be agree	d to by the Parti	es in accord	lance with	the terms (of this Contr	act -
(includi 25. [*]	ing the incorporation in a Contract Amendment in acc	cordance with A	rticle 37.5) (or as deter	mined in ac	cordance wi	th Article
1.28	"Execution Date" means the execution date set fort	h in the Preamb	le.			٠	
		6 .			•		
<u></u>	The state of the s						
				-			

[*] CONFIDENTIAL TREATMENT REQUESTED BY ICO GLOBAL COMMUNICATIONS (HOLDINGS) LIMITED.

[*]	CONFIDENTIAL TREA	ATMENT REQUESTED BY	ICO GLOBAL (COMMUNICATIONS (HOLDINGS) LIMITED	Į

1.29

	ed and amended from time to time oct in accordance with Article 24.	ie in accordance witi	n the terms ner	eor, without limite	ition, Exhibits may	De added to this
1.30	"FCC" means the Federal Com	munications Commi	ssion or any si	accessor agency	or governmental a	uthority.
		the second				
1.31	"FCC Event(s)" has the meaning	ng set forth in Article	6.2.3A.			
4 30	P0.7				•	
1.32	[*]		•			
		e e e e e e e e e e e e e e e e e e e				the state of the s
1.33	"Einanaina Aaraamanta" maar	se any and all decomme	ente and sarea	monte evidencine	and/or coouring	nonice provided by
	Financing Agreements" mear" nancing Entity to Purchaser to fu	_	_		· · ·	
any m	eterial part of Purchaser's Space	Segment program.	•	,,	•	
	•					
1.34	"Financing Entity" means any	_, _, _, _, _, _, _,				
design	lual, which has been specifically n, development, construction, pro ent program.	identified in a writter ocurement, maintena	n notification to nce, or operation	Contractor provi on of all or any ma	ding money to Pui iterial part of Purc	chaser to fund the haser's Space
	•					
1.35	"Firm Fixed Price" has the me	aning eat forth in Art	iolo 4 1			
1.55	Fills Fixed Fire 1825 the life	ailing set iorth ill Art	icie 4. i.			
					•	
1.36	"Force Majeure" has the mean	ing set forth in Articl	la 17		*	
1.50	i orce majeure mas the moun	ing set lotti il Aldei	ic ir.	1		
					•	
1.37	"Foreground Intellectual Prope	erty" means intellect	tual Property. c	ther than Contrac	tor Space Segmen	of Foreground
intelle	ctual Property and Purchaser Spanance of this Contract or the Ori	ace Segment Foregre	ound intellectu	al Property, first n	nade, developed o	r created in the
			7		•	
			•			· .
						·
					and the second second	The second secon

"Exhibit(s)" means the exhibit(s) identified in Article 2.1 and attached hereto and incorporated herein, as may be added,

[*] CONFIDENTIAL TREAT	MENT REQUESTED BY I	CO GLOBAL CO	MMUNICATIONS (HOLDING:	S) LIMITED.
	- ,		and the second s	

- 1.39 "GBBF Incentive Payment" has the applicable meaning set forth in Article 5.2.3B.
- "Ground-Based Beam Forming Subsystem" or "GBBF Subsystem" means the ground-based beam forming subsystem that is to be Delivered by Contractor to Purchaser pursuant to this Contract, which GBBF Subsystem shall include but not be limited to a subsystem with not less than (i) two (2) ground-based beam forming assemblies, (ii) three (3) ground-based pointing beacon stations, and (iii) related software, in conformance with the specifications contained in Exhibit E, Space Segment Performance Specification, and as more fully described in Exhibit G, Space Segment Statement of Work and as verified under the test plan in accordance with Exhibit H, Space Segment Test Plan.
- 1.41 "In-Orbit Testing" or "IOT" means the testing of the Satellite on-orbit in accordance with Article 10.1.1 and Exhibit D, Satellite Test Plan.
- 1.42 "IOT Complete Date" has the meaning set forth in Article 10.1.1.

1.38

[1]

1.43 "Intellectual Property" means all designs, methods, concepts, layouts, software, inventions (whether or not patented or patentable), processes, technical data and documentation, technical information and drawings, know how, and similar matter in which an Intellectual Property Right may subsist.

1.44	"Intellectual Property Claim" has the meaning set form in Article 18.
1.45 applic time u	"Intellectual Property Right(s)" means all common law and statutory proprietary rights, including patent, patent ation, copyright, trademark, service mark, trade secret, mask work rights, data rights and similar rights existing from time to inder the intellectual property laws of the United States, any state or foreign jurisdiction, or international treaty regime.
1.46	"Intellectual Property Documentation Dates" means each of the following: [*]
1.47 is the "inten	"Intentional Ignition" means, with respect to the Satellite, the start of the ignition process for the purpose of Launch, which the command signal is sent to the Launch Vehicle. This definition shall be modified to reflect the definition of tional ignition" in the Launch Services Agreement applicable to Launch of the Satellite.
1.48	"Key Personnel" has the meaning set forth in Article 29.2.
1.49 incorp	"Launch" means, with respect to the Satellite, Intentional Ignition followed by lift-off. This definition shall be modified to porate the definition of "launch" from the Launch Services Agreement applicable to the Launch of the Satellite.
1.50	"Launch and In-Orbit Insurance Policy" has the meaning set forth in Article 35.1.
	the Samuel of Samuel and Samuel of S
1.51	"Launch Agency" means the Person responsible for conducting the Launch Services for the Satellite.
1.52 Laund	"Launch Services" means those services provided by the Launch Agency for the Launch of the Satellite pursuant to the characteristics.
	9

[*] CONFIDENTIAL TREATMENT REQUESTED BY ICO GLOBAL COMMUNICATIONS (HOLDINGS) LIMITED.

- 1.53 "Launch Services Agreement" or "LSA" means the contract between Purchaser and the Launch Agency that provides for Launch Services for the Satellite, as such contract may be amended from time to time in accordance with its terms.
- 1.54 "Launch Site" means the location that will be used by the Launch Agency for purposes of launching the Satellite, but in the case of Sea Launch it shall mean the home port located in Long Beach, CA.
- 1.55 "Launch Support" or "Launch Support Services" means those services specified in Exhibit A, Statement of Work to be provided by Contractor in support of Launch Services.
- 1.56 "Launch Vehicle" means the launch vehicle used for Launch Services for the Satellite.
- 1.57 "Less than Satisfactorily Operating Satellite" has the meaning set forth in Article 13.2.1.
- 1.58 "Less than Satisfactorily Operating Space Segment" has the meaning set forth in Article 13.2.1.
- 1.59 "London Inter-Bank Offer Rate" or "LiBOR" means the rate per annum shown, on the third (3rd) London business day preceding the day of commencement of an interest calculation period, on page 3750 of the Dow Jones & Company Telerate screen or any successor page as the composite offered rate for London interbank deposits in an amount approximately equal to the amount on which the interest is to be applied for a three-month period (the "Rate Base"), as shown under the heading "USD" as of 11:00 a.m. (London Time); provided that in the event no such rate is shown, LIBOR shall be the rate per annum (rounded to the nearest 1/100th of one percent) based on the rates at which U.S. dollar deposits approximately equal in principal amount to the Rate Base and for a three-month period are displayed on page "LIBO" of the Reuters Monitor Money Rates Service or such other page as may replace the LIBO page on that service

two suc rates ar	purpose of displaying London interbank offered rates of ma h rates appear on such page, the rate will be the arithmetic re displayed, or if no such rate is relevant, LIBOR shall be t oximately 11:00 a.m. (London Time) to prime banks in the imately equal in principal amount to the aggregate principa	mean of su he rate per London in	ich displ annum terbank	ayed rates); equal to the market on d	provided that rate offered l eposits in U.S	t in the event re by Credit Suisse L dollars in an a	wer unan two . New York I	sucn
) }			4			٠.
1.60	"Losses" has the meaning set forth in Article 20.1.						-	
							•	
1.61	"Major Subcontract" means a subcontract related t	to the peri	iormano	e of this C	ontract and	valued at Two	Million Five	•
Hundr Purch	ed Thousand United States dollars (U.S. \$2,500,000) c aser and Contractor as material.	or more ar	nd such	other subc	ontracts the	it have been a	greed by	
1.62	"May FCC Order" has the meaning set forth in Artic	cle 5.2.3A.	• .					
					*			
1.63	"Milestone" means a specific numbered event as d	lescribed	in Exhil	oit F, Paym	ent Plan and	l Termination	Liability Am	ounts
1.64 as set	"Milestone Date" means, with respect to a particul forth in Exhibit F, Payment Plan and Termination Lia			date which	such Milest	one is schedu	led for com	pletio
	"Mission Operations Support Services" means the nent of Work and/or Exhibit G, Space Segment Staten Segment.	orbit-rais	sing, IO ork, to b	T, SSIOT a pe performe	nd related se ed by Contra	ervices specifictor for the Sa	ed in Exhibi Itellite and t	it A, the
1.66	"Net Partial Termination Claim" has the meaning s	et forth in	Article	21.2.2.				
1.67	"Net Whole Termination Claim" has the meaning s	et forth in	Article	21.2.1.				

"Notice of Election" has the meaning set forth in Article 20.3.1.

1.68

1.69	"Notice of Termination for Default" has the meaning set forth in Article 23.					
1.70	"Notice of Termination for Convenience" has the meaning set forth in Article	e 21.1.				
					•	
1.71	"Notice of Termination for Excessive Force Majeure" has the meaning set fo					
1.71	Notice of Territalidor for Excessive Porce Majeure has the meaning set to	orui in Aruc	He 23.3.			
		*				
1.72	"Notice to Stop Work" has the meaning set forth in Article 16.2.					
٠						
1.73	"NSP" means not separately priced and included in the Firm Fixed Price.	, i			•	
1.74	"Option Space Segment" has the meaning set forth in Article 24.5.					
1.75	"Orbital Performance Incentives" shall mean, collectively, the Satellite Orbit	tal Perform	ance ince	ntives and t	he Snace	
_	nt Orbital Performance Incentives.			inives and t	ire opace	
				•		
1.76	"Orbital Storage" means, with respect to an otherwise operational Satellite,	any period	of time of	f intentional	non-use by	,
Purcha	ser of such Satellite after Acceptance.					
	<u></u>					
1.77	"Original Deadline Date" has the meaning set forth in Article 5.2.3A.					
	•					
1.78	"Partial Termination Claim" has the meaning set forth in Article 21.2.2.					
1.79	rı			1		
1.80	"Party" or "Parties" means Purchaser, Contractor or both, as the context re	quires.				
	12			-		
		· .				

[*] CONFIDENTIAL TREATMENT REQUESTED BY ICO GLOBAL COMMUNICATIONS (HOLDINGS) LIMITED.

1.81 "Payment Plan" means the payment plan for the Satellite, the Space Segment and the DSS, as set forth in Exhibit F, Payment Plan and Termination Liability Amounts. The Payment Plan shall be adjusted to include other Deliverable Items in accordance with Article 24, if the applicable options are exercised, as provided in Exhibit F. The Payment Plan shall also be adjusted in accordance with Article 16, if Change Requests are implemented.

1.82 "Permitted Purposes" means [*]

1.83 "Performance Specification" means the applicable performance specifications for the Satellite, the Space Segment, the GBBF Subsystem, and other Deliverable Items in the context of the applicable clause, as such specification may be amended from time to time in accordance with the terms hereof.

- 1.84 "Person" means any individual, corporation (including any non-profit corporation), general or limited partnership, limited liability company, joint venture, estate, trust, association, organization, labor union or other entity.
- 1.85 "PMO" means the Purchaser program management office to be designated by Purchaser.
- 1.86 "Product Assurance Program Plan" means the product assurance program plan attached as Exhibit C, as may be amended from time to time in accordance with the terms hereof.
- 1.87 "Proprietary Information" has the meaning set forth in Article 31.
- 1.88 "Purchaser" has the meaning set forth in the preamble and any successor or assignee permitted hereunder.
- 1.89 "Purchaser Background Intellectual Property" means Intellectual Property first made, developed, or created by Purchaser (or by others, other than Contractor or any of its subcontractors, acting on behalf of Purchaser) prior to the negotiation or performance of the Original Satellite Contract or otherwise first made, developed, or created by Purchaser (or by others, other than Contractor or any of its subcontractors, acting on behalf of Purchaser) other than in performance of or pursuant to this Contract or the Original Satellite Contract and necessary for the use of any Deliverable Item.
- 1.90 "Purchaser Delay" has the meaning set forth in Article 18.
- 1.91 "Purchaser Space Segment Foreground Intellectual Property" means Intellectual Property first made, developed or created by Purchaser (or by others acting on behalf of Purchaser other than Contractor or its subcontractors) with respect to the GBBF Subsystem under the Original Satellite Contract and/or the Space Segment under this Contract. Notwithstanding the foregoing, "Purchaser Space Segment Foreground Intellectual Property" shall not include any Intellectual Property in or relating specifically to

Intelleg	froi Duonordy Sunt made developed on accept J. St., Co., C				
Intellet.	tual Property first made, developed or created after Space Segment Acceptance o	r after termination o	Tthis Contract.		
1.92	"Satellite" or "Spacecraft" means the communications satellite that is t red to Purchaser pursuant to this Contract.	o be manufactured	by Contractor	and to be	
Denver	ed to Furchaser pursuant to tins contract,				٠.
1.93 reason	"Satellite Anomaly" means, with respect to the Satellite on-orbit, a con- able opinion, a material adverse impact on the Satellite Stated Life or per	dition or occurrence formance of such \$	e that has, in F Satellite.	'urchaser's	
			*		
1.94	"Satellite Option" has the meaning set forth in Article 24.				
1.95	#Cotallide Ontion Function! has the manufacture of the second				
1.95	"Satellite Option Exercise" has the meaning set forth in Article 24.				
1.96	"Satellite Orbital Performance Incentive Period" has the meaning set for	rth in Article 13.1.			
1.97	"Satellite Orbital Performance Incentives" has the meaning set forth in A	Article 13.1.			
			* · ·		•
1.98	"Satellite Pre-Shipment Review" or "SPSR" has the meaning set forth in	n Article 9.			
1.99 accord	"Satellite Test Plan" means the Satellite program test plan attached as E ance with the terms hereof.	Exhibit D, as may b	e amended froi	m time to tim	ne in
1.100	"Satellite Stated Life" or "Satellite Mission Life" means, with respect to	a Satellite, the con	tracted-for life	of fifteen (1	5)
years fo of such	or such Satellite as set forth in Section 2.7 of Exhibit B, Satellite Performa Satellite.	ince Specification,	commencing u	pon Accepta	ince
1.101	"Satellite Storage Plan" has the meaning set forth in Article 24.4 (a).				

"Satisfactorily Operating Satellite" has the meaning set forth in Article 13.2.1.

"Satisfactorily Operating Space Segment" has the meaning set forth in Article 13.2.1.

1.102

1.103

"SCF" means satellite control facility. 1.104 "SCCS/E" has the meaning set forth in Article 24.2. 1.105 "Space Segment" means the space segment comprised of the Satellite and the GBBF Subsystem, integrated for the purpose of communications services, Delivered by Contractor to Purchaser pursuant to this Contract, as more fully described in Exhibit E, Space Segment Performance Specification and Exhibit G, Space Segment Statement of Work. 1.107 "Space Segment Acceptance Deadline" has the meaning set forth in Article 10.1.2. "Space Segment Acceptance Payment" has the meaning set forth in Article 5.2.3C. 1.108 "Space Segment IOT" or "SSIOT" means the Space Segment In-Orbit Test in accordance with Article 10.1.2 and as more fully described in Exhibit G, Space Segment Statement of Work and Exhibit H, Space Segment Test Plan. "Space Segment IOT Complete Date" has the meaning set forth in Article 10.1.2. 1.110 "Space Segment Option" has the meaning set forth in Article 24.5. 1.111 1.112 "Space Segment Option Exercise" has the meaning set forth in Article 24.5. 1.113 "Space Segment Option Price" has the meaning set forth in Article 24.5. "Space Segment Orbital Performance Incentive Period" has the meaning set forth in Article 13.1. 1,114 "Space Segment Orbital Performance Incentives" has the meaning set forth in Article 13.1. 1.115

time in accordance with the terms hereof.

"Space Segment Test Plan" means the Space Segment test plan attached as Exhibit H, as may be amended from time to

- [*] CONFIDENTIAL TREATMENT REQUESTED BY ICO GLOBAL COMMUNICATIONS (HOLDINGS) LIMITED.
- 1.117 "Space Segment Warranty Termination Payment" has the meaning set forth in Article 10.1.2.
- 1.118 "Spacecraft Launch Readiness Review" or "SLRR" or "LRR" means the testing and review in accordance with Article 9.2 of this Contract and Section 2.2.7.2 of Exhibit A, Statement of Work.
- 1.119 "SSIOT Deadline" has the meaning set forth in Article 10.4.
- 1.120 "Special Adjustment Factors" means [*]
- 1.121 "Standard Adjustment Factors" means [*]
- 1.122 "Statement of Work" or "SOW" means, as applicable, the statement of work attached as Exhibit A, Exhibit G or Exhibit J, as the same may be amended from time to time in accordance with the terms hereof.
- 1.123 "Storage" has the meaning set forth in Article 24.4.

- 1.124 "Summary in-Orbit Test Report" shall have the meaning set forth in Exhibit A, Statement of Work.
- 1.125 "Taxes" has the meaning set forth in Article 4.2.
- 1.126 "Terminated Ignition" means that, following the time when the electronic signal is sent to command the opening of any first stage propellant valves, the first stage engines of the Launch Vehicle are shut down for any reason before the hold down mechanism is released and the launch pad is declared safe by the Launch Agency. This definition shall be modified to incorporate the definition of "terminated Ignition" from the Launch Service Agreement applicable to the Launch of the Satellite.
- 1.127 "Third Party Intellectual Property" means Intellectual Property owned by any third party (other than a subcontractor acting for or on behalf of Contractor or Purchaser, as applicable) that is directly incorporated in and necessary for the continued use of any Deliverable Item.
- 1.128 "Total Loss" means the destruction or total operational failure of the Satellite or any other condition of the Satellite that precludes the occurrence of the IOT Complete Date. This definition shall be modified to reflect the definition of "total loss" or "total constructive loss" in the Launch and In-Orbit Insurance Policy, if such policy of insurance is obtained.
- 1.129 "Whole Termination Claim" has the meaning set forth in Article 21.2.1.
- 1.130 "Work-in-process" or "WIP" has the meaning set forth in Article 23.1.3.

1.131 "Work" means all design, development, construction, manufacturing, labor, and services, including tests to be performed, and any and all Deliverable Items, including the Satellite, the Space Segment, the DSS and the GBBF Subsystem, and software, Deliverable Data, Mission Operations Support Services, Launch Support Services, training, and equipment, materials, articles, matters, services, and things to be furnished and rights (including, but not limited to, rights with respect to Intellectual Property) to be transferred to Purchaser under this Contract, or any subcontract hereunder entered into by Contractor.

ARTICLE 2- SCOPE OF WORK

2.1	<u>Provision</u>	of Services	and Materials.
-----	------------------	-------------	----------------

Contractor hereby agrees to perform the Work in accordance with this Contract, including the following Exhibits, which are attached hereto and made a part hereof:

- 2.1.1 Exhibit A, Statement of Work, [*]
- 2.1.2 Exhibit B, Satellite Performance Specification, [*]
- 2.1.3 Exhibit C, Product Assurance Program Plan, [*]
- 2.1.4 Exhibit D, Satellite Test Plan, dated [*]
- 2.1.5 Exhibit E, Space Segment Performance Specification, dated [*]
- 2.1.6 Exhibit F, Payment Plan and Termination Liability Amounts, dated [*]
- 2.1.7 Exhibit G, Space Segment Statement of Work, dated [*]
 - 2.1.8 Exhibit H, Space Segment Test Plan, dated [*]
 - 2.1.9 Exhibit J, ICO DSS SOW and Functional Requirements, dated [*]

[*] CONFIDENTIAL TREATMENT REQUESTED BY ICO GLOBAL COMMUNICATIONS (HOLDINGS) LIMITED. ARTICLE 3- DELIVERABLE ITEMS AND DELIVERY SCHEDULE

3.1 <u>Deliverable Items.</u>

Subject to the other terms and conditions of this Contract, the items to be delivered under this Contract are specified in the table below and Delivery by Contractor shall occur on or before the corresponding dates in the table below and to the specified locations in the table below (or such other locations as the Parties may agree):

liem	Description	Delivery Date	Delivery Location
l, is in	Space Segment	Upon Acceptance under Article 10.1.2	In orbit
2	Satellite out the state of the	1986年,李建立,李建立	Confractor's Facility in Palo Allo, GA
3.	GBBF Subsystem		Purchaser's Gateways and Pointing Beacon Station Locations Per SOW, Exhibit G
4.	Deliverable Data	Per SOW, Bichilbits A sind ()	e PMO S
5. 1	Training	Per SOW, Exhibits A and C	Per SOW, Exhabits A and G
6	Lausch Support Services	Per SOW, Exhibit A	P≥ SOW, Exhibit A
7 Fl	Mission Operations Support	Per SOW, Exhibits A and O	Per SOW, Exhibits A and G
8. (1) (1)	Dynamie Satellite Simulator (DSS)		U.S. CONUS Location TBD Per Exhibit J

3.2 <u>Delivery</u>.

Delivery of each Deliverable Item other than the Satellite and the Space Segment, shall occur upon Acceptance of such Deliverable Item in accordance with Article 11. Delivery of the Satellite shall occur upon successful completion of the Satellite Pre-Shipment Review pursuant to Article 9. Delivery of the Space Segment shall occur upon Acceptance of the Space Segment pursuant to Article 10.1.2.

ARTICLE 4 -- PRICE

Firm Fixed Price.

Subject to Article 4.2 hereof, the price to be paid by Purchaser to Contractor for the Deliverable Items set forth in Article 3.1 and for the other rights and services to be provided to Purchaser pursuant to this Contract shall be a firm fixed price of [*] (the "Firm Fixed Price"), which shall be allocated among the Deliverable Items as follows (or such other allocations as the Parties may agree):

Description	Price (U.S. S)
Satellite Construction Construction of the ICO GEO Satellite, Deliverable Data, Training, Launch Support Services, Mission Operations Support Services, and other services as set forth in Article 3 and Exhibit A, Statement of Work Satellite Orbital Performance Intentives (note 1) [*] Total Satellite Construction	
Space Segment/Ground Based Beam Forming Subsystem (Not Including Satellite) Construction of the ICO Space Segment/GBBF Subsystem (not including Satellite), Deliverable Data, Training, and other services as set forth in Article 3 and Exhibit G, Statement of Work, and Space Segment Acceptance Payment set forth in Article 5.2.3C 23	[*]

GBBB Subsystem Delivery Incentive Payment (note 3) Space Segment Orbital Performance Incentives (note 4) Total Space Segment/Ground Based Beam Forming Subsystem	6 (10 de 60 % pr) s *
Dynamic Satellite Simulator	# 2 (*)
Total Contract Price	[2]

Note (1): The price indicated above includes the maximum potential Satellite Orbital Performance Incentives, subject to the terms of Article 13

[*]
Note (3): The price indicated above includes the maximum potential GBBF Incentive Payment, subject to the terms of Article 5.2.3B
Note (4): The price indicated above includes the maximum potential Space Segment Orbital Performance Incentives, subject to the terms of Article 13.
Notwithstanding the above, as provided in Article 13.1, under certain circumstances, the Space Segment Orbital Performance Incentives may increase up to

Except as otherwise expressly provided in this Contract, the Firm Fixed Price is not subject to any escalation or to any adjustment or revision. The price for those items subject to an option under this Contract is set forth in Article 24.

Except as expressly provided by this Contract, the item price for the Satellite, the Space Segment, the GBBF Subsystem, and the other Deliverable Items specified herein includes all parts and services to be provided pursuant to this Contract including, without limitation, on—going design, manufacturing, installation, tests,

Deliverable Data, rights and interests in Intellectual Property as set forth in this Contract, training, and insurance support services as required by Article 35, packing and transport of the Satellite to the Launch Site, transit insurance and such other insurance as is required by Article 28, all in accordance with the terms and conditions of this Contract, as specified herein.

4.2 <u>Taxes</u>.

The Firm Fixed Price includes all applicable taxes, duties and similar liabilities (including interest, fines, penalties, or additions attributable or imposed on or with respect to, any such taxes, duties and similar liabilities) imposed by any United States federal, state, or local government in connection with this Contract and the taxing authorities having jurisdiction over Launch Services and the Launch Site ("Taxes") with respect to the Work; provided, however, the Firm Fixed Price does not include any Taxes on the sale of the GBBF Subsystem, the DSS or the SCCS/E (if the applicable options are exercised), the payment of which Taxes shall be the responsibility of Purchaser. Subject to Article 12.3, Contractor has made or will make all the necessary filings in order to deliver the Work (except the GBBF Subsystem, the DSS, and the SCCS/E with respect to Taxes that are the responsibility of Purchaser under this Contract) free and clear of any liens or encumbrances for Taxes. Subject to the indemnification procedures set forth in Article 20.3, in the event any governmental or taxing authority imposes or assesses Taxes against Purchaser in connection with any Deliverable Item (except as otherwise provided), Contractor shall indemnify Purchaser for any Taxes, interest and penalties on such Taxes paid by or assessed against Purchaser and shall reimburse Purchaser for the reasonable actual costs incurred by Purchaser for defense of any action for payment of such Taxes (whether or not Purchaser actually pays such Taxes).

- 4.3 The Firm Fixed Price includes Launch Support Services for Launch of the Satellite on a U.S. domestic Launch Vehicle, including the Sea Launch Launch Vehicle. In the event that the Purchaser elects to Launch the Satellite on a non-U.S. domestic Launch Vehicle, the Firm Fixed Price shall be increased by one of the following amounts:
 - i. For Launch on an Ariane Launch Vehicle:
 - ii. For Launch on a Proton Launch Vehicle:

ARTICLE 5 - PAYMENTS

5.1 Payment Plan.

Payments by Purchaser to Contractor of the Firm Fixed Price set forth in Article 4, and of the amounts for options, if exercised by Purchaser pursuant to this Contract (other than pursuant to Article 24.5), shall be in accordance with Exhibit F, Payment Plan and Termination Liability Amounts, as applicable thereto. Contractor shall deliver the applicable invoice for payment and any required certification within fifteen (15) days after the later of the completion of the relevant Milestone or the applicable Milestone Date.

- 5.2 Payment Conditions.
- 5.2.1 <u>Orbital Performance Incentive Payments</u>, All Satellite Orbital Performance Incentive and Space Segment Orbital Performance Incentive payments shall be made in accordance with Article 13.
- 5.2.2 <u>Milestone Payments</u>. Subject to Article 5.6, all payments due from Purchaser upon the completion of a Milestone described in the Payment Plan shall be paid no later than thirty (30) days after the receipt by Purchaser of an invoice and certification (delivered in accordance with Article 5.1 above) in the form attached hereto as Attachment A that the Milestone has been completed in accordance with the requirements of this Contract, together with the necessary or appropriate supporting data and documentation as required hereunder, if any, or as Purchaser may reasonably request within ten (10) days of receipt of invoice. Notwithstanding the foregoing, and without prejudice to Purchaser's rights under Article 5.6, Purchaser, in its sole discretion, may agree to make a partial payment to Contractor for partial completion of a Milestone or for completion of a Milestone prior to the applicable Milestone Date.

5.2.3B GBBF incentive Payment.

Without limiting Contractor's right to any Space Segment Orbital Performance Incentives earned pursuant to Article 13, Contractor may earn, and if earned Purchaser shall pay, [*] (the "GBBF Incentive Payment") upon Purchaser's Acceptance of the GBBF Subsystem in accordance with Article 11.2.2 on or prior to [*]. If Purchaser's Acceptance of the GBBF Subsystem does not occur on or before [*], then, except as provided in this Article 5.2.3B, no GBBF Incentive Payment shall be earned regardless of whether such failure is due to Force Majeure or any other cause or event provided that the date may be extended to the extent provided by and subject to Article 18.

[*]

The amount payable to Contractor under this Article 5.2.3B shall be made no later than thirty (30) days after receipt of an invoice and certification from Contractor that the GBBF event has been completed and is due in accordance with this Article 5.2.3B and documented in accordance with the requirements of this Contract together with the necessary or appropriate supporting data and documentation as required under this Contract. Payment of the GBBF Milestone for purposes of this Contract.

5.2.3C Space Segment Acceptance Payment.

Without limiting Contractor's right to any Space Segment Orbital Performance Incentives earned pursuant to Article 13, if (but only if) the Space Segment is established to be a Conforming Space Segment at the time of Space Segment Acceptance, Purchaser shall pay to Contractor an amount equal to [*] The amount payable to Contractor under this Article 5.2.3C shall be made no later than thirty (30) days after receipt of an invoice and certification from Contractor that the Space Segment Acceptance Payment is due in accordance with this Article 5.2.3C, together with the necessary or appropriate supporting data and documentation supporting such certification as required under this Contract. Payment of the Space Segment Acceptance Payment shall also be subject to the verification process set forth in Article 5.2.2.

5.2.4 Other Payments.

Unless as otherwise specified herein, all amounts payable to Contractor with respect to other payments not covered in Articles 5.2.1, 5.2.2 and 5.2.3 above, shall be made no later than thirty (30) days after receipt of an invoice and certification in the form attached hereto as Attachment A that such other work has been completed and documented in accordance with the requirements of this Contract together with the necessary or appropriate supporting data and documentation as required under this Contract, if any, or as Purchaser may reasonably request within ten (10) days of receipt of invoice.

5.2.5 Obligation to Pay.

If Contractor shall not have delivered any invoice required hereunder within the time specified therefor, the relevant payment due from Purchaser shall be suspended (without interest) and shall become payable on or before thirty (30) days after receipt of such invoice.

5.3 <u>Late Payment.</u>

- 5.3.1 <u>Interest.</u> In the event that any payment due under this Contract is not made when due hereunder, without prejudice to the other rights and remedies under this Contract or at law or in equity under this Contract of the Party entitled to such payment, such Party shall also be entitled to interest at the annualized rate of [*] on the unpaid balance thereof from the date such payment is due hereunder until such time as payment is made.
- 5.3.2 Stop Work. If any payment is not made by Purchaser by the date thirty (30) days after the due date, Contractor may issue to Purchaser a written notice of reminder for payment. Subject to Article 5.6, if Purchaser does not make the payment within five (5) days of the receipt of the written reminder notice, without prejudice to Contractor's other rights and remedies under this Contract, Contractor may, upon written notice to Purchaser, elect to suspend performance of the Work. If Contractor subsequently resumes performance of the Work, this Contract shall be Equitably Adjusted.

5.4 <u>Invoices</u>.

Invoices required to be delivered by Contractor hereunder shall be submitted by facsimile and air mail to Purchaser (original plus one (1) copy, including supporting documentation and data) at the following address:

ICO Satellite Management LLC 222 N. Sepulveda Boulevard Suite 1770 El Segundo, CA 90245 Attn: Contract Manager Facsimile: [*] Telephone: [*]

With a copy to:
ICO Satellite Management LLC
2300 Carillon Point
Kirkland, WA 98033
Attn: CFO
Facsimile: [*]
Telephone: [*]

or to such other address as Purchaser may specify in writing to Contractor.

5.5 Payment Bank.

All payments made to Contractor hereunder shall be in U.S. currency and shall be made by electronic funds transfer to the following account: SPACE SYSTEMS/LORAL, INC.

[*]

or such other account or accounts as Contractor may specify in writing to Purchaser.

5.6 <u>Disputed Amounts</u>.

In the event Purchaser determines in good faith that the Milestone covered by Contractor's invoice has not been completed in accordance with the requirements of this Contract, Purchaser shall so notify Contractor in writing within fifteen (15) days of receipt of the invoice and applicable certification. Such notification shall state in reasonable detail the area(s) Purchaser considers not to be completed in accordance with the requirements of this Contract. Upon completion of the Milestone in accordance with the requirements of the Contract, the Contract or shall issue a replacement invoice. Until the Milestone is completed in accordance with the Contract and issuance of the replacement invoice not disputed by Purchaser in good faith, Purchaser shall have no obligation for payment and no interest shall accrue. Notwithstanding the foregoing, in the event that the noted discrepancy(ies) in Milestone completion does not in Purchaser's good faith judgment substantially impair the completion of the Milestone, Purchaser shall still pay such invoice; Purchaser, however, may withhold ten percent (10%) of the affected payment which Contractor may immediately invoice upon correction of the discrepancy. Upon receipt of notification from Purchaser contending a Milestone had not been completed, the Parties' respective program managers shall meet and use good faith efforts to resolve such disagreement.

5.7 <u>Set Off</u>.

In the event one Party has not paid any amount that is due and payable under this Contract, the other Party shall have the right to set off such amount against any payments due to the other Party under this Contract, provided any amount in dispute pursuant to Article 5.6 shall not be considered eligible for setoff while the dispute is being resolved.

ARTICLE 6 - PURCHASER-FURNISHED ITEMS

Purchaser-Furnished Support.

6.1

To enable Contractor to install the GBBF Subsystem and to perform Launch Support Services and Mission Operations Support Services, Purchaser shall timely make available to Contractor the Purchaser-furnished equipment, facilities and services described in Section 2.4 (Purchaser Furnished Items) of Exhibit A, Statement of Work and Section 2.4 (Purchaser Furnished Items) of Exhibit A, Statement of Work and Section 2.4 (Purchaser Furnished Items) of Exhibit B, Space Segment Statement of Work at the applicable times set forth in such Exhibits, as the same may be adjusted. Such equipment, facilities and services shall be in good working condition and adequate for the required purposes and, for the installation of the GBBF Subsystem and the Launch of and Mission Operations Support Services for the Satellite hereunder, shall be made available free of charge for Contractor's use (solely for the purposes of this Contract) during the period(s) specified in Exhibit A, Statement of Work or Exhibit G, Space Segment Statement of Work, as the case may be. Purchaser and Contractor will conduct an interface meeting on the date established therefor at the technical interchange meeting described in Section 2.3 (Deliverable Hardware, Software, and Services) of Exhibit A, Statement of Work to confirm the availability and adequacy of such Purchaser-furnished equipment, facilities and services.

6.2 <u>Communications Authorizations.</u>

Contractor shall be responsible, at its cost and expense, for preparing, coordinating and filing all applications, registrations, reports, licenses, permits and authorizations required of Contractor to perform its obligations under this Contract.

Purchaser shall be responsible, at its cost and expense, for preparing, coordinating and filing all applications, registrations, reports, licenses, permits and authorizations with the FCC if required to do so and with any other national governmental agencies having jurisdiction over Purchaser, for the construction,

launch and operation of the Satellite and Space Segment. Contractor shall provide such reasonable cooperation and support as Purchaser may reasonably request in support of Purchaser's preparation, coordination and filing of such applications, registrations, reports, licenses, permits and authorizations. From Actual Costs reasonably incurred in connection with the provision of such support plus a markup of [*] with such Actual Costs and associated markup to be invoiced and paid in accordance with Article 5.

6.3 Radio Frequency Coordination.

Purchaser shall be responsible for the timely preparation and submission of all filings required by the International Telecommunication Union (or any successor agency thereto) and all relevant domestic communications regulatory authorities regarding radio frequency and orbital position coordination. Such regulations of all domestic communications regulatory authorities having jurisdiction over Purchaser. Contractor shall, at its own cost and expense, provide such reasonable cooperation and support as Purchaser may reasonably request in support of Purchaser's efforts in the preparation and submission of such filings. From and after Acceptance of the Space Segment, for any support provided by Contractor under this Article 6.3, Contractor shall be entitled to reimbursement of Actual Costs reasonably incurred in connection with the provision of such support plus a markup of [*] with such Actual Costs and

6.4 Satellite Performance Data.

Commencing with the first full calendar month following Acceptance of the Satellite, Purchaser shall provide a report to Contractor describing the general health and operating status of such Satellite. Such report shall be provided to

Contractor on a monthly basis thereafter, delivered to Contractor promptly after the end of each month during the Satellite Stated Life. In the event of a Satellite Anomaly that occurs during the Satellite Stated Life, Purchaser shall timely provide Contractor with or give Contractor access to any data Contractor may reasonably require to investigate or correct (if Contractor is able to do so) such Satellite Anomaly and/or support Purchaser in making and perfecting claims for insurance recovery relating to such Satellite Anomaly as set forth in Article 35.2. In no event shall Purchaser have liability for failing to deliver any report under this Article 6.4 or for the contents thereof except to the extent, if any, such failure to report is a material reason Contractor shall fail to earn Satellite Orbital Performance Incentives in accordance with Article 13 hereof (and then only with respect to such unearned Satellite Orbital Performance Incentives).

6.5 Late Delivery of Purchaser-Furnished Items or Services.

The late delivery of Purchaser-furnished items or services, individually or combined, shall be subject to Article 18.

6.6 Launch Services.

Purchaser shall be responsible for furnishing Launch Services for the Satellite and Purchaser shall no later than twelve (12) months prior to Delivery of the Satellite, notify Contractor of its selection of the Launch Vehicle for the Launch of the Satellite. The Launch Vehicle shall be one of the candidate Launch Vehicles identified in Paragraph 2.1 (Launch Requirements) of Exhibit B, Satellite Performance Specification, and the Launch Date shall be compatible with the Satellite Delivery schedule set forth in Article 3.1 hereof.

ARTICLE 7 – COMPLIANCE WITH U.S. LAWS AND DIRECTIVES

7.1 General.

Each Party shall, at its expense, perform their respective obligations hereunder in accordance with all applicable laws, regulations, and policies of the United States and the conditions of all applicable United States Government approvals, permits, or licenses.

7.2 Compliance with U.S. Export Control Laws.

- 7.2.1 Contractor shall, at its expense, perform the Work in accordance with all applicable export control laws, regulations, and policies of the United States and the conditions of all applicable United States Government approvals, permits, and licenses and shall provide Purchaser such assistance as Purchaser may reasonably request.
- 7.2.2 Any obligation of Contractor hereunder to provide hardware, software, Deliverable Data, other technical information or technical services to Purchaser and its personnel and/or its representatives shall be subject to applicable U.S. Government export control and security laws, regulations, policies and license conditions, as construed by Contractor in good faith. The Parties shall work cooperatively and in good faith to implement this Contract in compliance with such laws, regulations, policies and license conditions. If and to the extent required by U.S. law, Purchaser and its personnel and/or representatives shall enter into U.S. Government-approved agreement(s), separate from this Contract, governing Contractor's provision of hardware, software, Deliverable Data, other technical information or technical services in connection with this Contract.

7.3 <u>Licenses and Other Approvals.</u>

Contractor shall at its expense timely apply for and, once issued, maintain all U.S. Government export licenses, agreements and other approvals that are required for "foreign person" personnel and/or representatives of Purchaser

(including, but not limited to, all foreign Affiliates and related entities of Purchaser involved with the procurement) as well as Purchaser's insurance providers to have access to Contractor facilities, hardware, software, Deliverable Data, other technical information or technical services in connection with As early as practicable, Purchaser shall provide Contractor with a list of countries (if other than the U.S.) of which "foreign person" personnel and/or Purchaser's insurance providers if such personnel, representatives, insurance providers will or may have access to U.S. export—controlled items under this licenses, agreements and other approvals, and shall promptly notify Contractor of any occurrence or change in circumstances of which it becomes aware involved with the procurement) as a named party in any application to the U.S. government for approval of such export licenses, agreements and other approvals as a named party in any application to the U.S. government for approval of such export licenses, agreements and other procurement) as a named party in any application to the U.S. government for approval of such export licenses, agreements and other providers may receive from or discuss with Contractor export—controlled technical data. Contractor shall provide the parties to such export licenses and agreements, including any U.S. government provisos related to same.

NOTWITHSTANDING ANY PROVISION IN THIS CONTRACT, IN NO EVENT SHALL CONTRACTOR BE OBLIGATED UNDER THIS CONTRACT TO PROVIDE ACCESS TO CONTRACTOR OR SUBCONTRACTOR FACILITIES; PROVIDE ACCESS TO OR FURNISH HARDWARE, SOFTWARE, DELIVERABLE DATA OR OTHER TECHNICAL INFORMATION; OR PROVIDE TECHNICAL SERVICES, TO ANY PERSON EXCEPT IN COMPLIANCE WITH APPLICABLE U.S. EXPORT CONTROL LAWS, REGULATIONS, POLICIES AND LICENSE CONDITIONS, AS CONSTRUED BY CONTRACTOR IN GOOD FAITH.

7.4 No Unauthorized Exports or Retransfers.

PURCHASER SHALL NOT EXPORT OR TRANSFER TO ANY "FOREIGN PERSON" ANY HARDWARE, SOFTWARE, DELIVERABLE DATA, OTHER TECHNICAL INFORMATION OR TECHNICAL SERVICES FURNISHED HEREUNDER, EXCEPT AS EXPRESSLY AUTHORIZED BY THE U.S. GOVERNMENT IN ACCORDANCE WITH THE EXPORT LICENSES, AGREEMENTS AND OTHER APPROVALS REFERENCED IN ARTICLES 7.2 AND 7.3 OR AS OTHERWISE EXPRESSLY AUTHORIZED UNDER U.S. EXPORT CONTROL LAWS.

IF PURCHASER IS A "FOREIGN PERSON", PURCHASER UNDERSTANDS AND WARRANTS THAT IT SHALL NOT RE-EXPORT, RE-TRANSFER OR DIVERT TO ANY THIRD PARTY ANY ITEM EXPORTED TO PURCHASER UNDER OR IN CONNECTION WITH THIS CONTRACT, EXCEPT AS EXPRESSLY AUTHORIZED BY THE U.S. GOVERNMENT IN ACCORDANCE WITH THE EXPORT LICENSES, AGREEMENTS OR OTHER APPROVALS REFERENCED IN ARTICLES 7.2 AND 7.3 OR AS OTHERWISE EXPRESSLY AUTHORIZED UNDER U.S. EXPORT CONTROL LAWS.

ARTICLE 8 - ACCESS TO WORK-IN-PROCESS

8.1 General.

Contractor represents and warrants that, subject to Article 7, the access to the Work (which shall include access to the Work—in-process) to be provided Purchaser personnel (and Purchaser's consultants and agents) under this Contract is, or shall be, substantially similar to the access Contractor provides to its extensive as required by this Contract, Contractor shall promptly remedy that situation.

8.2 Work-in-Process at Contractor's Facility.

Subject to Article 7 and compliance with Contractor's normal and customary safety and security regulations and practices, Purchaser personnel (and Purchaser's consultants and agents) shall be allowed access to all Work being performed at Contractor's facility for the Satellite, the GBBF Subsystem, the Space Segment, the DSS and other Deliverable Items, for the purpose of observing the progress of such Work. Subject to Article 7, Purchaser shall be provided not less than [*] non-escort permanent badges and not less than [*] escort badges to agreed work areas where the Work is being performed. For Purchaser's personnel, consultants and agents without such badges, access to the Work shall be, upon reasonable prior notice to Contractor, and shall occur

8.3 Work-in-Process at Subcontractors' Facilities.

In the case of Contractor's Major Subcontracts (which, for purposes of this Article 8.3, shall include Contractor's subcontract with the principal GBBF Subsystem supplier regardless of the value of such subcontract), and such other non-major subcontracts under which the effort in support hereunder involves significant design or qualification or the subcontractor is experiencing performance

difficulties which may have a material adverse effect on the Delivery schedule or performance or provision of a Deliverable Item, Contractor shall use commercially reasonable efforts to require that each such subcontract contain a provision substantially similar to this Article 8 with respect to access to the applicable subcontractor's facilities and performance of the Work. Such access shall be subject to Article 7 and (i) each such subcontractor's safety and contractor's safety a security regulations of which Purchaser is advised in writing prior to any visit, and (ii) the right of Contractor to accompany Purchaser on any such visit to a

On-Site Facilities for Purchaser's Personnel.

For the purpose of monitoring the progress of the Work being performed by Contractor hereunder, Contractor shall provide office facilities at or proximate to Contractor's plant for not less than [*] resident Purchaser personnel (and/or Purchaser's duly appointed consultants and agents) through Acceptance of the last Satellite ordered hereunder (or through Acceptance and completion of all testing of the associated Space Segment, if Acceptance therefor is later). The office facilities to be provided shall include such services as Purchaser shall reasonably require including a reasonable amount of office space, office internet access and to the extent available, videoconference rooms, and car parking facilities, to the extent necessary to enable Purchaser personnel to monitor the progress of Work under this Contract. In the case of Contractor's Major Subcontracts related to the payload, Contractor shall use commercially reasonable efforts to ensure that reasonable office facilities at the sites of such major subcontractors for up to [*] Purchaser personnel (and/or Purchaser's duly appointed consultants and agents) are provided on a temporary basis to attend meetings or witness tests.

8.5 <u>Purchaser Representatives as Competitors/Foreign Persons.</u>

Purchaser's consultants and agents that are provided access to the Work shall not be in direct Competition with and shall not currently be employed by companies or entities that are Competitors of Contractor. Purchaser shall notify Contractor in writing of the name, title or function, business relationship, employer, citizenship status under U.S. export laws and such other information as may be reasonably requested by Contractor, with respect to each of its intended consultants and agents, and cause each such consultant and agent to (i) execute a confidentiality agreement directly with Contractor in form and substance reasonably satisfactory to Contractor and containing terms substantially the same as those set forth in Article 31 and (ii) pursuant to Article 7, execute a Technical Assistance Agreement or other agreement to ensure compliance with applicable U.S. export control laws and regulations to the extent required by applicable U.S. export laws or regulations as interpreted by Contractor in good faith. Contractor may in good faith deny any consultant or agent of Purchaser access to Contractor facilities, products or information.

8.6 <u>Interference with Operations.</u>

Purchaser shall in good faith exercise its rights under this Article 8 in a manner that does not unreasonably interfere with Contractor's or its subcontractors' normal business operations or Contractor's performance of its obligations under this Contract or any agreement between Contractor and its subcontractors.

8.7 <u>Financing Entities</u>.

Subject to the provisions of Article 7, and compliance with Contractor's normal and customary safety and security regulations and practices of which Purchaser has received prior written notice, each Financing Entity shall have access to the Work in the same manner and to the same extent as Purchaser's consultants and agents under this Article 8.

ARTICLE 9 – SATELLITE PRE-SHIPMENT REVIEW (SPSR) AND SPACECRAFT LAUNCH READINESS REVIEW (SLRR)

- 9.1 Satellite Pre-Shipment Review (SPSR).
- 9.1.1 <u>Contractor to Review Each Satellite Prior to Shipment</u>, Contractor shall conduct a review of the Satellite prior to Contractor's shipment of such Satellite to the Launch Site or placement in Storage, as applicable. Upon Contractor determining in good faith that the Satellite is ready for Delivery, Contractor shall conduct a Satellite Pre-Shipment Review. This review shall be in accordance with the terms of this Article 9 and Section 2.2.4 (Satellite Pre-Shipment Review) of Exhibit A, Statement of Work (a "Satellite Pre-Shipment Review" or "SPSR").
- 9.1.2 <u>Time, Place and Notice of SPSR: Failure to Conduct</u>, The SPSR shall take place at Contractor's facility. Contractor shall notify Purchaser in writing on or before thirty (30) days prior to the scheduled date for the commencement of such SPSR for the Satellite. If such time is not convenient for Purchaser, Contractor shall make reasonable efforts to accommodate Purchaser's scheduling requirements. If Purchaser should fail to participate in the SPSR with respect to the Satellite or fails to provide written notice of objections in accordance with Article 9.1.5 within seventy—two (72) hours of completion of the SPSR. Purchaser shall be deemed to have notified Contractor pursuant to Article 9.1.5 hereof that SPSR for such Satellite has been successfully completed.
- 9.1.3 Conduct and Purpose of SPSR, Each SPSR shall be conducted in accordance with the terms of this Article 9 and Section 2.2.4 (Satellite Pre-Shipment Review) of Exhibit A, Statement of Work. The purpose of each SPSR shall be to (i) review test data and analyses for the Satellite, (ii) demonstrate all testing has been completed in accordance with Exhibit D, Satellite Test Plan, and (iii) determine whether such Satellite meets applicable Satellite performance specification requirements, as set forth in

Exhibit B, Satellite Performance Specification (except those requirements that have been waived pursuant to Article 9.3 below), and is ready for shipment to the Launch Site.

- 9.1.4 <u>Purchaser's Inspection Agents.</u> Purchaser may, subject to prior written notice to Contractor, cause any Purchaser personnel, consultant or agent designated by Purchaser to observe the SPSR pursuant to this Article 9; provided, however, that the provisions of Article 7 and Article 8.5 shall apply to any such Person.
- 9.1.5 SPSR Results. In the event that such SPSR demonstrates (i) that all testing has been performed in accordance with Exhibit D, Satellite Test Plan and (ii) conformity of such Satellite to the applicable requirements of Exhibit B, Satellite Performance Specification, (including any waivers or deviations approved by Purchaser pursuant to Article 9.3), Purchaser shall within seventy—two (72) hours of completion of the relevant SPSR direct Contractor in writing either to ship the Satellite to the Launch of Including the Satellite in storage in accordance with Article 24.4. In the event that such SPSR discloses: (i) a failure of conduct testing in accordance with Exhibit D, Satellite Test Plan, or (ii) a non—conformance of such Satellite to the requirements of Exhibit B, Satellite Performance Specification, either of which is not the subject of any waivers or deviations approved by Purchaser pursuant to Article 9.3, Purchaser shall, within seventy—two (72) hours after completion of the relevant SPSR, deliver a written notice to Contractor setting forth the testing and/or contractual requirements that Purchaser believes have not been met, and Contractor shall correct or repair each such non—conformance and resubmit such Satellite for additional testing, in accordance with applicable requirements of Exhibit D, Satellite Test Plan and a second SPSR shall be held in accordance with the procedures of this Article 9. Any additional testing and a subsequent SPSR shall be conducted to the extent relevant

and necessary to demonstrate that the Satellite conforms to the requirements of Exhibit B, Satellite Performance Specification. Upon direction from Purchaser to Contractor to ship the Satellite to the Launch Site, Contractor shall transport such Satellite, in accordance with Contractor's Contractor be required to ship the Satellite to the Launch Site until all non-conformances are corrected, repaired or have an approved waiver (or deviation) pursuant to Article 9.3.

- 9.1.6 <u>Inspection Costs Borne by Purchaser</u>. All costs and expenses incurred by Purchaser and its agents in the exercise of its inspection rights under this Article 9, including travel and living expenses, shall be borne solely by Purchaser.
- 9.1.7 Correction of Deficiencies after SPSR. If at any time following the SPSR for the Satellite and prior to Intentional Ignition (or, in the event of a Terminated Ignition, prior to any subsequent Intentional Ignition), such Satellite fails to meet Exhibit B, Satellite Performance Specification, as may be modified as of costs occurred as a result of Terminated Ignition) promptly undertake to correct such deficiencies prior to Intentional Ignition (or, in the case of a Terminated Ignition, prior to any subsequent Intentional Ignition).

Spacecraft Launch Readiness Review (SLRR).

9.2

After shipment of the Satellite to the Launch Site and prior to Launch, Contractor shall conduct a Spacecraft Launch Readiness Review in accordance with the requirements of Section 2.2.7.2 (Spacecraft Launch Readiness Review) of Exhibit A, Statement of Work. Contractor shall give Purchaser personnel reasonable advance notice of such SLRR. The purpose of the SLRR is to verify that all testing required under Exhibit D, Satellite Test Plan, has been successfully completed and the Satellite meets all the parameters required to be tested thereunder (including any waivers or deviations approved by Purchaser pursuant to Article 9.3). The SLRR will be deemed successfully completed when Contractor has verified and the Parties agree that the above-stated purposes of the SLRR have been met and the Satellite is ready to be integrated with the Launch Vehicle. Within twenty-four (24) hours after completion of the SLRR, Purchaser shall notify Contractor of its (i) concurrence with the results of the SLRR, including any waiver of its right to compet correction of those non-conformances to the applicable provisions of this Contract or (ii) specific non-conformances of such Satellite to the requirements of this Contract which are not the subject of any waivers or deviations approved by Purchaser. Purchaser's notice shall state each such non-conformance (with reference to the applicable requirements of this Contract deemed not met) it requires to be corrected or repaired, and Contractor shall correct or repair each such non-conformance and resubmit such Satellite for Purchaser's concurrence. Purchaser's failure to provide notice of specific non-conformance within twenty-four (24) hours after receipt of Contractor's verification shall be deemed concurrence.

9.3 Waivers and Deviations.

Contractor shall timely notify Purchaser in writing from time to time of any failure of the Work to meet the requirements of the Contract. With respect to any deviation or waiver from the Contract that Contractor shall in reasonable good faith believe that Purchaser should waive or approve or that Purchaser shall determine to waive or approve, Contractor shall submit to Purchaser a request for a waiver of, or deviation from, the provisions(s) of the Performance Specification applicable to the Satellite or Space Segment, which request shall set forth in detail the extent of the impact on the Performance Specifications. A request for waiver or deviation shall be deemed granted only if it has been approved in writing by a duly authorized representative of Purchaser and expressly approving the modification to the Performance Specifications. Each such waiver or deviation approved by Purchaser shall be deemed an amendment to the Performance Specification for such Satellite or Space Segment, permitting such waiver thereof, or deviation therefrom, effective on or after the date of such approval for such Satellite or Space Segment. Purchaser shall consider each waiver or deviation request in good faith.

ARTICLE 10 - SATELLITE AND SPACE SEGMENT ACCEPTANCE

10.1 <u>In-Orbit Testing (IOT)</u>.

10.1.1 Satellite. Acceptance of the Satellite shall be deemed to occur upon the earlier of: (i) the IOT Complete Date (which shall occur as set forth below unless there is a Total Loss of the Satellite); or (ii) the instant immediately prior to an event on or after Intentional Ignition resulting in the Total Loss of the Satellite (or the Satellite being reasonably determined to be a Total Loss) ("Acceptance" for a Satellite). Upon arrival of the Satellite at its specified orbital location, Contractor shall, in accordance with the Exhibit A, Statement of Work and Exhibit D, Satellite Test Plan, perform the IOT. When the IOT has been completed for the Satellite, Contractor shall conduct an IOT Data Review Meeting (in accordance with Exhibit A, Statement of Work), during which Contractor shall submit the IOT results to Purchaser. Within three (3) days after the completion of such IOT Data Review Meeting, Purchaser shall notify Contractor of its acceptance of the IOT results. The IOT Complete Date shall occur upon the earlier of: (i) the date on which Purchaser shall have notified Contractor in writing of its Acceptance of IOT results and (ii) the third day following the IOT Data Review Meeting ("IOT Complete Date"). Notwithstanding any of the foregoing, Purchaser may direct Contractor to perform testing other than as required for IOT at any time during IOT or after receipt of the IOT results and Contractor agrees to undertake such additional tests subject to: (i) agreement on who will pay the costs of such additional tests, and (ii) commencement of the Satellite Orbital Performance Incentive Period for purposes of earning the Satellite Orbital Performance Incentives to be paid pursuant to Article 13 hereof.

10.1.2 Space Segment.

A. Acceptance of the Space Segment shall occur upon the earlier of (i) the instant immediately prior to an event on or after Intentional Ignition resulting in the Total Loss of the Satellite (or the Satellite being reasonably determined to be a Total Loss); (ii) the establishment in accordance with Article 10.1.2B prior to the Space Segment Acceptance Deadline that the Space Segment [*]

Nothing in this Article 10.1.2 shall in any manner affect Acceptance of the Satellite under Article 10.1.1 or otherwise establish obligations or liabilities of Contractor with respect to the Satellite other than those set forth in this Contract without this Article 10.1.2.

[*]

[*]

B. Contractor shall perform the Space Segment IOT in accordance with the applicable provisions of Exhibit G, Space Segment SOW, and Exhibit H, Space Segment Test Plan. When the Space Segment IOT has been completed, Contractor shall conduct a Space Segment IOT Review (in accordance with Exhibit G, Space Segment Statement of Work), during which Contractor shall submit the Space Segment IOT results to Purchaser. Within seventy—two (72) hours after completion of such review meeting. Purchaser shall notify Contractor of its acceptance of the IOT results (the "Space Segment IOT Complete Date"). Should the results of Space Segment IOT reviewed at the Space Segment IOT Review establish that the Space Segment is a [*] the Space Segment shall be Accepted on such basis pursuant to the criteria set forth in subclause (ii) of the first paragraph of Article 10.1.2A. Notwithstanding any of the foregoing: (i) should the results of Space

Segment IOT reviewed at the Space Segment IOT Review establish that the Space Segment is a [*] then Purchaser may direct Contractor, or Contractor may elect, at the review meeting to continue appropriate efforts to achieve the performance of a [*] (whichever shall first occur) up until Acceptance of the Space Segment upon the Space Segment Acceptance Deadline; and (ii) Purchaser may direct Contractor to perform testing other than as required of SSIOT at any time during the Space Segment IOT or after receipt of the Space Segment IOT results and Contractor agrees to undertake such additional tests subject to: (a) agreement on who will pay the costs of such additional tests, and (b) commencement of the Space Segment Orbital Performance Incentive Period for purposes of earning the Space Segment Orbital Performance Incentives to be paid pursuant to Article 13 hereof. Notwithstanding the above, all tests to establish that the Space Segment [*] shall be borne by Contractor and Space Segment Orbital Performance Incentives shall in no event commence prior to Acceptance of the Space Segment.

Purchaser may, subject to prior written notice to Contractor, cause any Purchaser personnel, consultant or agent designated by Purchaser to observe the Space Segment IOT.

Unless otherwise agreed, Acceptance of the Space Segment pursuant to subclause (ii) or (iii) of the first paragraph of Article 10.1.2A shall not occur prior to the SSIOT Complete Date.

C. Except as the Parties may agree in their discretion, SSIOT shall not commence prior to Acceptance of the GBBF Subsystem, provided that upon the request of either Party, the other Party will give good faith consideration to any request to commence SSIOT prior to Acceptance

of the GBBF Subsystem, taking into account pertinent factors including but not limited to material non-conformances of the GBBF Subsystem with Annex B, Specification for the GBBF Subsystem, of Exhibit E, Space Segment Performance Specification. Notwithstanding the above, the FCC certification as Subsystem.

D. [*]

E. [*]

10.2 Consequences of Acceptance of the Satellite and Space Segment,

10.2.1 <u>Consequences of Acceptance of the Satellite</u>. After Acceptance of the Satellite, Contractor shall have no further obligation with respect to such Satellite, except as otherwise provided in Articles 10.1.1 (if Purchaser orders additional testing), 10.3, 15.2, and 15.2, moreover, except as otherwise provided in Article 15.1.3, 15.2, 19, and 35.2, Contractor's loss of Orbital Performance Incentives pursuant to Article 13 shall be Purchaser's sole and exclusive remedy with respect to the Satellite's performance after Intentional Ignition.

10.2.2 <u>Consequences of Acceptance of the Space Segment</u>, Upon and after Acceptance of the Space Segment, Contractor shall have no further obligation with respect to such Space Segment, [*] This Article 10.2.2 shall not affect Purchaser's other express rights under this Contract directly relating to the Space Segment including under Articles 19 and 32.

10.3 Post-Eclipse Report.

In the event no solar eclipse occurs during IOT, Contractor shall, at Purchaser's request, and subject to making the Satellite and necessary Purchaser facilities available to Contractor, for the first solar eclipse following completion of IOT, conduct eclipse testing in accordance with Exhibit D, Satellite Test Plan, and prepare and provide Purchaser an eclipse test report detailing the Satellite's performance against the applicable Satellite performance requirements during an eclipse in accordance with Exhibit B, Satellite Performance Specifications. If the Satellite does not meet all the Exhibit B, Satellite Performance Specifications during such first solar eclipse season, the Satellite Orbital Performance Incentives shall be subject to adjustment in accordance with Article 13.2.10.

ARTICLE 11 – ACCEPTANCE INSPECTION FOR DELIVERABLE ITEMS OTHER THAN SATELLITE OR SPACE SEGMENT

11.1 <u>Deliverable Items of Hardware Other Than Satellite, Space Segment and/or GBBF Subsystem.</u>

- 11.1.1 Inspection. With respect to each Deliverable Item of hardware (including software integrated in hardware) other than the Satellite, the Space Segment and/or the GBBF Subsystem, Contractor shall perform such review and testing, if any, required by Exhibit A, Statement of Work or Exhibit G, Space Segment Statement of Work, as applicable. Upon Contractor's certification that Deliverable Item meets the requirements of this Contract and is available for delivery, Contractor shall ship or transfer the Deliverable Item to the location designated in Article 3.1. Purchaser shall perform an acceptance inspection within ten (10) days after such Deliverable Item arrives and is otherwise ready for inspection. The purpose of the acceptance inspection shall be to determine whether each such Deliverable Item meets applicable Performance Specification requirements as of the date of such delivery, as such requirements may have been modified pursuant to Article 11.7.
- 11.1.2 Acceptance inspection Results. Within fifteen (15) days after completion of the acceptance inspection pursuant to this Article 11.1 for such Deliverable Item, Purchaser shall notify Contractor in writing of the results of such acceptance inspection with respect to such Deliverable Item. In the event that such acceptance inspection demonstrates conformity of such Deliverable Item to the requirements of the applicable Performance Specification, such Deliverable Item shall be accepted by Purchaser for all GBBF Subsystem), and Purchaser's notice shall so state. In the event that Purchaser's acceptance inspection discloses any

such Deliverable Item to the applicable Performance Specification, Purchaser's notice shall identify each such non-conformance (with reference to the applicable requirement of the Performance Specification deemed not met), and Contractor shall correct or repair such non-conformance and resubmit such Deliverable Item for a second acceptance inspection in accordance with this Article 11. Such second acceptance inspection shall be conducted by Purchaser in accordance with this Article 11 to the extent the Purchaser deems relevant and necessary to verify the hardware (including integrated software) conforms to the requirements of the applicable Performance Specification. If Purchaser fails to provide notice within the fifteen (15) day period, Acceptance shall be deemed to have occurred with respect to such Deliverable Item.

11.2 GBBF Subsystem.

- 11.2.1 Notice and Shipment. Contractor shall notify Purchaser in writing, at least ten (10) days prior to shipment, that Contractor is prepared to ship the GBBF Subsystem to Purchaser's gateways and pointing beacon station locations for final installation and testing. Upon Contractor's certification that the GBBF Subsystem meets the requirements of this Contract and is available for delivery, Contractor shall ship the GBBF Subsystem (in its constituent parts as appropriate) to the Purchaser's gateways and pointing beacon station locations, as applicable, for final integration and testing in accordance with the applicable provisions of Exhibit G, Space Segment Statement of Work and Exhibit H, Space Segment Test Plan.
- 11.2.2 Acceptance. Contractor shall install and test the GBBF Subsystem as specified in the applicable Exhibits and certify to Purchaser in writing at the GBBF Subsystem acceptance test review meeting the results of the testing performed on the installed GBBF Subsystem and confirming that the GBBF Subsystem meets the applicable performance requirements of Annex B, Specification for the GBBF Subsystem, of Exhibit E, Space

Segment Performance Specification. Promptly (but in no event more than three (3) days) after the completion of the acceptance review meeting. Purchaser shall either accept the GBBF Subsystem in writing as complying with the applicable requirements of Annex B, Specification for the GBBF Subsystem, of Exhibit E, Space Segment Performance Specification ("Acceptance" for the GBBF Subsystem) or notify Contractor in writing as to how the GBBF Subsystem fails to comply with the applicable specified requirement(s). Upon receipt of Purchaser's notice of defects (if any), Contractor shall repair or replace any non-conforming unit(s) of the GBBF Subsystem, at Contractor's election and expense, and resubmit the GBBF Subsystem for Acceptance with certification, of the results of the corrective action taken. Promptly (but in no event more than three (3) days) after receipt of Contractor's written certification, and provided that Purchaser shall have determined in good faith that all applicable non-conformities have been corrected, Purchaser shall shall be deemed to have occurred with respect to the GBBF Subsystem.

If not earlier Accepted as provided in the foregoing provisions of this Article 11 or terminated for default pursuant to Article 23.1, the GBBF Subsystem shall [*]

11.3 <u>Deliverable Data</u>.

Purchaser shall, within ten (10) days of delivery by Contractor to the location designated in Article 3.1 of Deliverable Data requiring Purchaser approval pursuant to the Exhibit A, Statement of Work or Exhibit G, Space Segment Statement of Work, as the case may be, notify Contractor in writing that such

Deliverable Data has been accepted in accordance with the applicable Exhibit ("Acceptance" with respect to each such item of Deliverable Data), or that such Deliverable Data does not comply with the applicable requirements of the applicable Exhibit, identifying each such non-conformance (with reference to the applicable requirement of the Exhibit deemed not met). Contractor shall promptly correct any non-compliant aspect of such Deliverable Data identified in such notice from Purchaser, and re-submit it to Purchaser for a second acceptance inspection pursuant to this Article 11.3. The provisions of this Article 11.3 shall thereafter apply to the corrected Deliverable Data. If Purchaser fails to provide notice within such ten (10) day period, Acceptance shall be deemed to have occurred with respect to such Deliverable Data.

11.4 <u>Training</u>.

Acceptance of training, or any part thereof, required by Exhibit A, Statement of Work or Exhibit G, Space Segment Statement of Work, as the case may be, shall occur upon Contractor furnishing training, or such part thereof, to Purchaser in accordance with the Delivery schedule and in a condition conforming to the applicable Exhibit. Any training furnished to Purchaser shall be accompanied by written notice from Contractor specifying that portion of the training being furnished. If such training or part thereof is unacceptable, Purchaser shall timely notify Contractor in writing that the training, or part thereof, does not conform to the requirements of the applicable Exhibit, identifying each such non-conformance (with reference to the applicable requirement of the applicable Exhibit deemed not met). Contractor shall, at its expense, promptly correct such non-conformance and shall notify Purchaser that the corrections have been made. The provisions of this Article 11.4 shall thereafter apply to the corrected training.

11.5 Other Services.

Acceptance of other services provided hereunder (e.g., Launch Support Services and Mission Operations Support Services), or any part thereof, shall occur upon

Contractor furnishing such services, or such part thereof, to Purchaser in accordance with the Delivery schedule set forth in Article 3.1 and in a condition conforming to the requirements of this Contract. To the extent feasible, any such services furnished to Purchaser shall be accompanied by written notice from Contractor specifying that portion of the services being furnished. If such services or part thereof are unacceptable, Purchaser shall timely notify contractor that the services, or part thereof, do not conform to the requirements of the applicable Exhibit identifying each such non-conformance (with feasible and shall notify Purchaser that the corrections have been made. The provisions of this Article 11.5 shall thereafter apply to the corrected services.

11.6 Purchaser's Inspection Agents.

Purchaser may, upon giving prior written notice to Contractor, cause any Purchaser personnel, consultant or agent designated by Purchaser to observe or conduct the acceptance inspection pursuant to this Article 11 in whole or in part; provided, however, that the provisions of Article 7 and Article 8.5 shall writing in advance of such inspection.

11.7 Waivers and Deviations.

Contractor shall timely notify Purchaser in writing from time to time of any failure of the Work to meet the requirements of the Contract. With respect to any deviation or waiver from the Contract that Contractor shall in reasonable good failth believe that Purchaser should waive or approve or that Purchaser shall determine to waive or approve, Contractor shall submit to Purchaser a request for a waiver of, or deviation from, the provisions(s) of the Performance Specification applicable to the Deliverable Item, which request shall set forth in detail the extent of the impact on the Performance Specifications. A request

waiver or deviation shall be deemed granted only if it has been approved in writing by a duly authorized representative of Purchaser and expressly approving the modification to the Performance Specifications. Each such waiver or deviation approved by Purchaser shall be deemed an amendment to the Performance Specification for such Deliverable Item, permitting such waiver thereof, or deviation therefrom, effective on or after the date of such approval for such Deliverable Item. Purchaser shall consider each waiver or deviation request in good faith.

11.8 Inspection Costs Borne by Purchaser,

All costs and expenses incurred by Purchaser or its agents in the performance of its inspection rights under this Article 11, including travel and living expenses, shall be borne solely by Purchaser.

11.9 Warranty Obligations.

Except as expressly provided in Article 10.2.2 with respect to the Space Segment, in no event shall Contractor be released from any of its warranty obligations applicable to any Deliverable Item (other than a Satellite as set forth in Article 15) as a result of Purchaser's Acceptance pursuant to this

ARTICLE 12 – TRANSFER OF TITLE AND RISK OF LOSS

12.1 Satellite.

Except as provided in Article 21 or Article 23 or Article 24.4 (if the Storage option is exercised), title (free and clear of all liens and encumbrances of any kind) and risk of loss or damage to the Satellite to be delivered under this Contract shall pass from Contractor to Purchaser at the time of Intentional Ignition of the Launch Vehicle used for the Launch of such Satellite; provided, however, in the event of a Terminated Ignition for the Satellite, title to and risk of loss or damage to such Satellite shall revert to Contractor upon such Terminated Ignition and shall again pass to Purchaser upon the subsequent Intentional Ignition of the Launch Vehicle used for Launch of such Satellite.

UPON AND AFTER INTENTIONAL IGNITION OF THE LAUNCH VEHICLE FOR THE SATELLITE, UNLESS AND TO THE EXTENT OF A TERMINATED IGNITION AS PROVIDED HEREIN, CONTRACTOR'S SOLE FINANCIAL RISK, AND THE SOLE AND EXCLUSIVE REMEDIES OF PURCHASER OR ANY PARTY ASSOCIATED WITH PURCHASER, WITH RESPECT TO THE USE OR PERFORMANCE OF SUCH SATELLITE (INCLUDING WITH RESPECT TO ANY ACTUAL OR CLAIMED DEFECT CAUSED OR ALLEGED TO BE CAUSED AT ANY TIME BY CONTRACTOR'S OR ANY OF ITS SUBCONTRACTORS' NEGLIGENCE OF ANY DEGREE) SHALL BE AS SET FORTH IN ARTICLES 10.1.1 and 10.3, 13, 14 (AS REGARDS PROVIDING NOTIFICATION), 15.2, 19, AND 35.2, IN ALL CASES SUBJECT TO THE LIMITATION OF LIABILITY SET FORTH IN ARTICLE 30,

In the event of a Terminated Ignition, Contractor shall inspect the Satellite and provide Purchaser with a report on the condition of such Satellite along with a recommendation for repair or replacement, if any is required. Thereafter, Purchaser shall direct Contractor pursuant to Article 16.1 as to how to proceed with any required or desired repairs.

In the event of a Terminated Ignition, after Contractor re-acquires title and risk of loss to the Satellite pursuant to this Article 12.1, the provisions of Article 9.1.7 shall apply except as to any damage to such Satellite that may have occurred as a result of the Intentional Ignition followed by Terminated Ignition, the costs of which shall, as between the Parties, be the responsibility of the Purchaser.

12.2 <u>Deliverable Items Other Than the Satellite.</u>

Title (free and clear of all liens, claims and encumbrances of any kind) and risk of loss of or damage to each Deliverable Item (including, without limitation, the GBBF Subsystem and the DSS) other than the Satellite shall pass from Contractor to Purchaser upon Acceptance of such Deliverable Item pursuant to Article 11, except for Deliverable Data. With respect to Deliverable Data, Purchaser's rights in Deliverable Data are as set forth in Article 32. With respect to the Space Segment, title (free and clear of all liens, claims and encumbrances of any kind) and risk of loss of or damage shall pass from Contractor to Purchaser only as and to the extent title and risk of loss or damage pass from Contractor to Purchaser for each of the Satellite and the GBBF Subsystem.

12.3 <u>Liens</u>.

Any reference to Contractor being obligated to provide title to any Deliverable Item free and clear of liens or encumbrances shall be without regard to any liens or encumbrances created by Purchaser.

ARTICLE 13 - SATELLITE ORBITAL PERFORMANCE INCENTIVES AND SPACE SEGMENT ORBITAL PERFORMANCE INCENTIVES

13.1 General.

Except as otherwise provided in this Article 13, Contractor shall earn, and Purchaser shall pay Contractor to the extent earned, (i) the Satellite Orbital Performance Incentives for the Satellite over the Satellite Orbital Performance Incentive Period with respect thereto, and (ii) the Space Segment Orbital Performance Incentives for the Space Segment over the Space Segment Orbital Incentive Period with respect thereto, all in accordance with this Article 13. As used herein, "Satellite Orbital Performance Incentives" means with respect to the Satellite, an amount equal to [*] which may be earned by Contractor based on on—orbit performance of such Satellite as set forth in this Article 13. "Satellite Orbital Performance Incentive Period" means, with respect to the Satellite, the period commencing at (and including) 12:01 a.m. Greenwich Mean Time on the first day after Acceptance for such Satellite and ending on (but not including) the same date fifteen (15) years thereafter.

"Space Segment Orbital Performance Incentives" means one and only one of the following:

If at Space Segment Acceptance, the Space Segment is a Conforming Space Segment, then an amount up to [*]

- (b) If at Space Segment Acceptance, the Space Segment is a [*]
- (c) If at Space Segment Acceptance, the Space Segment is a Fully Degraded Space Segment, then the Contractor shall not be entitled to any Space Segment Orbital Performance Incentives (regardless of the condition of the Space Segment after Acceptance).

"Space Segment Performance Incentive Period" means, with respect to the Space Segment, the period commencing at (and including) 12:01 a.m. Greenwich Mean Time on the first day after Acceptance of the Space Segment and ending on but not including the [*]. "Applicable Space Segment Rate" means the applicable daily rate for Space Segment Orbital Performance Incentives determined in accordance with this Article 13.1.

All measurements, computations and analyses made pursuant to this Article 13 shall be made in accordance with good engineering practice, applying standards generally applicable in the aerospace industry.

On-board redundancy and/or spare components shall be taken into consideration to maintain service on the Satellite or the Space Segment, as applicable, and such use shall be deemed normal operating procedure for purposes of this Article 13 so long as the applicable criteria of Exhibit B, Satellite

Specification or Exhibit E, Space Segment Performance Specification, as applicable, are met.

13.2 <u>Earning Satellite and Space Segment Orbital Performance Incentives.</u>

13.2.1 Daily Rate of Satellite and Space Segment Orbital Performance Incentives.

A. Contractor shall earn Satellite Orbital Performance Incentives over the Satellite Orbital Performance Incentive Period at a daily rate of [*] for each day that the Satellite is a Satisfactorily Operating Satellite (the "Satellite Daily Rate"). "Satisfactorily Operating Satellite" means that all performance parameters for the Satellite meet the requirements of Exhibit B, Satellite Performance Specification. A "Less than Satisfactorily Operating Satellite" means a Satellite that does not meet all of the performance requirements of Exhibit B, Satellite Performance Specification.

B. Subject to Article 13.1, Contractor shall earn Space Segment Orbital Performance Incentives over the Space Segment Orbital Performance Incentive Period at the Applicable Space Segment Rate for each day that the Space Segment is a Satisfactorily Operating Space Segment (the "Space Segment Daily Rate"). "Satisfactorily Operating Space Segment" means that all performance parameters for the Space Segment meet the requirements of Exhibit E, Space Segment Performance Specification. A "Less than Satisfactorily Operating Space Segment" means a Space Segment that does not meet all performance requirements of Exhibit E, Space Segment Performance Specification.

- C. For each day during the Satellite Orbital Performance Incentive Period or the Space Segment Orbital Performance Incentive Period where the Satellite Daily Rate or the Space Segment Daily Rate, as the case may be, is not earned in whole or in part due to the operation of this Article 13, such portion of the Satellite Orbital Performance Incentive or the Space Segment Orbital Performance Incentive, as the case may be, shall be lost (subject to Article 13.2.6 as applicable).
- 13.2.2 Partial Incentives. For any day during (i) the Satellite Orbital Performance Incentive Period that the Satellite is a Less than Satisfactorily Operating Satellite, or (ii) the Space Segment Orbital Performance Incentive Period that the Space Segment is a Less than Satisfactorily Operating Space Segment, the Satellite Daily Rate or the Space Segment Daily Rate, as applicable, shall be reduced to an amount calculated in accordance with the tables and formulae set forth in this Article 13.2.2.

A. [*]

[*]

C. [*]

- D. The Satellite Orbital Performance Incentives earned for each day will be determined by application of the lowest "Maximum Incentives Earned" value derived from tables 1 through 4 above, as provided for in paragraphs (A), (B), or (C) above, applicable to such Satellite Orbital Performance Incentives.
- E. The Space Segment Orbital Performance Incentives earned for each day will be determined by application of the lowest "Maximum Incentives Earned" value derived from tables 1 and 3 above, as provided for in paragraphs (A), (B), or (C) above, applicable to such Space Segment Orbital Performance Incentives.
- 13.2.3 Orbital Storage. Except as provided in Article 13.2.5, if Purchaser places the Satellite in Orbital Storage, Contractor shall earn Space Segment Orbital Incentives with respect to the Space Segment and Satellite Orbital Performance Incentives with respect to the Satellite at the same Satellite and/or Space Segment Daily Rate as Contractor would be earning if the Satellite had not been put in Orbital Storage.

13.2.4 [*]

[+]

13.2.5 Complete Loss of Unearned Incentives, Contractor shall no longer be entitled to earn any remaining unearned Satellite Orbital Performance Incentives, if: (i) Purchaser permanently withdraws the Satellite from operational service (including use as an in-orbit spare or back-up) and gives Contractor written notice thereof (which notice shall be given within thirty (30) days of the expiration of the [*] period referred to below in this sentence), and Contractor has lost [*] or more of Satellite Orbital Performance Incentives by operation of Article 13.2.2 during the proceeding [*]; or (ii) Contractor has lost more than [*] of the Satellite Orbital Performance Incentives pursuant to operation of Article 13.2.2. With respect to loss of Satellite Orbital Performance Incentives pursuant to clause (i) in the preceding sentence, if Purchaser thereafter places the Satellite back in operational service (including use as an in-orbit spare or back-up) for itself or any other Person, Purchaser shall make a one-time payment to the Contractor for all Satellite Orbital Performance Incentives that the Satellite would have otherwise earned during the time it was withdrawn from service, plus interest calculated on such amount in accordance with Article 13.3. Thereafter, Contractor shall earn the remaining balance of the Satellite Orbital Performance Incentives in accordance with this Article 13.

13.2.6 <u>Purchaser Operation of the Satellite or Space Segment/Non-Contract Elements.</u> If, as a result of any act or omission on the part of Purchaser or Purchaser's representatives, consultants or subcontractors in the Launch, operation of, testing of, or communication with, the Satellite or the Space Segment, or as a result of other elements of Purchaser's system that are

not provided pursuant to this Contract, the Satellite or Space Segment, as the case may be, operates in a manner that is not in accordance with any requirements of Exhibit B, Satellite Performance Specification or Exhibit E, Space Segment Performance Specification, as applicable, Contractor shall continue to earn Satellite Orbital Performance Incentives with respect to such Satellite and/or Space Segment Orbital Performance Incentives with respect to such Space Segment at the rate that applied prior to the act or omission resulting in degraded performance (subject to later adjustments pursuant to Articles 13.2.2, 13.2.4, or 13.2.5 with respect to failures to meet the requirements of Exhibit B, Satellite Performance Specification or Exhibit E, Space Segment Performance Specification, as applicable, not resulting from any such act or omission of Purchaser's representatives, consultants or subcontractors or other elements of Purchaser's system that are not provided pursuant to this Contract).

13.2.7. In—Orbit Data and Performance Calculation. During the life of the Satellite, Contractor shall have access to applicable performance data of the Satellite, Space Segment and GBBF Subsystem for purposes of evaluating any degradation in the performance of the Satellite, Space Segment and/or GBBF Subsystem resulting in any potential loss of Satellite Orbital Performance Incentives or Space Segment Orbital Performance Incentives as promptly notified by Purchaser. The Parties shall agree on or before EDC plus eighteen (18) months as to the methodologies, precise equipment and calibrations to be used for taking and/or making measurements (incorporating measurement uncertainty), computations and analyses to determine whether any reduction in Satellite Orbital Performance Incentives or Space Segment Orbital Performance Incentives, that may be earned by the Contractor, pursuant to Article 13.2.2 is warranted.

- 13.2.8 Earn Back of Lost Incentives for Extended Use. Should Purchaser elect to continue the use of the Satellite that is operating beyond the Satellite Orbital Incentive Performance Period, Purchaser shall pay to Contractor Satellite Orbital Performance Incentives at the Daily Rate calculated in accordance with this Article 13.2 for each day during such period of continued use by Purchaser (the "Extended Incentives"). In no event shall Purchaser be obligated under this Article 13.2 to pay any Extended Incentives to the extent that the cumulative Extended Incentives paid hereunder exceed the amount, paid hereunder, when added to the total Satellite Orbital Performance Incentives paid during the Satellite Orbital Performance Incentives amount set forth in Article 13.1 above [*] (not including interest).
- 13.2.9 Total Loss. In the event that, at or after Intentional Ignition but prior to completion of IOT, the Satellite is deemed to be a Total Loss, Contractor shall not earn any Satellite Orbital Performance Incentives or Space Segment Orbital Performance Incentives, unless the Total Loss is primarily due to causes attributable to Purchaser or its suppliers or contractors (including the Launch Agency, but not including Contractor or its subcontractors), in which case the Contractor shall be entitled to earn (i) the Satellite Orbital Performance Incentives for the Satellite Orbital Performance Incentive Period; [*] (iv) the Space Segment Orbital Performance Incentive Period (provided Delivery and Acceptance of the GBBF Subsystem has occurred prior to the determination of Total Loss); and (v) the Space Segment Acceptance Payment set forth in Article 5.2.3C. Payments of the Satellite

Orbital Performance Incentives and the Space Segment Orbital Performance Incentives shall be made in accordance with Article 13.3.

13.2.10 Eclipse. In the event the IOT or such later testing in accordance with Article 10.3 shows non-compliance with the applicable performance requirements during an eclipse, then there shall be an equitable adjustment to the amounts Contractor may earn under Article 13 (considering without limitation the impact of operation during an eclipse) as may be agreed by the Parties or determined in accordance with Article 25.

13.3 Payment and Interest

Amounts due Contractor pursuant to Article 13 shall be paid as follows: (i) the first payment shall be due no later than the end of the first full month following Acceptance of the Satellite or the Space Segment, as applicable, and shall cover the first partial month of earned Satellite Orbital Performance Incentives or Space Segment Orbital Performance Incentives (as the case may be), if any, and (ii) thereafter, all subsequent payments shall be due no later than thirty (30) days following the end of each calendar month for the prior month's earned Satellite Orbital Performance Incentives or Space Segment Orbital Performance Incentives, if any, Included with payment, if any, Purchaser shall provide Contractor with a report of the Satellite Orbital Performance Incentives and/or Space Segment Orbital Performance Incentives due for the prior month. Purchaser shall pay interest on the Satellite Orbital Performance Incentives payable for any period calculated from the date of Acceptance of the Satellite and ending on the date the payment is made by Purchaser, at the annual interest rate of [*]. For the avoidance of doubt, interest shall not accrue on any Space Segment Orbital Performance Incentive.

ARTICLE 14 - CORRECTIVE MEASURES IN SATELLITE

If the data available from any satellite manufactured by Contractor (whether in-orbit or on the ground) shows that the performance of such satellite deviates materially during the life thereof from that specified in the applicable performance specification for such satellite, Contractor shall (i) provide Purchaser prompt written notice thereof, provided Contractor shall not be required to disclose to Purchaser information that is confidential to a customer of Contractor other than Purchaser and (ii) at Contractor's sole cost, take appropriate corrective measures, if any, with the unlaunched Satellite (including any option satellite) so as to eliminate therefrom the causes of such material deviation.

ARTICLE 15 - WARRANTY

15.1 <u>Terms and Period of Warranty.</u>

15.1.1 Satellite. Notwithstanding any prior inspection, Contractor warrants that the Satellite shall be free from defects in materials and workmanship and shall have been manufactured and will perform in conformity with the requirements of the Contract including Exhibit B, Satellite Performance Specifications (as may be modified or deviated from pursuant to Article 9.3). After successful completion of the SPSR, but prior to Intentional Ignition, Contractor's only liability under the preceding sentence shall be as and to the extent set forth in Article 9.1.7 hereof. NOTWITHSTANDING THE ABOVE, AFTER INTENTIONAL IGNITION OF THE LAUNCH VEHICLE FOR THE SATELLITE, UNLESS AND TO THE EXTENT OF A TERMINATED IGNITION AS PROVIDED IN ARTICLE 12.1, NEITHER CONTRACTOR NOR ITS SUPPLIERS OR AGENTS AT ANY TIER SHALL INCUR ANY LIABILITY WHATSOEVER WITH RESPECT TO THE SATELLITE'S DESIGN, WORKMANSHIP, CONFORMITY TO SPECIFICATION OR IN-ORBIT PERFORMANCE, INCLUDING ANY ASSISTANCE OR ADVICE (ACTUAL OR ATTEMPTED) PROVIDED OR OMITTED AS CONTEMPLATED BY ARTICLES 15.2 AND 35.2 HEREOF, ARISING FROM ANY CAUSE OR LEGAL THEORY WHATSOEVER, INCLUDING NEGLIGENCE OF ANY DEGREE, WHETHER ARISING BEFORE OR AFTER INTENTIONAL IGNITION, EXCEPT AS OTHERWISE SPECIFICALLY PROVIDED IN ARTICLES 13 AND 19 HEREIN AND EXCEPT TO PROVIDE THE SERVICES SET FORTH IN ARTICLES 16 AND 36.2. NOTWITHSTANDING THE FOREGOING, CONTRACTOR SHALL CONTINUE TO PROVIDE ANY NOTICES REQUIRED UNDER ARTICLE 14.

15.1.2 Deliverable Hardware Other Than the Satellite and GBBF Subsystem.

- A. Contractor warrants that each Deliverable Item of hardware (other than the Satellite, which is treated separately in Article 15.1.1 above, and the GBBF Subsystem, which is treated as part of the Space Segment in accordance with Article 15.1.3 below), delivered under this Contract and Accepted pursuant to Article 11 hereof, shall be manufactured in conformity with the requirements of the Performance Specification applicable to such Deliverable Item (as may be waived pursuant to Article 11.7) and will be free from defects in materials and workmanship during the period commencing on the date of Acceptance of the applicable Deliverable Item and ending on the [*] thereafter, except that the warranty period for the DSS shall be as set forth in Exhibit J.
- B. Any non-conformance or defect discovered in any Deliverable Item of hardware subject to this Article 16.1.2 during the relevant warranty period set forth in subsection (A) above and notified to Contractor shall be remedied by Contractor at Contractor's expense by repair or replacement of the defective component (at Contractor's election) and, with respect to the DSS, as otherwise set forth in Exhibit J. For such Deliverable Item, Contractor shall determine if repair or replacement is required to be performed at Contractor's plant or may be conducted at another location. If required, Purchaser shall ship to Contractor's designated facility such Deliverable Item or defective component (as commercially reasonable). Contractor shall be responsible for the cost of shipment (including transportation and transit insurance) to such facility in accordance with its standard commercial practice (including any taxes and/or duties) of such Deliverable Item or defective component, and the cost of packing, return shipment (including transportation and transit insurance) in accordance with its standard commercial practice of any such Deliverable Item or defective component once repaired or replaced to Purchaser at the

location designated therefor by Purchaser. Risk of loss for such Deliverable Item or defective component shall transfer to Contractor upon delivery of such Deliverable Item or defective component to the shipping carrier by Purchaser, and risk of loss shall transfer to Purchaser for any such Deliverable Item or defective component once repaired or replaced pursuant to this Article 15.1.2(B) upon receipt thereof by Purchaser at the location designated by Purchaser. If Contractor fails to use reasonable best efforts to repair or replace such Deliverable Item or defective component within a reasonable period of time after notification from Purchaser, Purchaser may, by contract or otherwise, repair or replace such Deliverable Item or defective component and Contractor shall be liable for the reasonable cost thereof.

15.1.3 Space Segment. Without limiting Contractor's obligations under Articles 15.1.1 and 15.1.2;

A. Contractor warrants that, from Space Segment Acceptance (or with respect to the GBBF Subsystem, if earlier, Acceptance of the GBBF Subsystem) through the [*] of such Space Segment Acceptance (as such warranty period may be extended), the Space Segment shall perform in conformity with those requirements of Exhibit E, Space Segment Performance Specification, that are tested and/or validated during Space Segment IOT in accordance with Exhibit H, Space Segment Test Plan (as such requirements may be waived or deviated from for purposes of Space Segment Acceptance under Article 10.1.2), such that the baseline for the warranty shall be the Space Segment's condition as of Space Segment Acceptance; provided, however, that such warranty baseline shall be subject to revision to reflect improved performance pursuant to Article 15.1.3(C) below.

- B. Contractor's only liability under the preceding subsection (A) shall be with respect to the GBBF Subsystem and Satellite as follows:
- With respect to the GBBF Subsystem, the warranty set forth in Article 15.1.3(A) is based on the condition of the GBBF Subsystem as Accepted, subject to revision of such condition to reflect improved performance pursuant to Article 15.1.3(C). Any non-conformance or defect discovered in the GBBF Subsystem during the warranty period set forth in subsection (A) above and notified to Contractor shall be remedied by Contractor at Contractor's expense by repair or replacement of the defective component or by other appropriate corrective measures with respect to the GBBF Subsystem, including but not limited to GBBF hardware and/or software modifications, as set forth in Section 2.3.7 of Exhibit G, Space Segment SOW (at Contractor's election). In fulfilling its obligations under this Article 15.1.3(B)(1) Contractor shall be responsible to correct only for defects or non-conformances to the applicable requirements of Annex B, Specification for the GBBF Subsystem, of Exhibit E, Space Segment Performance Specification (to the extent of GBBF Subsystem compliance with such requirements upon GBBF Subsystem Acceptance under Article 11), subject to revision of such warranty baseline requirements to reflect improved performance pursuant to Article 15.1.3(C) below. Contractor shall determine whether such repair or replacement shall be performed at Contractor's plant. If necessary, Purchaser shall ship to Contractor's designated facility the defective component(s). Contractor shall be responsible for the cost of shipment (including transportation and transit insurance) to the facility in accordance with its standard commercial practice (including any taxes and/or duties) of such defective component(s), and the cost of packing,

return shipment (including transportation and transit insurance) in accordance with its standard commercial practice of any such defective component(s) once repaired or replaced to Purchaser at the location designated therefor by Purchaser. Risk of loss for such defective component(s) shall transfer to Contractor upon delivery of such defective component(s) to the shipping carrier by Purchaser, and risk of loss shall transfer to Purchaser for any such defective component(s) once repaired or replaced pursuant to this Article 15.1.3(B)(1) upon receipt thereof by Purchaser at the location designated by Purchaser. If Contractor fails to use reasonable best efforts to repair or replace such defective component(s) within a reasonable period of time after notification from Purchaser, Purchaser may, by contract or otherwise, repair or replace such defective component(s) and Contractor shall be liable for the reasonable cost thereof.

(2) If the GBBF Subsystem does not exhibit a defect or non-conformance subject to Article 15.1.3(B)(1), and performance of the Space Segment nonetheless degrades with respect to the Space Segment performance baseline for Space Segment Acceptance (as may be adjusted in accordance with Article 15.1.3(C)), Contractor shall take reasonable steps to analyze and recommend corrective action as set forth in Section 2.3.7 of Exhibit G, Space Segment SOW. Purchaser shall provide reasonable investigation support for possible causes in Purchaser furnished items, such as equipment in the gateway. If and to the extent such analysis and recommended corrective action pertain to the Satellite, the Parties shall proceed as set forth in Article 15.1.3(B)(3). If and to the extent such analysis and recommended corrective action pertain to the GBBF Subsystem and/or other elements of the Space Segment (not including the Satellite), Contractor shall notify Purchaser and

describe the proposed corrective action together with associated costs (as applicable) and terms for proceeding with the action. Purchaser shall approve or disapprove the proposed action and associated costs (as applicable) and terms within a reasonable time. If Purchaser authorizes the proposed action, Contractor shall proceed therewith: (a) at Contractor's expense if the degradation in Space Segment performance is primarily attributable to the GBBF Subsystem and/or other elements of the Space Segment (not including the Satellite), or (b) at Purchaser's expense (but only to the extent set forth in Article 15.1.3(B)(3)) if such degradation is primarily attributable to the Satellite, or (c) at Purchaser's expense if such degradation is primarily attributable to Purchaser-furnished equipment, or (d) in any other cases, as otherwise agreed by the Parties based on an equitable allocation of risk in the circumstances. Annex B, Specification for the GBBF Subsystem, of Exhibit E, Space Segment Performance Specification, shall be modified to incorporate any changes thereto required to reflect the corrective actions approved by Purchaser hereunder.

(3) Without limiting Contractor's obligations under Article 15.2, in the event that, after the IOT Complete Date, for causes attributable to Contractor (or any subcontractor, agent or representative of Contractor), the Satellite fails to meet the applicable requirements of Exhibit B, Satellite Performance Specification (as such requirements may have been waived or deviated from for purposes of Satellite Acceptance upon the IOT Complete Date), and the performance of the Space Segment degrades with respect to the Space Segment performance baseline for Space Segment Acceptance, Contractor shall take reasonable steps to analyze and recommend corrective action as set forth in Section 2.3.7 of Exhibit G, Space Segment SOW. The Contractor's corrective action

include and is limited to analysis, diagnosis, testing, GBBF procedures, coefficient updates, Satellite configuration commands, and Satellite operating procedures. Contractor shall notify Purchaser and describe the proposed corrective action together with associated costs (as applicable) and terms for proceeding with the action. Purchaser shall approve or disapprove the proposed action and associated costs (as applicable) and terms within a reasonable time. If Purchaser authorizes the proposed action, Contractor shall proceed therewith: (a) at Contractor's expense, subject to a cumulative total annual sum of [*] for the first year of the warranty and [*] for each extension of the warranty pursuant to Article 24.3, for such corrective actions, and (b) at Purchaser's expense after the monetary threshold in clause (a) immediately above is met, on the basis of [*] to be invoiced and paid in accordance with Article 5. If for any reason the corrective action is unsuccessful, and, as a result thereof, the Satellite suffers any loss, including but not limited to degraded performance, Purchaser's sole and exclusive remedies with respect to such loss and any consequences therefrom shall be those set forth in Articles 13 and 35.2 as applicable, subject to the conditions and limitations of such Articles and in all cases subject to the limitation of liability stated in Article 15.1 above and Article 30.

(4) The warranty under this Article 15.1.3 with respect to the Space Segment shall terminate upon payment in full of the Space Segment Warranty Termination Payment.

[*]

(2) Contractor shall notify Purchaser of corrective actions identified as set forth in the immediately preceding paragraph, and the Parties shall agree on the appropriate implementation, or Purchaser may direct Contractor to proceed with the recommended action. Contractor shall proceed therewith: (a) at Contractor's

expense if the deficiencies in Space Segment performance are primarily attributable to the GBBF Subsystem and/or other elements of the Space Segment (not including the Satellite), or (b) at Purchaser's expense (but only to the extent set forth in Article 15.1.3(B)(3)) if such deficiencies are primarily attributable to the Satellite, or (c) at Purchaser's expense if such deficiencies are primarily attributable to Purchaser-furnished equipment, or (d) in any other cases, as otherwise agreed by the Parties based on an equitable allocation of risk in the circumstances.

[*]

(4) Except as set forth in the applicable Articles referenced in Article 10.2.2, in no event shall Contractor incur liability related to the use, condition or performance of the Space Segment arising from its actions pursuant to this Article 15.1.3(C), [*]

[*]

- 15.1.4 <u>Training and Services.</u> Contractor warrants that the training and other services it provides to Purchaser pursuant to this Contract will conform to reasonable industry standards of the commercial aerospace and satellite communications industry practice for work similar in type, scope, and complexity to the applicable Work at the time such training or other services are provided. In the event Contractor breaches this warranty, as Purchaser's sole remedy, Contractor shall apply reasonable efforts to correct the deficiencies in the provision of such training and other services where it is practicable to do so (e.g., cannot correct deficiencies in or redo Launch Support Services from and after Launch).
- 15.1.5 <u>Third-Party Warranties</u>. Without in any way relieving Contractor of its obligations hereunder, Contractor shall assign to Purchaser to the extent assignable, or make available to Purchaser to the extent permitted, the benefit of all third party warranties Contractor has from any other party with respect to any Deliverable Item.
- 15.1.6 <u>Disclaimer</u>. EXCEPT AND TO THE EXTENT EXPRESSLY PROVIDED IN ARTICLES 15.1.1, 15.1.2, 15.1.3, 15.1.4 AND 15.1.5 CONTRACTOR HAS NOT MADE NOR DOES IT HEREBY MAKE ANY REPRESENTATION OR WARRANTY, WHETHER WRITTEN OR ORAL, EXPRESS OR IMPLIED, INCLUDING WITHOUT LIMITATION, ANY WARRANTY OF DESIGN, OPERATION, CONDITION, QUALITY, SUITABILITY OR MERCHANTABILITY OR FITNESS FOR USE OR FOR A PARTICULAR PURPOSE, ABSENCE OF LATENT OR OTHER

[*] CONFIDENTIAL TREATMENT REQUESTED BY ICO GLOBAL COMMUNICATIONS (HOLDINGS) LIMITED.

DEFECTS, WHETHER OR NOT DISCOVERABLE, WITH REGARD TO THE SATELLITE OR ANY OTHER DELIVERABLE ITEM.

15.2 <u>Satellite Anomalies</u>.

At no cost to Purchaser, Contractor shall investigate any Satellite Anomaly in the Satellite delivered on-orbit arising during the life of such Satellite and known to it or as notified in writing by Purchaser and undertake anomaly resolution support services in accordance with Section 2.3.9 (Standard On-Orbit Support) of Exhibit A, Statement of Work. If for any reason any such Satellite Anomaly cannot be or is not corrected (after reasonable efforts to effect a correction) as set forth in the immediately preceding sentence and as a result thereof, such Satellite suffers any loss, Purchaser's sole and exclusive remedies with respect to such loss and any consequences therefrom shall be those set forth in Articles 13 and 35.2 as applicable, subject to the conditions and limitations of such Articles and in all cases subject to the limitation of liability stated in Article 15.1 above and Article 30. To the extent the occurrence of a Satellite Anomaly is due to causes not attributable to Contractor (including any subcontractor, agent or representative of Contractor), Purchaser shall pay Contractor the Actual Costs reasonably incurred by Contractor associated with such anomaly services, plus a markup of [*], which shall be invoiced and paid pursuant to the provisions of Article 5. To the extent the causes are attributable to Contractor (including any subcontractor, agent or representative of Contractor, agent or representative of Contractor), such costs shall be borne by Contractor.

Use Conditions Not Covered by Warranty. 15.3

With respect to Deliverable Items other than the Satellite, the warranty under this Article 15 shall not apply to the extent adjustment, repair or parts replacement is required as a result, directly or indirectly, of accident, unusual physical or electrical stress, negligence, misuse, failure of environmental control prescribed in operations and maintenance manuals, repair or alterations by any party other than Contractor (including any subcontractor, agent or representative of Contractor), or by causes other than normal and ordinary use. The warranty provided pursuant to this Article 15 is conditioned upon Contractor being given access, if required, to Deliverable Items delivered at Purchaser's facility in order to accomplish any repair or replacement thereof. If the defect repaired or remedied by Contractor is not covered by the warranty provided pursuant to this Article 15, Purchaser shall pay Contractor the Actual Costs reasonably incurred by Contractor associated with such repair or remedy, plus a markup of [*], which shall be invoiced and paid pursuant to the provisions of Article 5. Contractor makes no warranty with respect to the performance of any Launch Vehicle.

[*] CONFIDENTIAL TREATMENT REQUESTED BY ICO GLOBAL COMMUNICATIONS (HOLDINGS) LIMITED. ARTICLE 16— CHANGES AND STOP WORK

16.1 Changes Requested by Purchaser,

16.1.1 Purchaser may at any time in writing request a change to the Work in whole or in part including changes to any Deliverable Item or service (as may be modified from time to time, a "Change Request").

Contractor shall respond in writing to a Change Request within thirty (30) days after receipt thereof, which response shall include Contractor's reasonable good faith proposal of the changes to the Contract that would be caused by such Change Request including, without limitation, the estimated increase or decrease to the Firm Fixed Price [*] the impact on the delivery schedule and other changes to the Contract necessitated thereby. The proposal shall include reasonably sufficient detail setting forth the basis used by Contractor (including a detailed description of the work to be performed, break down showing estimated labor and major material costs together with such other information/explanation as Purchaser may reasonably request).

16.1.2 If Purchaser desires to implement the Change Request, Purchaser and Contractor shall negotiate in good faith and in a timely manner the changes, if any, to the Contract caused by implementation of the Change Request. Except as provided in Article 16.1.3, any changes made pursuant to this Article 16.1 shall be formalized by the execution of an amendment to this Contract in accordance with Article 37.5. Except as set forth in such amendment, the terms of the

- 16.1.3 If the Parties are not able to agree on the changes to the Contract necessitated by such Change Request, then within ten (10) days of a request from either Party, each Party shall deliver to the other a good faith offer of the changes to the Contract (the "Purchaser's Offer" and "Contractor's Offer"). Thereafter, Purchaser has the right to direct Contractor to implement the requested changes. In such event:
- (a) the Parties shall continue to negotiate in good faith as to the terms of any changes to the Contract;
- (b) either Party shall have the right to invoke the dispute resolution procedures of Article 25; and
- (c) Purchaser shall pay to Contractor the sums as and when provided in the Purchaser's Offer and pay any difference between the Purchaser's Offer and Contractor's Offer relative to amounts as and when due into an interest bearing escrow account, the establishment and terms and conditions of which shall be subject to the reasonable mutual agreement of the Parties. Upon the reaching of a resolution (either through mutual agreement of the parties or through operation of Article 25), this Contract shall be amended accordingly pursuant to Article 37.5, the escrow shall be closed and any funds contained therein
- 16.1.4 If Purchaser directs that certain goods or services be provided by Persons other than Contractor, Contractor agrees to provide reasonable cooperation with such Persons, subject to Contractor's reasonable requirements concerning access, confidentiality and compliance with law, as set forth in this Contract.

16.2 Stop Work Order

Purchaser shall have the right at any time (and from time to time) to direct Contractor to stop the Work. To exercise the right hereunder, Purchaser shall deliver a written notice to stop the Work (the "Notice to Stop Work"). Upon delivery of the Notice to Stop Work (and during the pendancy of the stop Work): (a) Purchaser's liability under this Contract shall be limited to such liability as Purchaser would have if Purchaser had delivered a Notice of necessary or prudent by Contractor in order to minimize any losses Contractor would suffer if the Work was not restarted, including terminating and reaching good faith settlements with third parties performing any portion(s) of the Work. Upon request from Purchaser, Contractor shall submit to Purchaser increase the liability of Purchaser. If after delivery of a Notice to Stop Work Purchaser elects thereafter to restart the Work, then such request shall be Work within twelve (12) months after delivery to Contractor of the Notice to Stop Work, then Contractor shall have the right, upon written notice to this Contract in whole in accordance with Article 21.1 effective as of the date of the Notice to Stop Work. In the event Purchaser fails to elect within ten effective as of the date of the Notice to Stop Work. In the event Purchaser fails to elect within ten effective as of the date of the Notice to Stop Work.

16.3 Changes Requested by Contractor.

Contractor may at any time request a change in the Work. Any proposed change shall contain a written proposal identifying such change, the reasons therefore and the impact thereof on the price, Delivery schedule, or other terms of this Contract. Purchaser will attempt to respond to such change request from Contractor within thirty (30) days of receipt. If the Parties reach an agreement on such change request in writing, it shall be incorporated into the Contract by the execution of an amendment to this Contract in accordance with Article 37.5. If the Parties are unable to reach an agreement in writing on the proposed change request, then such request shall be deemed rejected.

ARTICLE 17 - FORCE MAJEURE

17.1 Force Majeure Defined.

17.1.1 Subject to Articles 5.2.3 and 23.3, Contractor shall not be responsible to Purchaser for late Delivery, delay of the final completion date or nonperformance of its contractual obligations due to Force Majeure. Force Majeure shall be any event beyond the reasonable control of a Party or its suppliers and subcontractors and shall include, but not be limited to: (1) acts of God; (2) acts of a public enemy; (3) acts of a government in its sovereign or contractual capacity (including any action or inaction affecting the import or export of items); (4) war and warlike events; (5) catastrophic weather conditions such as hurricanes, tornadoes and typhoons; (6) fire, earthquakes, floods, epidemics, quarantine restrictions, (7) strikes, lockouts and other industrial disputes (other than at the facilities of Contractor), (8) sabotage, riot and embargoes; (9) technical problems generally affecting the commercial telecommunications industry, and (10) other unforeseen and extraordinary events ("Force Majeure"); provided written notice is of such Force Majeure and further provided such Force Majeure is not caused by the failure of Contractor or its suppliers or subcontractors to perform their obligations under this Contract. Notwithstanding the foregoing, in no event will financial hardship or insolvency of Contractor or any of its respective Affiliates constitute Force Majeure.

Any notice to be provided by Contractor, as required by the preceding provisions, shall include a detailed description of the performance of Contractor known to be affected by such delay as well as Contractor's plans for minimizing the effects of such event upon the performance of its obligations under the Contract. In all cases, Contractor shall use

reasonable best efforts to avoid or minimize and/or work around such delay through the implementation of any work—around plans, alternate sources, or other means Contractor may reasonably utilize to minimize a delay in performance of the Work. Contractor shall also provide Purchaser prompt written notice when the event constituting Force Majeure as to Contractor appears to have ended and shall during the period of Force Majeure provide such other reports as Purchaser may reasonably request (including, without limitation, estimates when the matter is to be rectified and the steps made to meet the applicable Contract requirements thereafter).

- 17.1.2 In the event Purchaser disputes any Force Majeure, Purchaser shall inform Contractor in writing within ten (10) Business Days from the date of receipt of written notice of the event purportedly constituting the Force Majeure and, if the Parties have not be resolved the dispute within ten (10) Business Days of Contractor's receipt of such written notice from Purchaser, the dispute shall be resolved pursuant to Article 25.
- 17.2 Adjustment to Delivery Schedule and Other Affected Terms.
- 17.2.1 Subject to Articles 5.2.3 and 23.3, upon the occurrence of Force Majeure that causes a delay in Contractor's performance of its obligations hereunder, an Equitable Adjustment shall be negotiated in the Delivery schedules of this Contract and the Space Segment Acceptance Deadline, as applicable, affected by Force Majeure. Such adjustment in Delivery schedules and/or Space Segment Acceptance Deadline shall not affect any other provisions of this Contract (including the Firm Fixed Price).
- 17.2.2 Any adjustment made pursuant to this Article 17.2 shall be formalized by the execution of an amendment to this Contract in accordance with Article 37.5.

ARTICLE 18 - PURCHASER DELAY OF WORK

If the performance of all or any part of the Work required of Contractor under this Contract is delayed or interrupted by Purchaser's failure to perform its contractual obligations hereunder within the time specified in this Contract or within a reasonable time if no time is specified, or by an act or failure to act of Purchaser that unreasonably interferes with and delays Contractor's performance of its obligations under this Contract ("Purchaser Delay"), Contractor shall promptly give Purchaser written notice thereof. In the event of a Purchaser Delay, Contractor shall cooperate with Purchaser to develop a plan using best reasonable efforts to avoid or minimize and/or work around such delay (including work—around plans, use of alternate sources, or other means to minimize the delay). Within fifteen (15) working days of Contractor's issuance of its notice of Purchaser Delay, Contractor shall provide Purchaser with a written estimate of the impacts to the Contract caused by such Purchaser Delay. Subject to such notice, and issuance of the written estimate as provided immediately above, Contractor shall be entitled to an Equitable Adjustment for Purchaser Delay. Any Equitable Adjustment made pursuant to this Article 18 shall be incorporated into this Contract by the execution of an amendment in accordance with Articles 16 and 37.5. At Purchaser's written request pursuant to Article 16, Contractor shall use best reasonable efforts to avoid or minimize and/or work around any delay resulting from such Purchaser Delay through the implementation of any work—around plans, alternate sources, or other means Contractor may utilize or expect to utilize to minimize a delay in performance of the Work.

ARTICLE 19 - INTELLECTUAL PROPERTY INDEMNITY

19.1 <u>Indemnification</u>.

Contractor, at its own expense, hereby agrees to defend and to indemnify and hold harmless Purchaser and its Affiliates, and their respective officers, directors, employees, shareholders, and agents and their successors and assigns from and against any claim or suit based on, arising from, or in connection with any allegation that the Work or the manufacture of any Deliverable Item or any part thereof or the normal intended use, lease or sale of any Deliverable Item or any part thereof infringes any third party Intellectual Property Right ("Intellectual Property Claim"), and shall pay any royalties and other liabilities adjudicated (or provided in settlement of the matter) to be owing to the third party claimant as well as costs and expenses incurred in defending or settling (including court costs and reasonable attorneys' fees and disbursements, costs of investigation, expert fees, litigation, settlement, judgment, interest and penalties) such intellectual Property Claim. Contractor's obligations under this Article 19.1 shall be subject to the indemnification process set forth in Article 20.3. This Article 19.1 shall be subject to the limitations set forth in Article 32.6.

19.2 Infringing Equipment

If the manufacture of any Deliverable Item or the normal intended use, lease or sale of any Deliverable Item under this Contract is enjoined as a result of an Intellectual Property Claim or is otherwise prohibited, Contractor shall at its option and expense (i) resolve the matter so that the injunction or prohibition no longer pertains, (ii) procure for Purchaser or its permitted successors and/or assigns the right to use the infringing item and/or (iii) modify the infringing item so that it becomes non-infringing while remaining in compliance with the Performance Specification (as may be modified or waived pursuant to Article 9.3 and/or 11.7). Purchaser shall reasonably cooperate with Contractor to mitigate or remove any infringement. If Contractor is unable to accomplish (i), (ii) or (iii) as stated above within (x) one hundred twenty (120) days if the injunction or

prohibition relates to the Satellite, or (y) ninety (90) days if the injunction or prohibition relates to any other Deliverable Item (including the Space Segment, other than the Satellite), Purchaser shall have the right, notwithstanding any other provision herein, to terminate this Contract in whole or in part, with respect to such Deliverable Item (and any other Deliverable Item whose utility has been adversely impacted by Purchaser's loss of use of the infringing Deliverable Item), return such Deliverable Item(s) to Contractor (in space, with respect to an in-orbit Satellite), and within thirty (30) days of such return, Contractor shall refund the price paid for such terminated Deliverable Item(s) less a reasonable allowance for use and depreciation. This Article 19.2 shall survive the termination of this Contract as to Deliverable Items Accepted or otherwise retained and paid for prior to or promptly following such termination.

19.3 Combinations and Modifications.

Contractor shall have no liability under this Article 19 for any Intellectual Property Claim to the extent arising directly from (i) use of any Deliverable Item in combination with other items not provided, recommended, or approved by Contractor or its suppliers or subcontractors or (ii) modifications of any Deliverable Item after Delivery by a Person other than Contractor (or its suppliers or subcontractors acting through Contractor) unless (A) in the case of a Deliverable Item other than the Satellite, such modification was made by such other Person due to Contractor's failure to perform a contractual obligation hereunder (except where Contractor believes such modification would result in an infringement of a third party's rights) or (B) is directed in writing by Purchaser to which Contractor has taken written exception including by way of disclaimer.

19.4 Sole Remedies.

The remedies set forth in this Article 19 are Purchaser's sole and exclusive remedies for or related to any Intellectual Property Claim.

ARTICLE 20 - INDEMNITY FOR PERSONAL INJURY AND PROPERTY DAMAGE

20.1 <u>Contractor's Indemnity.</u>

Contractor, at its own expense, shall defend, indemnify and hold harmless Purchaser and its Affiliates and their respective directors, officers, employees, shareholders, and agents from and against any losses, damages, and other liabilities, adjudicated (or provided for in settlement of the matter) to be owing to the claimant as well as costs and expenses (including court costs, and reasonable attorneys' fees and disbursements, costs of investigation, expert fees, litigation, settlement, judgment, interest and penalties) (collectively, "Losses") based on, arising from or in connection with any third party claims for injury to persons or property damage, but only if such Losses were caused by, or resulted from, a negligent act or omission or willful misconduct of Contractor, or no indemnity obligations pursuant to this Article 20 for any Losses with respect to the Satellite after Intentional Ignition of the Launch Vehicle for such omission of Contractor or its employees, representatives, contractors or subcontractors at any tier (including suppliers of any kind) prior to Intentional Ignition.

20.2 Purchaser's Indemnity.

Purchaser, at its own expense, shall defend, indemnify and hold harmless Contractor and its Affiliates, and their respective directors, officers, employees, shareholders and agents, from and against any Losses based on, arising from or in connection with any third party claims for injury to persons or property damage but only if such Losses were caused by, or resulted from, a negligent act or omission or willful misconduct of Purchaser or its employees or

20.3 <u>Indemnification Process.</u>

In connection with any claim for indemnification under Article 20.1 and Article 20.2 the Parties agree as follows:

20.3.1 The Party seeking indemnification shall promptly advise the other Party in writing of the filling of any suit or of any written or oral claim for which it seeks indemnification upon receipt thereof and shall provide the other Party, at its written request, with copies of all documentation relevant to such suit or claim. Failure to so notify the indemnifying Party shall not relieve the indemnifying Party of its obligations under this Contract except to the extent it can demonstrate that it was prejudiced by such failure. Within fifteen (15) days following receipt of written notice from the Party seeking indemnification relating to any claim, but no later than a reasonable time before the date on which any response to a complaint or summons is due, the indemnifying Party shall notify the Party seeking indemnification in writing if the indemnifying Party elects to assume control of the defense or settlement of that claim (a "Notice of Election") when not contrary to the governing rules of procedure. A Notice of Election shall require the indemnifying Party to assume the defense in full and without reservation of rights.

20.3.2 If the indemnifying Party delivers a Notice of Election relating to any claim within the required notice period, so long as it is actively defending such claim, the indemnifying Party shall be entitled to have sole control over the defense and settlement of such claim; provided that (i) the Party seeking indemnification shall be entitled to participate in the defense of such claim and to employ counsel (reasonably acceptable to the indemnifying Party) at its own expense to assist in the handling of such claim provided there is no conflict of interest and that such participation would not adversely affect the conduct of the proceedings; (ii) where the Party seeking indemnification is so represented, the indemnifying Party shall keep

counsel of the Party seeking indemnification informed of each step in the handling of any such claim; and (iii) the Party seeking indemnification shall provide, at the indemnifying Party's request and expense, such assistance and information as is available to the Party seeking indemnification for the defense and settlement of such claim and (iv) the indemnifying Party shall notify the Party seeking indemnification before entering into any non-monetary settlement of such claim or ceasing to defend against such claim. After the indemnifying Party has delivered a Notice of Election relating to any claim in accordance with the preceding paragraph, the indemnifying Party shall not be liable to the Party seeking indemnification for any legal expenses incurred by the Party seeking indemnification in connection with the defense of that claim. In addition, the indemnifying Party shall not be required to indemnify the Party seeking indemnification for any amount paid or payable by the Party seeking indemnification in the settlement of any claim for which the indemnifying Party has delivered a timely Notice of Election if such amount was agreed to without the prior written consent of the indemnifying Party. The indemnifying Party may not settle any claim on behalf of the indemnified Party except claims solely for monetary damages paid by the indemnifying Party in full.

20.3.3 If the indemnifying Party does not deliver a Notice of Election relating to any claim within the required notice period or falls actively to defend such claim, the Party seeking indemnification shall have the right to defend and/or settle the claim in such manner as it may deem appropriate, at the cost and expense of the indemnifying Party. Provided that the Party seeking indemnification acts in good faith, it may settle such claim on any terms it considers appropriate under the circumstances without in any way affecting its right to be indemnified hereunder. The indemnifying Party shall promptly reimburse the Party seeking indemnification for all such costs and expenses.

20.4 Waiver of Subrogation.

Each Party shall use best reasonable efforts to obtain a waiver of subrogation and release of any right of recovery against the other Party and its Affiliates, contractors and subcontractors at any tier (including suppliers of any kind) and their respective directors, officers, employees, shareholders and agents, that are involved in the performance of this Contract from any insurer providing coverage for the risks such Party has agreed to indemnify against under this Article 20.

ARTICLE 21 - TERMINATION FOR CONVENIENCE

- 21.1 Right to Terminate. Purchaser may at any time terminate this Contract without cause, in whole or in part, by giving Contractor written notice of termination ("Notice of Termination for Convenience"); provided, however, Purchaser may not terminate this Contract as to a Satellite after Intentional Ignition of the Launch Vehicle for Launch of such Satellite and provided further that, absent the consent of Contractor, if Purchaser shall terminate the Satellite portion of this Contract, Purchaser shall be required to terminate the Contract as a whole. Upon receipt by Contractor of the Notice of Termination for Convenience, Contractor shall: (a) stop Work under this Contract on the date and to the extent specified in the Notice of Termination for Convenience; (b) place no further orders or subcontracts for materials, services, or facilities, except as may be necessary for completion of such portion of the Work as is not terminated; (c) terminate orders and subcontracts to the extent that they relate to the performance of Work terminated by the Notice of Termination for Convenience; (d) settle all outstanding liabilities and all claims arising out of such termination or orders, subcontracts for materials, services, or facilities; and (e) with respect to the portion(s) of the Work subject to the Notice of Termination for Convenience, take such action as may be necessary, or as Purchaser may direct, for the protection and or may acquire an interest.
- 21.2 <u>Termination Liability</u>. Subject to Article 32.6:
- 21.2.1 <u>Termination in Whole</u>. If Purchaser terminates this Contract in whole, Purchaser shall be liable to Contractor for the firm fixed price amount set forth in the applicable Whole Termination Liability Schedules of Exhibit F Payment Plan and Termination Liability Amounts as applicable on the date of issuance of the Notice of Termination of Convenience ("Whole Termination Claim"), reduced by all amounts already paid by Purchaser to

Contractor ("Net Whole Termination Claim"). Contractor shall, within fifteen (15) days of the Notice of Termination for Convenience, provide Purchaser with a calculation of the Net Whole Termination Claim, together with an invoice for payment of any deficiency, if any, which invoice shall be paid in accordance with Article 5. In the event that the Net Whole Termination Claim results in moneys being due to Purchaser, Contractor shall at the time of delivery of its calculation of the Net Whole Termination Claim refund to Purchaser said excess funds. In no event shall the amounts payable by Purchaser pursuant to this Article 21.2.1 exceed the Net Whole Termination Claim.

21.2.2 Termination in Part. If Purchaser terminates this Contract only in part and with respect to a Deliverable Item for which a Termination Liability Schedule exists as part of Exhibit F Payment Plan and Termination Liability Amounts (a "Partial Termination for Convenience"), Purchaser shall be liable to Contractor for the firm fixed price amount set forth in the applicable Termination Liability Schedules of Exhibit F Payment Plan and Termination Liability Amounts for such Deliverable Item(s) ("Partial Termination Claim"), reduced by all applicable amounts already paid by Purchaser to Contractor with respect to the terminated Work (the "Net Partial Termination Claim"). Contractor shall, within fifteen (15) days of Purchaser's Notice of Termination for Convenience, provide Purchaser with a calculation of the Net Partial Termination Claim, together with an invoice for payment of any deficiency, if any, which invoice shall be paid in accordance with Article 5. In the event that the Net Partial Termination Claim results in moneys being due to Purchaser, Contractor shall at the time of delivery of its calculation of the Net Partial Termination Claim refund to Purchaser said excess funds. In no event shall the amounts payable by Purchaser pursuant to this Article 21.2.2 exceed the Net Partial Termination Claim. Further, the firm fixed price amount set forth in the Termination Liability Schedule in Exhibit F with respect to the Space

Segment shall be deemed to include payment with respect to the licenses by Contractor pursuant to Article 32 and no additional sums shall be due to Contractor with respect to the ongoing rights of Purchaser under Article 32.

21.3 <u>Disposition of Work</u>.

Title and/or the rights to all items of Work, which would have been incorporated into a Deliverable Item or otherwise conveyed to Purchaser under this Contract [*] (not including title to Deliverable Data or any other portion of the Work to which Contractor would not have otherwise been obligated to transfer title hereunder had the Contract been completed) and which are in progress before the Notice of Termination for Convenience shall, upon payment in full of the Net Whole Termination Claim or the Net Partial Termination Claim, as applicable, vest in Purchaser, and Contractor shall deliver, FOB Contractor's plant, subject to U.S. Export Regulations for which Contractor has no liability in the event of failure to obtain stated export authorizations (other than due to Contractor's fault or negligence), all such items to Purchaser who shall remove such items from Contractor's (or applicable subcontractor's) facility within ninety (90) days of the Notice of Termination for Convenience. All terminated Work which is under the custody or control of Contractor or any subcontractor shall be insured by Contractor at its cost and risk, until the earlier of: (i) Purchaser's removal of all such items of Work from Contractor's (or applicable subcontractor's) facility or (ii) ninety—one (91) days after the Notice of Termination for Convenience, and risk of loss shall pass to Purchaser at such time. Alternatively, Purchaser may request Contractor to make a reasonable, good faith effort to sell all or some of the Work and to remit any sales proceeds to Purchaser less a deduction for Actual Costs reasonably incurred in such sales efforts plus a markup of [*] Within twenty (20) days after the written request of Purchaser, Contractor shall submit a good faith long term storage proposal for the storage of the Work.

[*] CONFIDENTIAL TREATMENT REQUESTED BY ICO GLOBAL COMMUNICATIONS (HOLDINGS) LIMITED. ARTICLE 22 – LIQUIDATED DAMAGES

- 22.1 Liquidated Damages for Late Delivery of Satellite and/or GBBF Subsystem.
- 22.1.1 The Parties acknowledge and agree that failure to meet the Delivery schedule specified in Article 3 for the Satellite and/or the GBBF Subsystem may result in substantial financial loss or damage being sustained by Purchaser. The Parties further acknowledge and agree that the following liquidated damages are believed to represent a genuine and reasonable estimate of certain of the losses (including non-productive time and increased cost of money) that would be suffered by Purchaser by reason of any such delay (which losses would be difficult or impossible to calculate with certainty).
- 22.1.2 If Delivery of the Satellite does not occur on or before the date specified for Satellite Delivery in Article 3.1 (as such date may be adjusted pursuant to Articles 16, 17 or 18), then Contractor shall pay Purchaser as liquidated damages and not a penalty for Contractor's late Delivery the following per day amounts for each such late day:

[*]

Notwithstanding the above, the maximum amount of liquidated damages that Contractor may be required to pay Purchaser for late Delivery of the Satellite shall not exceed [*] for the Satellite.

22.1.3 If Delivery of the GBBF Subsystem does not occur on or before [*] then Contractor shall pay Purchaser as liquidated damages and not a penalty for Contractor's late Delivery the following per day amounts for each such late day:

[*]

Notwithstanding the above, the maximum amount of liquidated damages that Contractor may be required to pay Purchaser for late Delivery of the GBBF Subsystem shall not exceed [*]

In the event the GBBF Subsystem has not been Accepted prior to the commencement of the earlier of [*]

22.3 <u>Payment of Liquidated Damages.</u>

Any amounts due in accordance with this Article 22 shall be, at Purchaser's election, either (i) credited to Purchaser against any outstanding or future invoices hereunder or (ii) paid by Contractor to Purchaser within thirty (30) days of receipt of an invoice from Purchaser.

22.4 Remedy.

The liquidated damages set forth in Articles 22.1 and 22.2 shall be Purchaser's sole remedy and compensation for Contractor delays with respect to late Delivery of the Satellite and/or GBBF Subsystem and [*] and no other liquidated damages shall be payable to Purchaser; provided, however, Purchaser retains all rights and remedies under Article 21 and Article 23 (it being understood and agreed that Purchaser shall have no right to terminate this Contract pursuant to Article 23.1.1 based solely on [*]

[*] CONFIDENTIAL TREATMENT REQUESTED BY ICO GLOBAL COMMUNICATIONS (HOLDINGS) LIMITED. ARTICLE 23 – TERMINATION FOR DEFAULT AND EXCESSIVE FORCE MAJEURE

23.1 Contractor Default.

- 23.1.1A <u>Right to Terminate in Whole or in Part</u>. Subject to Article 23.1.4 and further subject to Articles 16, 17 and 18, Purchaser may terminate this Contract in whole or in part by written notice to Contractor ("Notice of Termination for Default") if:
- (a) Contractor fails to Deliver the Satellite by the date specified in Article 3.1 for Delivery of the Satellite plus [*] (or such longer time as may be agreed to in writing by Purchaser);
- (b) [*]
- (c) Contractor commits a material breach of this Contract and fails, [*] (or such longer period as may be agreed to in writing by Purchaser) after receipt from Purchaser of written notice thereof, to cure such material breach in full.

A "material breach" under subclause (c) of this Article 23.1.1A shall not include grounds for termination set forth in subclauses (a) and (b) above or in Article 23.1.1B, and shall not include delays in Delivery with respect to the Space Segment except to the extent of Article 23.1.1A (a) and (b) and Article 23.1.1B.

23.1.1B Right to Terminate Contract as it Relates to the GBBF Subsystem, In addition to and without limiting Article 23.1.1A, but subject to Article 23.1.4 and further subject to Articles 16, 17 and 18, Purchaser may terminate this Contract as it relates to the GBBF Subsystem, if:

[*	CONFIDENTIAL	TREATMENT RE	QUESTED BY IC	CO GLOBAL COMMUNICATIONS	(HOLDINGS	LIMITED.

- (a) Contractor fails to Deliver the GBBF Subsystem by [*] or such longer time as may be agreed to in writing by Purchaser; or
- (b) It is reasonably certain (or admitted by Contractor) that Contractor shall be unable to Deliver the GBBF Subsystem by the date set forth in subclause (a) above.

23.1.2 <u>Termination Liability</u>.

- (a) In the event of termination pursuant to Article 23.1.1A, but subject to Article 23.1.3, Purchaser shall be entitled to a refund by Contractor of all payments made by Purchaser to Contractor in accordance with this Contract for the terminated Work plus interest thereon at the interest rate stipulated in Article 5.3.1 hereof from the date payment was received by Contractor to the date the refund is received by Purchaser. In addition, Purchaser shall be entitled to payment of any liquidated damages for delay of the terminated Satellite and/or GBBF Subsystem [*] as applicable, which accrued pursuant to Article 22 prior to the date of the Notice of Termination for Default.
- (b) In the event of termination pursuant to Article 23.1.1B, but subject to Article 23.1.3, Purchaser shall be entitled to a refund by Contractor of all payments made by Purchaser to Contractor in respect of the "Space Segment/Ground Based Beam Forming Subsystem" as provided in Article 4.1 (but which shall not include payments made by Purchaser in respect of the Satellite) plus interest thereon at the interest rate stipulated in Article 5.3.1 hereof from the date payment was received by Contractor to the date the refund is received by Purchaser. In addition, Purchaser shall be entitled to payment of any liquidated damages for delay of the

GBBF Subsystem which accrued pursuant to Article 22 prior to the date of the Notice of Termination for Default.

(c) Contractor shall pay any amounts owing to Purchaser under this Article 23.1.2 no later than thirty (30) days after Contractor's receipt of the Notice of Termination for Default. Payment of such amounts shall be Purchaser's sole remedy in case of a termination pursuant to this Article 23.1.

23.1.3 Contractor's Reimbursement for Terminated Work. Purchaser shall not be deemed to retain any Work-in-process other than Work-in-process designated by Purchaser in writing to be retained (and, for example, Contractor shall be required to refund Purchaser with respect to training even if such training had already been completed). In the event of termination pursuant to Article 23.1.1A, Purchaser may in its discretion elect to retain portions of the Work-in-process ("WIP") [*] Purchaser shall deliver notice of its intent to retain any WIP within ten (10) days after the giving of any Notice of Termination for Default. Subject to Article 32.6, the amounts not to be refunded with respect to any retained Deliverable Items shall be: (i) at the price [*] Notwithstanding the above, this Article 23.1.3 shall be subject to Article 32.6.

- 23.1.4 Special Provision Limiting Purchaser's Remedies. Purchaser shall have no right to terminate this Contract pursuant to Article 23.1 with respect to the Satellite after Satellite Delivery, or with respect to the GBBF Subsystem after GBBF Subsystem Acceptance (including in the event of Space Segment Acceptance). Purchaser shall have no right to terminate the Space Segment except to the extent of its rights to terminate the Contract in whole in part in accordance with Article 23.1.1 and with respect to the GBBF Subsystem in accordance with Article 23.1.2.
- 23.1.5 Disposition of the Work. Unless Purchaser retains WIP pursuant to Article 23.1.3 hereof, but subject to Purchaser's rights under Article 32, Contractor shall retain title to any and all Work, Work-in-process, parts or other material, inventories, and any associated warranties, and any subcontracted items Contractor has specifically produced, acquired, or entered into in accordance with this Contract. If Purchaser elects to retain any WIP pursuant to Article 23.1.3, Purchaser may require Contractor to transfer to Purchaser in the manner and to the extent directed by Purchaser, title to and possession of any items comprising all or any part of the WIP (including all Work-in-process, parts and materials, all inventories, and associated warranties) and Contractor shall, upon direction of Purchaser, protect and preserve such WIP at Purchaser's expense in the possession of Contractor or its subcontractors in which Purchaser has an interest and shall facilitate access to and possession by Purchaser of items comprising all or part of the Work terminated. Within twenty (20) days of request of Purchaser, Contractor shall submit a good faith long term storage proposal for the storage of the WIP to be retained by Purchaser and stored with Contractor.

23.1.6 <u>Invalid Default Termination</u>, If, after termination pursuant to this Article 23.1, it is determined pursuant to Article 25 or written agreement of Purchaser that Contractor was not in default under Article 23.1.1, or that the default was excusable under Articles 17 or 18, the rights and obligations of the Parties shall be the same as if Purchaser had, rather than deliver Notice of Termination for Default, delivered a Notice of Termination for Convenience under Article 21 (whether in whole or in part shall be determined whether Purchaser's original Notice of Termination for Default was in whole or in part).

23.2 Purchaser Default.

- 23.2.1 Right to Terminate. Contractor may terminate this Contract upon written notice to Purchaser ("Contractor Notice of Default") if Purchaser fails to cure any default in the payment of any amount that has become due and payable hereunder within thirty (30) days after receiving written notice of such default (or such longer period as may be agreed to in writing by Contractor). Notwithstanding Article 23.2, Purchaser will not be considered in default with respect to any amount that it is disputing in accordance with Article 5.6 and will have ten (10) Business Days after the resolution (by agreement or in accordance with Article 25.2) of such dispute to satisfy any payment required by such resolution.
- 23.2.2 <u>Termination Liability</u>. In the event of termination pursuant to this Article 23.2, it shall be treated as a termination for convenience of the whole of the Contract pursuant to Article 21.1 (effective as of the date of the Contractor Notice of Default). The rights and remedies provided to Contractor under this Article 23.2.2 shall be exclusive and in lieu of any other rights and remedies under this Contract or otherwise provided by law or in equity in relation to the termination of this Contract for Purchaser's default of its obligations under this Contract.

Work-in-process and use or dispose of the same in any manner Contractor may elect. In such case, the fair market value of any applicable Deliverable Items or Work-in-process retained or disposed of by Contractor shall be set-off against Purchaser's Net Whole Termination Claim as established in accordance with Article 21. Upon payment of the amount due under this Article 23.2.3, Purchaser may require Contractor to transfer to Purchaser in the manner and to the extent directed by Purchaser, title to (not including Deliverable Data or any other portion of the Work to which Contractor would not have otherwise been obligated to transfer title hereunder had the Contract been completed) and possession of any items comprising all or any part of the applicable Work terminated (including all Work-in-process, parts and materials, all inventories, and associated warranties) not used or disposed of by Contractor pursuant to the foregoing sentence, and Contractor shall, upon direction of Purchaser, protect and preserve property at Purchaser's expense in the possession of Contractor or its subcontractors in which Purchaser has an interest and shall facilitate access to and possession by Purchaser of items comprising all or part of the applicable Work terminated. Alternatively, Purchaser may request Contractor to make a reasonable, good faith effort to sell such items and to remit any sales proceeds to Purchaser less a deduction for Actual Costs reasonably incurred by Contractor in the sales of such items plus a markup of [*] Within twenty (20) days of written request of Purchaser, Contractor shall submit a good faith long term storage proposal for the storage of the applicable Work.

23.2.4 <u>Invalid Default Termination</u>. If, after termination pursuant to this Article 23.2, it is finally determined pursuant to Article 25 or written agreement of Contractor that Purchaser was not in default under Article

23.2.1, the Contract shall be deemed to have been terminated in whole pursuant to Article 23.1 and Purchaser shall have all rights thereunder.

23.3 Excessive Force Majeure.

Purchaser may, upon written notice to Contractor ("Notice of Termination for Excessive Force Majeure"), immediately terminate this Contract, in whole or in part: if either: (i) Contractor fails to Deliver the Satellite by the date specified in Article 3.1 for Delivery of the Satellite plus [*] due to a Force Majeure in accordance with Article 17, or (ii) the cumulative delay in failure to Deliver the Satellite due to Contractor Force Majeure plus any other non-excused Contractor delay in Delivery of the Satellite by the date specified in Article 3.1 for Delivery of the Satellite shall equal or exceed [*] Upon the occurrence of any of these events and issuance by Purchaser of a Notice of Termination for Excessive Force Majeure, Purchaser shall have the right to terminate the Contract in whole or in part and the rights and obligations of the Parties shall be the same [*]. After the occurrence of an event that would otherwise entitle Purchaser to deliver Notice of Termination for Excessive Force Majeure, Within thirty (30) days of receipt of Contractor's written request, Purchaser will deliver the Notice of Termination for Excessive Force Majeure. Within thirty (30) days of receipt of Contractor's written request, Purchaser will notify Contractor as to its decision to either terminate this Contract or to waive its rights to terminate for such Excessive Force Majeure provided that it shall be a condition to any obligation of Purchaser to provide the requested notice of election that Contractor unconditionally acknowledge Purchaser's rights to terminate in accordance with this Article 23.3. Failure of Purchaser to provide such requested notice of election in the thirty (30) day period specified above

shall be deemed to be a waiver of Purchaser's rights to terminate for such Excessive Force Majeure under this Article 23.3.

B.

Purchaser may, upon written notice to Contractor ("Notice of Termination for Excessive Force Majeure") terminate this Contract as to the GBBF Subsystem under Article 23.1.1B if: (i) due to a Force Majeure in accordance with Article 17, Contractor fails to Deliver the GBBF Subsystem on or before the date specified in Article 3.1 for Delivery of the GBBF Subsystem (as such date may be extended pursuant to Article 18) [*] or (ii) due to a Force Majeure plus any other non-excused Contractor delay in Delivery of the GBBF Subsystem, the cumulative delay in failure to Deliver the GBBF Subsystem by the Delivery date therefor (as such date may be extended pursuant to Article 18) shall equal or exceed [*] Upon the occurrence of any of these events and issuance by Purchaser of a Notice of Termination for Excessive Force Majeure, Purchaser shall have the right to terminate the GBBF Subsystem, and the rights and obligations of the Parties shall be the same [*] After the occurrence of an event that would otherwise entitle Purchaser to deliver Notice of Termination or [*] for Excessive Force Majeure, Contractor shall have the right to deliver a written request to Purchaser that Purchaser advise Contractor whether Purchaser will deliver the Notice of Termination for Excessive Force Majeure. Within thirty (30) days of receipt of Contractor's written request, Purchaser will notify Contractor as to its decision to either terminate this Contract as to the GBBF Subsystem or to waive its rights to terminate for such Excessive Force Majeure under

this Article 23.3B but shall not affect any other rights of Purchaser hereunder.

23.4 Limitation on Right to Terminate.

Except as specified in this Contract, each Party shall have no right to terminate or suspend this Contract.

ARTICLE 24 - OPTIONS

24.1 Option Satellite.

Purchaser shall have an option (the "Satellite Option"), which Purchaser may exercise in writing at any time ("Satellite Option Exercise") during the period from [*] to order an additional satellite (the "Option Satellite") [*]. Upon Satellite Option Exercise, Contractor shall construct and Deliver the Option Satellite, and shall perform all Launch Support Services, Mission Operations Support Services, training and other services in accordance with the terms and conditions of this Contract, except as expressly modified by this Article 24. For purposes of clarification, there will be no FFC Incentive Payments (Article 5.2.3A) or liquidated damages for FCC Events (Article 22.2) with respect to the Option Satellite. Contractor shall Deliver the Option Satellite on or before [*]

24.1.1 The total Option Satellite price includes Launch Support Services, Mission Operations Support Services and other services required to be provided by Contractor under this Contract. The payment plan for the Option Satellite shall be the payment plan applicable at the time of the Satellite Option Exercise set forth on Exhibit F, Payment Plan and Termination Liability Amount (with reference to "Option Satellite Payment Plan"). The price for the Option Satellite (the "Satellite Option Price") shall be determined in accordance with the following table based upon the date of the Satellite Option Exercise:

Option Exercise Date

Satellite Option Price(*)

(*) The Satellite Option Price assumes a Launch on a U.S. domestic Launch Vehicle. Additional charges as noted in Article 4.3 shall apply for a non-U.S. domestic launch.

The Satellite Option Price for the Option Satellite includes Orbital Performance Incentive Payments in an amount equal to [*].

In the event that the Purchaser directs Contractor to place the Optional Satellite into Storage in accordance with Article 24.4 (Storage), Purchaser shall pay Satellite Orbital Performance Incentives for that Optional Satellite in accordance with Article 24.4(f) (Satellite Orbital Performance Incentives During Storage), but shall not be obligated to pay the remaining Launch Milestone, unless Purchaser directs Contractor to Launch a previously stored Satellite in accordance with Article 24.4(e) (Notification of Intention to Launch a Previously Stored Satellite), and that Launch Milestone has been completed and certified as complete by Contractor. 24.1.2 Purchaser shall be permitted to exercise the Satellite Option to order an Option

24.1.3 The Parties shall promptly incorporate the exercise of this option into the Contract through an Amendment according to Article 37.5.

24.2 Satellite Control Center Software/Equipment (SCCS/E).

Purchaser shall have an option, which Purchaser may exercise in writing at any time prior to January 15, 2006, to order Contractor to provide standard Satellite Control Software/Equipment, specifically Integrated Telemetry, Analysis, Control, and Simulation (iTACS) and Analysis and Planning Software (APS), to the designated Purchaser Primary Satellite Control and Backup Control Center (collectively, "SCCS/E"). Delivery and installation of the SCCS/E shall occur two (2) months prior to the scheduled Delivery of the Satellite. The price for the SCCS/E is [*]. The payment plan and termination liability for the SCCS/E shall be the payment plan and termination liability set forth on Exhibit F, Payment Plan with reference to the SCCS/E.

24.3 Extended Warranties for Space Segment. Purchaser shall have four (4) options to extend the entire Space Segment warranty set forth in Article 15.1.3 for additional [*] (for a total warranty of up to approximately [*] for an option price of [*] per option exercised, provided that Purchaser shall deliver written notice of its election to purchase the Space Segment warranty for an additional [*] period not less than thirty (30) days prior to the end of the initial or any subsequent warranty period.

24.4 Storage.

Purchaser, at its option to be exercised no later than four (4) months prior to Delivery of the Satellite, may direct Contractor to store the Satellite, after. Delivery of the Satellite, for a period of up to [*]

- (a) <u>Storage Location</u>. Storage for a Satellite shall be performed at a Contractor controlled facility and shall be performed pursuant to Section 7 (Storage and Post-Storage Test) of Exhibit D, Satellite Test Plan, the Satellite storage plan ("Satellite Storage Plan").
- (b) Storage Prices. The price for Storage of the Satellite placed into storage shall be [*] per month beginning on the first day of the [*] month following commencement of storage for every month the Satellite is stored longer than [*]. The price for Storage includes all costs associated with storage including the post-storage testing as set forth in Section 7 (Storage and Post-Storage Test) of Exhibit D, Satellite Program Test Plan. Purchaser may direct Contractor, through a Change Request, to perform additional post-storage verification testing in addition to those tests described in Section 7 (Storage and Post-Storage Test) of Exhibit D, Satellite Test Plan.
- (c) Payments. The monthly payment for storage costs shall be due thirty (30) days after delivery of the relevant invoice for payment (provided that the first invoice shall not be issued prior to [*] days after commencement of the storage period) and shall continue monthly until the Satellite is removed from storage. Payments under this Article 24.4 shall be made by in accordance with Article 5 hereof.

- (d) <u>Title and Risk of Loss.</u> Except as may be agreed between Purchaser and Contractor, title to a Satellite delivered to Storage and risk of loss (except to the extent of any deductible amount on the policy of insurance covering damage or destruction of the Satellite and allocable thereto, which deductible amount shall be paid by Purchaser to Contractor in the event of any insurance claim, in accordance with Article 5) shall remain with Contractor at all times during storage.
- (e) Notification of Intention to Launch a Previously Stored Satellite. Purchaser shall notify Contractor in writing that a Satellite in Storage pursuant to this option should be removed from storage and delivered to the designated Launch Site. This notification must be received by Contractor not less than six (6) months prior to the new date established by mutual agreement between the Purchaser and Contractor (subject to approval by the Launch Agency and consistent with any required U.S. export license obligations) for delivery of the Satellite to the designated Launch Site. Failure to notify Contractor in a timely manner may result in an Equitable Adjustment. Upon Purchaser direction to Contractor to remove the Satellite from Storage, Contractor shall conduct the post—storage verification tests as set forth in Section 7 (Storage and Post—Storage Test) of Exhibit D, Satellite Test Plan, and ship the Satellite to the designated Launch Site (or to such other location as Purchaser may direct).

(f) Orbital Performance Incentives During Storage. If the Satellite is in Storage for more than [*], then: (a) interest shall begin to accrue on the Satellite Orbital Performance Incentives as if Acceptance of the Satellite had occurred on the [*] and (b) Purchaser shall begin to pay monthly Space Segment Orbital Performance Incentives, Satellite Orbital Performance Incentives, plus the Space Segment Acceptance Payment, and applicable interest in accordance with Article 13 until the Satellite is Accepted (the "Pre-Paid Orbitals").

During such period in Storage, Contractor shall be paid the Space Segment Acceptance Payment and Orbital Performance Incentives as if the Satellite is a Satisfactorily Operating Satellite (i.e., 100% of applicable incentives based on the applicable Daily Rate).

All payments of the Space Segment Acceptance Payment and Orbital Performance Incentives paid during the Storage period shall be treated as prepayments of the Space Segment Acceptance Payment and Orbital Performance Incentives to be earned after Acceptance.

Upon Acceptance of the Satellite and Space Segment, as applicable, Contractor shall be eligible to earn the Space Segment Acceptance Payment and the Orbital Performance Incentives over the full fifteen (15) year Satellite Orbital Incentive Performance Period as provided in Article 13, with Purchaser being given credit for the Pre-Paid Orbitals (together with interest in accordance with Article 13.3 from the time of payment until the credit is applied).

Purchaser's credit for the Pre-Paid Orbitals plus associated interest would be applied as an offset to the Space Segment Acceptance Payment and any amount of Orbital Performance Incentives thereafter earned by

Contractor until all of the Pre-Paid Orbitals and the associated interest has been completely applied. After which, the Purchaser would pay the Orbital Performance Incentives in accordance with Article 13. If pursuant to Article 13 it is clear that the credit for Pre-Paid Orbitals will not be completely applied during the Satellite Orbital Incentive Performance Period, then the balance of any Pre-Paid Orbitals (including applicable interest) shall be re-paid to Purchaser within thirty (30) days of this determination.

- (g) Stored Satellite Refurbishment. By no later than [*] of storage of a Satellite, Purchaser shall notify Contractor in writing of its desire to have such Satellite refurbished once it is stored for [*] or to continue storage past [*] (identifying the extended storage requirement). Within sixty (60) days after receipt of Purchaser's notice electing refurbishment or continued storage, Contractor shall, at no cost to Purchaser, provide Purchaser with (i) a plan for refurbishment and a retest plan to verify the Satellite as launch—worthy (with the costs of implementation of said plan to be borne by Purchaser in accordance with Article 16.1), or (ii) a plan for continued storage, in either case together with proposed adjustments to the applicable provisions of this Contract.
- (h) <u>Delivery to Purchaser other than for Launch</u>, Purchaser shall have the right at any time to require Contractor to deliver to Purchaser a Satellite which is in Storage. Upon written request from Purchaser, Contractor shall at its own cost deliver the Satellite to its loading dock and assist Purchaser in the transport of said Satellite from Contractor's facilities. Contractor shall also provide Purchaser with an inventory of the Satellite and such other related documentation as Purchaser may reasonably request. Title to the Satellite and risk of loss thereto shall transfer to the Purchaser upon Contractor's delivery of the Satellite to its

loading dock and Purchaser shall be responsible for payment of any Taxes which may be imposed upon such transfer. In addition, in the event that Purchaser does not Launch the Satellite within ninety (90) days of title transfer pursuant to this paragraph, Contractor shall be entitled to the Space Segment Acceptance Payment and to start earning its Satellite Orbital Performance Incentives at one hundred percent (100%) of the Daily Rate which to be paid, including interest thereon (to the extent not previously prepaid) in accordance with Article 13, with the date of Acceptance of the Satellite being deemed the date of title transfer thereto.

24.5 Space Segment/GBBF Subsystem (Not Including Satellite) Option.

24.5.1 Purchaser shall have an option (the "Space Segment Option"), which Purchaser may exercise in writing at any time ("Space Segment Option Exercise") during the period from EDC until [*] to order an additional Space Segment (not including a satellite but including the GBBF Subsystem and those related items and services identified for the "Space Segment/Ground-Based Beam Forming Subsystem (Not Including Satellite)" in Article 4.1) (the "Option Space Segment") that is in all material respects substantially similar to the Space Segment. Upon Space Segment Option Exercise, Contractor shall construct and Deliver the Option Space Segment and shall perform all training and other services in accordance with the terms and conditions of this Contract, except as expressly modified by this Article 24. Contractor shall Deliver the Option Space Segment on or before [*] after exercise of this option, but in no event earlier than [*] after Acceptance of the first Space Segment currently scheduled for Delivery in accordance with Article 3.1.

24.5.2 The option price of the Space Segment Option ("Space Segment Option Price") shall be as follows:

Ontion Exercise Date				Option Price	- David	
[*]	all all the state of the state	and a contract of the second		State William or hade of	4	-
			• '			
 n 1 1111	المحادي والكواد التناسية والمرادي	وأحالهم والأمام المراجع المراج	an Ontina Sann	Command that is not in a	II matarial reces	ote e

- 24.5.3 Purchaser shall be permitted to exercise the Space Segment Option to order an Option Space Segment that is not in all material respects substantially similar to the GBBF Subsystem, in which case the Space Segment Option Price specified above shall be adjusted due to the material differences between the Space Segment and the Option Space Segment in accordance with Article 16 as applicable to changes requested by Purchaser.
- 24.5.4 The Parties shall promptly incorporate the exercise of this option into the Contract through an Amendment according to Article 37.5.

ARTICLE 25 - DISPUTE RESOLUTION

Any dispute, claim, or controversy between the Parties arising out of or relating to this Contract ("Dispute"), including any Dispute with respect to the interpretation, performance, termination, or breach of this Contract or any provision thereof shall be resolved as provided in this Article 25 (except that indemnity actions under Article 19 and/or Article 20 shall be resolved as between the indemnifying Party and a third party claimant in the applicable forum).

25.1 Informal Dispute Resolution.

Prior to the initiation of formal dispute resolution procedures, the Parties shall first attempt to resolve their Dispute informally, in a timely and cost-effective manner, as follows:

- 25.1.1 If, during the course of the Work, a Party believes it has a Dispute with the other Party, the disputing Party shall give written notice thereof, which notice will describe the Dispute and may recommend corrective action to be taken by the other Party. Contractor Program Manager shall promptly consult with Purchaser Contract Manager in an effort to reach an agreement to resolve the Dispute.
- 25.1.2 In the event agreement cannot be reached within ten (10) days of receipt of written notice under Article 25.1.1, either Party may request the Dispute be escalated, and the respective positions of the Parties shall be forwarded to an executive level higher than that under Article 25.1.1 above for resolution of the Dispute.
- 25.1.3 In the event agreement cannot be reached within twenty (20) days of receipt of written notice under Article 25.1.1, either Party may request the Dispute be escalated, and the respective positions of the Parties shall be forwarded to the Chief Executive Officer (CEO) or highest senior executive of each Party for resolution of the Dispute.

25.1.4 In the event (i) agreement cannot be reached within thirty (30) days of receipt of written notice in 25.1.1, after application of Articles 25.1.1, 25.1.2, 25.1.3 or (ii) a Party determines in good faith that amicable resolution through continued negotiation of the Dispute does not appear likely, either Party may proceed in accordance with Article 25.2 or Article 25.3, as applicable.

25.2 Expedited Arbitration for Certain Disputes.

Any Disputes regarding (i) amounts not paid by Purchaser to Contractor under Article 5.6, (ii) amounts in dispute under Article 16.1.3 or 16.2, (iii) payment disputes arising under Article 13 and Article 22, and/or (iv) as otherwise mutually agreed, shall be settled by final and binding arbitration pursuant to the JAMS Streamlined Arbitration Rules and Procedures. The expedited arbitration shall be before a single arbitrator and shall be completed within ninety (90) days after appointment of the arbitrator. The Parties shall have ten (10) days to mutually agree upon an arbitrator to conduct the expedited proceeding. Should the Parties fail to reach agreement within such ten (10) days period, a single arbitrator shall be appointed by JAMS. In the case of disputed payments under Article 5.6, the arbitrator shall determine whether Contractor completed the disputed Milestone(s) in accordance with the requirements of the Contract. The Parties shall each bear half of the costs of the arbitration (including arbitrator fees), except each Party shall bear its own attorneys' fees and any other such costs incurred by a Party on its own behalf. Any determination hereunder shall be enforceable in any court of competent jurisdiction as provided in Article 25.3. In the event that the arbitrator rules that a payment is due, the Party owing such payment shall have ten (10) Business Days to pay any and all amounts awarded by the arbitrator within such ten (10) Business Day period shall be an event of default pursuant to Article 23 hereof, and shall not require further notice or period for cure.

25.3 Litigation.

- 25.3.1 Notwithstanding Article 25.1 and 25.2, if a Party makes a good faith determination that (i) a breach by the other Party is such that a temporary restraining order or other preliminary injunctive or other equitable relief to enforce its rights or the other Party's obligations under the provisions of this Contract and which is necessary or (ii) litigation is appropriate to avoid the expiration of an applicable limitations period or to preserve a superior position with respect to creditors, either Party shall have the right to bring suit at any time.
- 25.3.2 Subject to Article 25.1 and 25.2, any Dispute shall be brought in the United States District Court for the Central District of California, and the Parties hereby waive any objection to that venue and that court's exercise of personal jurisdiction over the case; provided further that if, for any reason, such court does not have or refuses to exercise subject matter jurisdiction over the Dispute, then litigation as permitted herein may be brought in the Superior Court for the County of Los Angeles. The Parties hereby irrevocably consent to the exercise of personal jurisdiction by the state and federal courts in the State of California concerning any Dispute between the Parties. If, for any reason, neither the state nor federal courts in California will exercise jurisdiction over the Dispute, then litigation as permitted herein may be brought in any court of competent jurisdiction in the United States of America.

25.3.3 [*]

[*]

25.3.4 Nothing in this Contract precludes a Party prevailing on any claim, from initiating litigation in any appropriate forum to enter or enforce a judgment based on the award on that claim.

25.4 <u>Continued Performance</u>.

Unless the Dispute involves a termination of the Contract under Articles 21 or 23 hereof: (i) pending final resolution of any Dispute, each Party shall, unless directed otherwise by the other Party in writing, perform all its obligations under this Contract to the extent undisputed and practical to do so, including the obligation to take all steps necessary during the pendency of the Dispute to ensure the Work will be performed within the time stipulated or within such extended time as may be allowed under this Contract; and (ii) failure to pay disputed amounts under Article 5.6 shall not excuse failure to so perform the Work.

ARTICLE 26 - INTER-PARTY WAIVER OF LIABILITY FOR A LAUNCH

26.1 <u>Launch Services Agreement Inter-Party Waiver of Liability.</u>

- 26.1.1 Inter—Party Waiver. Each Party hereby agrees to be bound by the no-fault, no-subrogation inter—party waiver of liability and related indemnity provisions required by the Launch Services Agreement with respect to each Launch and to cause their respective contractors and subcontractors at any tier (including suppliers of any kind) that are involved in the performance of this Contract and any other person having an interest in the Satellite (including customers of Purchaser) to accede to such waiver and indemnity, which in every case shall include claims against the Launch Agency, either Party and their respective contractors and subcontractors at any tier (including suppliers of any kind) that are involved in the performance of this Contract. The Parties shall execute and deliver any instrument that may be reasonably required by the Launch Agency to evidence their respective agreements to be bound by such waivers.
- 26.1.2 <u>Waiver of Subrogation</u>. The Parties also shall use best reasonable efforts to obtain from their respective insurers, and shall require their respective contractors and subcontractors at any tier (including suppliers of any kind) that are involved in the performance of this Contract and any other person having an interest in the Satellite (including non-consumer customers of Purchaser) to use best reasonable efforts to obtain from their respective insurers, an express waiver of such insurers' rights of subrogation with respect to any and all claims that have been waived pursuant to this Article 26.
- 26.2 Indemnity Related to the Inter-Party Waiver of Liability.

Each Party shall indemnify against and hold harmless the other Party from and against any claim the other Party or its contractors and subcontractors at any tier

(including suppliers of any kind) that are involved in the performance of this Contract, made by the indemnifying Party or any of its contractors and subcontractors (including suppliers of any kind) that are involved in the performance of the Contract, or by any person having an interest in the Satellite (including customers of Purchaser), or by insurer(s) identified in Article 26.1, resulting from the failure of the indemnifying Party to waive any liability against, or to cause any other person the indemnifying Party is obligated to cause to waive any liability against, the Launch Agency, the other Party or either of their contractors and subcontractors at any tier (including suppliers of any kind) involved in the performance of this Contract. The Parties shall execute and deliver any instrument that may be reasonably required by the Launch Agency to evidence their respective agreements to be bound by such indemnifications.

26.3 Survival of Obligations.

The waiver, indemnification and hold harmless obligations provided in this Article 26 shall survive and remain in full force and effect, notwithstanding the expiration or termination of this Contract.

26.4 Additional Insured.

Purchaser shall use reasonable efforts to cause the Launch Agency to name, as additional insured under any third-party liability insurance procured by the Launch Agency under the Launch Services Agreement for the Launch of the Satellite, Contractor and any other person identified by Contractor in writing to Purchaser no later than one hundred and eighty (180) days before such Launch.

[*] CONFIDENTIAL TREATMENT REQUESTED BY ICO GLOBAL COMMUNICATIONS (HOLDINGS) LIMITED. ARTICLE 27 – MAJOR SUBCONTRACTS

27.1 Selection of Major Subcontractors for Satellite.

27.1.1 Selection of any Major Subcontractor for the Satellite, whether as an initial selection or as a replacement selection, shall be subject to Purchaser's prior written approval. [*] the Major Subcontractors listed on the Major Subcontracts List dated November 19, 2005, for use as either initial selection or as a replacement selection of Major Subcontractors. Contractor shall provide Purchaser with a copy of the full text of any Major Subcontract (including technical content but excluding price and payment schedule) promptly upon execution thereof. The Parties further agree that [*] shall be the principal GBBF Subsystem supplier (as reflected in Attachment E).

27.1.2 [*] any Major Subcontractor or subcontractor shall not relieve Contractor from any obligations or responsibilities under this Contract.

ARTICLE 28 - CONTRACTOR INSURANCE REQUIREMENTS

28.1 <u>Insurance Requirements.</u>

- 28.1.1 <u>Coverages.</u> Contractor represents that it has procured and will maintain at all relevant times during its performance of this Contract the following insurance coverages:
- A. Ground insurance ("Property Insurance") against all risks and loss or damage to the Satellite, the GBBF Subsystem, or DSS and to any and all component parts thereof and all materials of whatever nature used or to be used in completing the Work, in an amount not less than the greater of (i) the replacement value of or (ii) the amounts paid by Purchaser with respect to, the Satellite, the GBBF Subsystem, DSS and component parts thereof and all materials of whatever nature used or to be used in completing the Work. Such insurance shall provide (i) coverage for removal of debris, and insuring the structures, machines, equipment, facilities, fixtures, and other properties constituting part of the Work, (ii) transit coverage, including ocean marine coverage (unless insured by the supplier), (iii) off-site coverage covering any key equipment, and (iv) off-site coverage covering any property or equipment not stored on the construction site. For the Satellite, the GBBF Subsystem, and DSS such insurance shall cover the period beginning at EDC up to the moment risk of loss passes to Purchaser in accordance with this Contract. The deductible on such policy shall not exceed standard deductibles in the industry for like insurance.
- B. Worker's compensation insurance, including occupational illness or disease coverage, or other similar social insurance in accordance with the laws of any country, state, or territory exercising jurisdiction over the employee and employer's liability insurance in an amount not less than One Million United States dollars (U.S. \$1,000,000)

Contractor shall maintain		

- C. Comprehensive automobile liability insurance against liability claims for personal injury (including bodily injury and death) and property damage covering all owned, leased, non-owned, and hired vehicles used by Contractor in the performance of the Work. Such insurance shall be for an amount not less than One Million United States dollars (U.S. \$1,000,000) per occurrence for combined bodily injury and property damage. Contractor shall maintain such insurance until Acceptance of all Work, including remedial work, has occurred; and
- D. Such other insurance in types and amounts as is adequate to cover Contractor's potential liabilities under this Contract and applicable law.
- 28.1.2 <u>Additional Insured.</u> Purchaser and each Financing Entity shall be named as an additional named insured under Contractor's third-party liability coverages, provided that, with respect to each Financing Entity, such Financing Entity has an insurable interest recognized by the applicable insurance underwriters.
- 28.1.3 <u>Insurers Rating.</u> The insurers selected by Contractor to provide the insurance required by Article 28.1.1 above shall have a rating at least as high as those insurers providing coverage on Contractor's programs for its other major commercial customers.
- 28.1.4 Evidence of Insurance. Prior to commencing the Work, and whenever requested by Purchaser, Contractor shall produce evidence that the insurance required by Article 28.1.1 has been effected and is being maintained. Contractor shall, at the written request of Purchaser, provide Purchaser with a certificate of insurance evidencing the procurement of all required insurance policies and thirty (30) days written notice prior to any

modification that diminishes the insurance coverage required hereunder, cancellation, or non-renewal of such policies. If, after being requested in writing by Purchaser to do so, Contractor fails to produce evidence of compliance with Contractor's insurance obligations within fourteen (14) days, Purchaser may effect and maintain the insurance and pay the premiums. The amount paid shall be a debt due from Contractor to Purchaser and may be offset against any payments due Contractor by Purchaser. Purchaser may, at reasonable times upon reasonable notice, inspect any insurance policy required hereunder at Contractor's offices.

- 28.1.5 <u>Claims.</u> Contractor shall, as soon as practicable, inform Purchaser in writing of any occurrence with respect to the Work that may give rise to a claim under a policy of insurance required by paragraph (a) above. Contractor shall ensure that its Subcontractors similarly inform Purchaser of any such occurrences through Contractor.
- 28.1.6 <u>Waiver of Subrogation</u>. Contractor shall use best reasonable efforts to require its insurers to waive all rights of subrogation against Purchaser and Purchaser's Affiliates and their respective Associates.
- 28.1.7 Warranty. Contractor warrants and covenants that the insurance coverages and deductibles set forth in this Article 28 are substantially comparable to those provided to Contractor's other major commercial customers.

28.2 <u>Preparation of Claims.</u>

Each Party shall provide to the other Party any information that may reasonably be required to prepare, present, and substantiate an insurance claim at the other Party's written request.

28.2.1 Subject to compliance with Article 7, each Party warrants and covenants that it will not intentionally withhold from the other Party any material information it has or will have concerning anomalies, failures, or non-conformances with or deviations from the requirements of this Contract.

28.2.2 Upon written request of a Party, subject to Article 7, the other Party will respond or permit the first Party to respond to any insurers in relation to all specific and reasonable questions relating to design, test, quality control, launch, and orbital information. In addition, in the event of a Launch and In-Orbit Insurance claim, Contractor shall provide the support described in Article 35.2.

ARTICLE 29 - PERSONNEL AND KEY PERSONNEL

29.1 Personnel Qualifications.

Contractor shall assign properly qualified and experienced personnel to the program contemplated under this Contract.

29.2 Kev Personnel Positions.

Within fifteen (15) days after EDC, Contractor shall notify Purchaser of the program Key Personnel. The Key personnel ("Key Personnel") shall be the personnel filling the following or equivalent positions:

- 29.2.1 Contractor Program Manager;
- 29.2.2 Spacecraft Engineering Manager;
- 29.2.3 Payload Program Manager;
- 29.2.4 Assembly, Integration and Testing Manager, and
- 29.2.5 Space Segment Manager
- 29.2.6 Space Segment Technical Lead
- 29.3 <u>Assignment of Kev Personnel.</u>
- 29.3.1 Contractor will assign individuals from within Contractor's organization to the Key Personnel positions to carry out the Work. Such individuals shall be fully competent and skilled to perform the services required of them pursuant to this Contract.
- 29.3.2 Key Personnel will be familiar with programs similar to Purchaser's program.

29.3.3 Before assigning an individual to any Key Personnel positions, whether as an initial assignment or a subsequent assignment, Contractor shall notify Purchaser of the proposed assignment, shall introduce the individual to appropriate Purchaser representatives and, upon request, provide such representatives with the opportunity to interview the individual and shall provide Purchaser with the individual's resume. If Purchaser in good faith objects to the qualifications of the proposed individual within fifteen (15) Business Days after being notified thereof, then Contractor agrees to discuss such objections with Purchaser and attempt to resolve such concerns on a mutually agreeable basis; however, Contractor retains the unilateral right to make all decisions regarding the assignment of Contractor personnel to the program. The Key Personnel that have been identified as of the Effective Date of Contract are listed in Attachment B (Key Personnel). Should the individuals filling the positions of Key Personnel leave such positions for whatever reason, Contractor shall follow the procedures set forth in this Article 29 to select replacement personnel. Contractor shall use reasonable best efforts to retain such personnel on Purchaser's program for the duration of such program.

ARTICLE 30 - LIMITATION OF LIABILITY

30.1 Limitation of Liability.

30.1.1 No Consequentials. WITHOUT PREJUDICE TO THE PARTIES' OBLIGATIONS SPECIFIED IN ARTICLES 4.2 (WITH REGARD TO TAXES), 19.1, 20, 26.1.1 AND 26.2, NEITHER PARTY SHALL BE LIABLE DIRECTLY OR INDIRECTLY TO THE OTHER, TO THEIR OFFICERS, DIRECTORS, EMPLOYEES, CONTRACTORS OR SUBCONTRACTORS AT ANY TIER (INCLUDING SUPPLIERS OF ANY KIND), AGENTS OR CUSTOMERS, TO ITS PERMITTED ASSIGNEES OR SUCCESSOR OWNERS OF THE SATELLITE OR OTHER DELIVERABLE ITEM OR TO ANY OTHER PERSON CLAIMING BY OR THROUGH PURCHASER FOR ANY AMOUNTS REPRESENTING LOSS OF PROFITS, LOSS OF BUSINESS, OR INDIRECT, SPECIAL, INCIDENTAL, EXEMPLARY, CONSEQUENTIAL OR PUNITIVE DAMAGES, INCLUDING WITHOUT LIMITATION COSTS OF EFFECTING COVER, LOST PROFITS, LOST REVENUES OR COSTS OF RECOVERING THE SATELLITE OR SPACE SEGMENT ARISING FROM OR RELATING TO THE PERFORMANCE OR NONPERFORMANCE OF THIS CONTRACT OR ANY ACTS OR OMISSIONS ASSOCIATED THEREWITH OR RELATED TO THE USE OF ANY ITEMS DELIVERED OR SERVICES FURNISHED HEREUNDER, WHETHER THE BASIS OF SUCH LIABILITY IS BREACH OF CONTRACT, TORT (INCLUDING NEGLIGENCE OF ANY TYPE AND STRICT LIABILITY), STATUTE OR OTHER LEGAL OR EQUITABLE THEORY.

30.1.2 <u>Indemnity</u>. EACH PARTY SHALL INDEMNIFY THE OTHER PARTY AND HOLD THE OTHER PARTY HARMLESS FOR AND AGAINST ANY CLAIM ASSERTED DIRECTLY OR INDIRECTLY AGAINST SUCH OTHER PARTY THAT IS WITHIN THE SCOPE OF THE FOREGOING LIMITATION OF LIABILITY AND DISCLAIMER.

30.2 Can on Liability.

WITHOUT LIMITING CONTRACTOR'S OBLIGATION TO INDEMNIFY PURCHASER UNDER ARTICLES 19, 20 AND 26.2, IN NO EVENT SHALL CONTRACTOR'S TOTAL LIABILITY UNDER OR IN CONNECTION WITH THIS CONTRACT EXCEED AMOUNTS PAID TO CONTRACTOR HEREUNDER PLUS ANY INTEREST DUE PURSUANT TO ARTICLE 23.2.2 PLUS ANY LIQUIDATED DAMAGES DUE UNDER ARTICLE 22. WITHOUT LIMITING PURCHASER'S OBLIGATION TO INDEMNIFY CONTRACTOR UNDER ARTICLE 20 AND ARTICLE 26.2, IN NO EVENT SHALL PURCHASER'S TOTAL LIABILITY UNDER OR IN CONNECTION WITH THIS CONTRACT EXCEED PURCHASER'S LIABILITY AS SET FORTH IN ARTICLE 21.2.1. THIS ARTICLE 30 SHALL SURVIVE THE EXPIRATION OR TERMINATION OF THIS CONTRACT FOR WHATEVER CAUSE.

ARTICLE 31 - DISCLOSURE AND HANDLING OF PROPRIETARY INFORMATION

31.1 <u>Definition of Proprietary Information.</u>

31.1.1 Definition.

For the purpose of this Contract, "Proprietary Information" means all confidential and proprietary information (other than the Exhibits and Attachments to this Contract and Deliverable Data, which are subject to the provisions of Article 32) in whatever form transmitted, that is disclosed or made available directly or indirectly by such Party (hereinafter referred to as the "disclosing party") to the other Party hereto (hereinafter referred to as the "receiving party") and: (i) is identified as proprietary by means of a written legend thereon, or (ii) if disclosed orally, is identified as proprietary at the time of initial disclosure and then summarized in a written document, with the Proprietary Information specifically identified, that is supplied to the receiving party within ten (10) days of initial disclosure. In the case of Purchaser, Proprietary, Information also shall include, whether or not designated "Proprietary Information (subject to the exceptions contained in Article 31.1.2)," (i) correspondence under this Contract and (ii)) all information concerning Purchaser (and/or its Affiliates) regarding its operations, affairs and businesses, its financial affairs, and its relations with its customers, employees and service providers (including business plans, customer lists, customer information, account information and consumer markets).

31.1.2 Exceptions.

Proprietary Information shall not include any information disclosed by a Party that (i) is already known to the receiving party at the time of its disclosure, as evidenced by written records of the receiving party, without an obligation of confidentiality at the time of disclosure; (ii) is or becomes

publicly known through no wrongful act of the receiving party; (iii) is independently developed by the receiving party as evidenced by written records of the receiving party, or (iv) is rightfully obtained by the receiving party from any third party without restriction and without breach of any confidentiality obligation by such third party.

31.2 Terms for Handling and Use of Proprietary Information.

Subject to Article 31.1.2, for a period of ten (10) years after receipt of any Proprietary Information, the receiving party shall not disclose Proprietary Information that it obtains from the disclosing party to any person or entity except its employees, Affiliates (who are not direct competitors of the disclosing party), attorneys, agents and consultants (who are not direct competitors of the disclosing party) who have a need to know, who have been informed of and have agreed in writing (or, in the case of employees or attorneys are otherwise subject to confidentiality obligations consistent with the obligations set forth herein) to abide by the receiving party's obligations under this Article 31, and who are authorized pursuant to applicable U.S. export control laws and licenses or other approvals to receive such information. The receiving party shall use not less than the same degree of care to avoid disclosure of such Proprietary Information as it uses for its own Proprietary Information of like importance; but in no event less than a reasonable degree of care. Proprietary Information shall be used only for the purpose of performing the obligations under this Contract, or as the disclosing party otherwise authorizes in writing.

31.3 Reserved.

31.4 Legally Required Disclosures.

Notwithstanding the foregoing, in the event that the receiving party becomes legally compelled (including disclosures necessary or in good faith determined to be reasonably necessary under the Securities Act of 1933, as amended, and the Securities Exchange Act of 1934, as amended), to disclose Proprietary Information of the disclosing party, including this Contract or other supporting

document(s), the receiving party shall, to the extent practicable under the circumstances, provide the disclosing party with written notice thereof so that the disclosing party may seek a protective order or other appropriate remedy, and shall cooperate in good faith with the disclosing party to redact such portions of the Proprietary Information as the disclosing party shall require and applicable law shall permit. In any such event, the receiving party will disclose only such information as is legally required, and will cooperate with the disclosing party (at the disclosing party's expense) to obtain proprietary treatment for any Proprietary Information being disclosed.

31.5 Reserved.

31.6 Return of Confidential Information.

Upon the request of the Party having proprietary rights to Proprietary Information, the other Party in possession of such Proprietary Information shall promptly return such Proprietary Information (and any copies, extracts, and summaries thereof) to the requesting Party, or, with the requesting Party's written consent, shall promptly destroy such materials (and any copies, extracts, and summaries thereof), except for one (1) copy which may be retained for legal archive purposes, and shall further provide the requesting Party with written confirmation of same; provided, however, where both Parties have proprietary rights in the same Proprietary Information, a Party shall not be required to return such information to the other Party.

31.7 No License.

Except as expressly provided in this Contract, nothing in this Contract shall be construed as granting the receiving party whether by implication, estoppel, or otherwise, any license or any right to use any Proprietary Information received from the disclosing party, or use any patent, trademark, or copyright now or hereafter owned or controlled by the disclosing party.

ARTICLE 32- INTELLECTUAL PROPERTY RIGHTS

32.1 Ownership of IP and IP Rights Generally.

32.1.1 Purchaser's Intellectual Property.

Subject to the licenses granted in Article 32.2 and Article 32.3, and except as otherwise specified in Article 32.1.2, all Purchaser Background Intellectual Property, Purchaser Space Segment Foreground Intellectual Property, and Foreground Intellectual Property made, developed, or created by Purchaser (or by others, other than Contractor or any of its subcontractors, acting on behalf of Purchaser), and all Intellectual Property Rights therein, shall be the sole and exclusive property of Purchaser.

32.1.2 <u>Contractor's Intellectual Property.</u>

- A. Subject to the licenses granted in Article 32.2 and Article 32.3, all Contractor Background Intellectual Property, Contractor Space Segment Intellectual Property, and Foreground Intellectual Property made, developed, or created by Contractor (or by others acting on Contractor's behalf including any subcontractor), and all Intellectual Property Rights therein, shall be the sole and exclusive property of Contractor. Subject to the licenses granted in Article 32.2 and Article 32.3, and notwithstanding Article 32.1.1, all Intellectual Property incorporated into the Satellite, and all Intellectual Property Rights therein, are and shall be the sole and exclusive property of Contractor.
- B. For the avoidance of doubt but subject to the exceptions set forth in subsection (A) above, the Parties agree that Contractor shall retain title to all Deliverable Data (excluding Purchaser Space Segment Foreground Intellectual Property incorporated therein) and Contract Exhibits and Attachments thereto utilized or developed by Contractor during the performance of this Contract.

- 32.1.3 <u>Confidentiality</u>. Intellectual Property shall be subject to the protections for confidentiality under Article 31 to the extent such duties do not conflict with enjoyment of the rights granted with respect to such Intellectual Property under the licenses set forth in this Article 32.
- 32.2 License Rights in Respect of Contractor Space Segment Intellectual Property and Purchaser Space Segment Foreground Intellectual Property.

32.2.1 Grant by Contractor.

Subject to the terms and conditions stated herein and expressly including Article 32.6, to the extent Contractor has or acquires the right to grant such a license, Contractor grants to Purchaser and its permitted successors and assigns a fully paid-up, perpetual, irrevocable, worldwide, non-exclusive right and license, with right of sublicense, to use and have used, reproduce, and modify solely and exclusively for (i) the purpose of testing, operating, repairing, and/or maintaining any Deliverable Item (including Deliverable Data), and (ii) the Permitted Purposes, all Contractor Space Segment Intellectual Property including, to the extent necessary for the purpose of this license, those associated Intellectual Property Rights therein, now or hereafter owned by Contractor (and/or others acting on Contractor's behalf including any subcontractor).

32.2.2 Grant by Purchaser.

Subject to the terms and conditions stated herein, to the extent Purchaser has or acquires the right to grant such a license, Purchaser grants to Contractor and its permitted successors and assigns a fully paid-up, perpetual, irrevocable, worldwide, non-exclusive right and license, with right of sublicense, to use and have used, reproduce, and modify solely and exclusively for the purpose of conducting Contractor's business of designing, developing, manufacturing and selling space systems all Purchaser Space Segment Foreground Intellectual Property, including, to the extent necessary for the purpose of this license, those associated

Intellectual Property Rights therein, now or hereafter owned by Purchaser (and/or others acting on Purchaser's behalf including any third party). Notwithstanding the above, Purchaser shall have no obligation to provide any documents, data, materials or other manifestations of such Intellectual Property except as set forth in Article 32.2.5 below.

32.2.3 Reciprocal Covenant Not to Interfere with Exercise of License Rights.

In no event shall either Party be precluded from developing for itself or for others Intellectual Property, materials or products which are competitive with, or similar to, the Deliverable Items if such Intellectual Property, materials and products do not include any of the other Party's Intellectual Property or Proprietary Information except pursuant to a license granted under Articles 32.2 or 32.3. [*]

[*]

32.2.5 Notification of Purchaser Space Segment Foreground Intellectual Property.

[*]

[*]

32.3 <u>License Rights in Respect of Intellectual Property Other Than Contractor Space Segment Intellectual Property and Purchaser Space Segment Foreground Intellectual Property.</u>

32.3.1 Grants by Contractor.

Subject to the terms and conditions stated herein, and except for Contractor Space Segment Intellectual Property (which is subject to Article 32.2 above), to the extent Contractor has or may acquire the right to grant such a license, Contractor grants to Purchaser and its permitted successors and assigns a fully paid-up, perpetual, irrevocable, worldwide, non-exclusive right and license to use and have used, reproduce, and modify for the sole and exclusive purpose of testing, operating, repairing, and/or maintaining any Deliverable Item (including Deliverable Data), all Contractor Background Intellectual Property, Third Party Intellectual Property and Foreground Intellectual Property made, developed or created by Contractor (or by others acting on behalf of Contractor including any subcontractor) or a third party, as the case may be, incorporated into such Deliverable Items (for the purposes of this paragraph (A), Deliverable Items includes Contract Exhibits and Attachments thereto), including, to the extent necessary for the limited purpose of this license, those associated Intellectual Property Rights therein, now or hereafter owned by Contractor (and/or its subcontractors) or a third party, as the case may be. Purchaser shall have no rights in Deliverable Data other than as expressly stated in this Contract.

32.3.2 Grants by Purchaser.

Subject to the terms and conditions stated herein, and except for Purchaser Space Segment Foreground Intellectual Property (which is subject to Article 32.2 above), to the extent Purchaser has or acquires the right to grant such a license, Purchaser grants to Contractor and its permitted successors and assigns a fully paid-up, perpetual, irrevocable, worldwide, non-exclusive right and license to use and have used for the sole and exclusive purpose of performing under this Contract, all Purchaser Background Intellectual Property, Third Party Intellectual Property and Foreground Intellectual Property, including, to the extent necessary for the limited purpose of this license, those associated Intellectual Property Rights therein, now or hereafter owned by Purchaser (or others acting on behalf of Purchaser) or a third party, as the case may be.

32.4 Subcontracts and Other Affected Agreements

A. [*] In addition, Contractor shall, unless otherwise authorized or directed in writing by Purchaser, to the extent necessary to fulfill its obligations under Article 32.2 and Article 32.3 hereof, use reasonable efforts as practical to include in each additional subcontract issued hereunder a license rights clause pursuant to which each such subcontractor will grant to Purchaser (through Contractor) license rights in Intellectual Property incorporated in Deliverable Items hereunder and which Intellectual Property is developed by such subcontractor, and all associated Intellectual Property Rights therein, to the same extent as the license rights granted by Contractor in this Article 32. Notwithstanding any other provision of this Contract, this Article 32.4 (A) shall govern Contractor's obligations with respect to

subcontracts, and no additional rights shall arise, for example, because Intellectual Property of Contractor is defined to include Intellectual Property of subcontractors and other third parties.

- B. Purchaser shall, unless otherwise authorized or directed in writing by Contractor, to the extent necessary to fulfill its obligations under Article 32.2 and Article 32.3 hereof, use reasonable efforts as practical to include in each applicable third party and/or related party agreement a license rights clause pursuant to which each such party will grant to Contractor (through Purchaser) license rights in Intellectual Property within the scope of the licenses granted by Purchaser under Article 32.2 and Article 32.3, which Intellectual Property is developed by such party, and all associated Intellectual Property Rights therein, to the same extent as the license rights granted by Purchaser in this Article 32. Notwithstanding any other provision of this Contract, this Article 32.4 (B) shall govern Purchaser's obligations with respect to third party and/or related party agreements, and no additional rights shall arise, for example, because Intellectual Property of Purchaser is defined to include Intellectual Property of third parties.
- C. Upon execution of the subcontract with the GBBF Subsystem supplier, Contractor shall provide Purchaser with copies of all provisions and terms regarding Intellectual Property and will also furnish Purchaser with any amendments pertaining thereto (and shall not amend such provisions to the detriment of Purchaser without Purchaser's written consent).

32.5 No Limitation on Deliverable Items.

This Article 32 shall not be construed as limiting any right of Purchaser otherwise contained herein or at law (or any obligation of Contractor to grant Purchaser the right), subject to compliance with applicable laws and with no payment of additional compensation to use, have used, deliver, lease, sell, or otherwise dispose of the Satellite, the GBBF Network, the Space Segment or other Deliverable Item of hardware or any part thereof. Without limitation, the Parties agree that Purchaser shall be entitled to sublicense to any purchaser or lessee of a Deliverable Item from Purchaser a sublicense of the license and rights under Article 32.2.1 to use, reproduce and modify Contractor Space Segment Intellectual Property incorporated into a Deliverable Item for the sole and exclusive purpose of testing, operating, repairing and/or maintaining such Deliverable Item.

32.6 [*]

ARTICLE 33- PUBLIC RELEASE OF INFORMATION

33.1 Generally.

Each Party shall obtain the prior written approval of the other Party, which approval shall not be unreasonably withheld or delayed, concerning the content and timing of news releases, articles, brochures, advertisements, prepared speeches and other information releases concerning the Work performed or to be performed hereunder, within a reasonable time prior to the release of such information, except to the extent disclosure shall be required by law, which shall be subject to Article 31.4.

[*] CONFIDENTIAL TREATMENT REQUESTED BY ICO GLOBAL COMMUNICATIONS (HOLDINGS) LIMITED.

ARTICLE 34- NOTICES

34.1 Written Notification.

Each notice or correspondence required or permitted to be given or made hereunder shall be in writing (except where oral notice is specifically authorized) to the respective addresses, facsimile and telephone numbers and to the attention of the individuals set forth below, and any such notice or correspondence shall be deemed given on the earlier to occur of (i) actual receipt, irrespective of whether sent by post, facsimile transmission (followed by mailing of the original copy), overnight courier or other method, and (ii) seven (7) days after mailing by registered or certified mail, return receipt requested, postage

In the case of Purchaser:

ICO Satellite Management LLC 222 N. Sepulveda Boulevard Suite 1770 El Segundo, CA 98033 Attn: Contract Manager Telephone: [*] Facsimile: [*]

With a copy to:

Davis Wright Tremaine, LLP 1501 4th Avenue, Suite 2600 Seattle, WA 98101 Attn: [*] Facsimile: (206) 628-7699

[*] CONFIDENTIAL TREATMENT REQUESTED BY ICO GLOBAL COMMUNICATIONS (HOLDINGS) LIMITED.

In the case of Contractor:

Space Systems/Loral, Inc. 3825 Fabian Way, M/S G-44 Palo Alto, CA 94303-4697

[*]
Telephone No.:
Facsimile No.:

With a separately delivered copy to: Space Systems/Loral, Inc. 3825 Fabian Way, M/S G-40 Palo Alto, CA 94303-4697

[*]
Telephone No.:
Facsimile No.:

34.2 Change of Address.

Either Party may from time to time change its notice address or the persons to be notified by giving the other Party written notice (as provided above) of such new information and the date upon which such change shall become effective.

ARTICLE 35- RISK MANAGEMENT SERVICES

35.1 <u>Purchaser Responsibility.</u>

Purchaser may, at its election, procure a Launch and in-orbit insurance policy covering the risks of Launch and in-orbit failures with respect to partial loss, total loss, or constructive total loss of such Satellite ("Launch and In-Orbit Insurance Policy").

IN NO EVENT SHALL PURCHASER DISCLOSE OR TRANSFER CONTRACTOR-PROVIDED TECHNICAL INFORMATION OR PROVIDE TECHNICAL SERVICES BASED ON CONTRACTOR-FURNISHED TECHNICAL INFORMATION TO NON-U.S. INSURANCE BROKERS OR UNDERWRITERS OR OTHER NON-U.S. PERSONS OR ENTITIES (AS DEFINED IN 22 CFR SECTION 120.15 AND SECTION 120.16, AS AMENDED) WITHOUT CONTRACTOR'S PRIOR WRITTEN APPROVAL AND, WHERE REQUIRED, PRIOR APPROVAL OF THE U.S. GOVERNMENT.

35.2 Contractor Support.

35.2.1 Insurance Procurement. Contractor shall provide customary and normal support to assist Purchaser in obtaining a Launch and In-Orbit insurance Policy consisting of: (i) providing a comprehensive presentation package on the Satellite, suitable for presentation to the space insurance brokers and underwriters, (ii) supporting Purchaser with all necessary presentations (oral, written or otherwise), including attendance and participation in such presentations where requested by Purchaser, (iii) providing on a timely basis all reasonable and appropriate technical information, data and documentation, and (iv) providing documentation and

[*] CONFIDENTIAL TREATMENT REQUESTED BY ICO GLOBAL COMMUNICATIONS (HOLDINGS) LIMITED.

35.2.1 Claims Support. Contractor shall cooperate with and provide reasonable support to Purchaser making and perfecting claims for insurance recovery and as to any legal proceeding as may be brought by Purchaser associated with any claim for insurance recovery. Such support shall consist of (i) the support described in Article 35.2.1, (ii) providing on-site inspections as required by Purchaser's insurers and underwriters, (iii) participating in review sessions with a competent representative selected by the insurers and underwriters to discuss any continuing issue relating to such occurrence, including information conveyed to either Party, (iv) using its reasonable efforts to secure access for the insurers and underwriters to all information used in or resulting from any investigation or review of the cause or effects of such occurrence, (v) making available for inspection and copying all information necessary to establish the basis of a claim, and (vi) supporting Purchaser in establishing the basis of a total loss, constructive total loss and partial loss as those terms are defined in the Launch and In-Orbit Insurance Policy and provided Purchaser furnishes Contractor with the definitions of such terms. Notwithstanding Contractor's specifying such basis, Purchaser shall make the final determination of whether a partial loss, constructive total loss or total loss under its Launch and In-Orbit Insurance Policy has occurred. For all claims that may arise on or before Acceptance of the Space Segment as provided in Article 10 hereof, such Contractor support shall be provided within the overall price of this Contract; thereafter, Contractor shall be entitled to reimbursement of Actual Costs reasonably incurred in connection with the provision of such support plus a mark-up of [1], to be invoiced and paid in accordance with Article 5.

ARTICLE 36- ORDER OF PRECEDENCE

In the event of conflict among the terms of the Terms and Conditions (Preamble, Recitals, Articles 1 to 37 and Attachments A through D) of this Contract and the Exhibits, the following order of decreasing precedence shall apply:

Terms and Conditions

(Preamble, Recitals, Articles 1 through 37 and Attachments A through D)

Exhibit F

Payment Plan and Termination Liability Amounts

Exhibit A

Satellite Statement of Work

Exhibit B

Satellite Performance Specification

Exhibit G

Space Segment Statement of Work

Exhibit E

Space Segment Specification

Exhibit C

Product Assurance Program Plan

Exhibit D

Satellite Test Plan

Exhibit H

Space Segment Test Plan

Exhibit J

Ico Dss Sow and Functional Requirements

162

ARTICLE 37- GENERAL

37.1 Assignment.

- 37.1.1 <u>General</u>. Except as set forth herein, this Contract may not be assigned, either in whole or in part or by operation of law, by either Party without the express written approval of the other Party. The non-assigning Party shall provide its approval if in its reasonable judgment its rights under this Contract are not and would not be adversely affected thereby.
- 37.1.2 By Purchaser. Notwithstanding the foregoing, subject to applicable export control restrictions, Purchaser may assign or transfer this Contract or all its rights, duties, or obligations hereunder to (i) any Person provided that such Person has sufficient financial resources to fulfill Purchaser's obligations under this Contract, and the net worth of such Person is not less than the net worth of Purchaser immediately prior to such transfer, or (ii) any or all Financing Entities in connection with obtaining financing for the payment of Contractor's invoices and any and all other fees, charges or expenses payable under this Contract under any Financing Agreement: (iii) any Affiliate of Purchaser, or (iv) any Person who acquires (whether through acquisition, merger or other similar transaction), all or substantially all of the assets of Purchaser, provided in any case the assignee, transferee, or successor to Purchaser has expressly assumed in writing all the obligations of Purchaser and all terms and conditions applicable to Purchaser under this Contract.
- 37.1.3 By Contractor. Notwithstanding the foregoing, Contractor may assign or transfer this Contract or all of its rights, duties, or obligations hereunder (i) to any Affiliate of Contractor, or (ii) in connection with a sale, transfer, merger, or other reorganization affecting Contractor or its assets, provided in each such case pursuant to this Article 37.1.3: (a) such assignment or transfer does not have a material adverse effect on the assignee's, transferee's or successor to Contractor's technical ability to fulfill its

obligations under this Contract; (b) the net worth of the assignee, transferee or successor to Contractor is not less than the net worth of Contractor immediately prior to such assignment or transfer; (c) the assignee, transferee or successor to Contractor has expressly assumed in writing all the obligations of Contractor and all terms and conditions applicable to Contractor under this Contract; and (d) neither the assignee, transferee or successor to Contractor nor any Affiliate thereof is a competitor of Purchaser engaged or engaging in the business of satellite based delivery of mobile telephony.

37.1.4 <u>Security Interests</u>. Either Party, upon prior written notice to the other Party, may grant security interests in its rights hereunder to lenders that provide financing for the performance by such Party of its obligations under this Contract or for the subject matter hereof.

37.2 Binding Effect.

This Contract shall be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns. Assignment of this Contract shall not relieve the assigning Party of any of its obligations nor confer upon the assigning Party any rights except as provided in this Contract.

37.3 Severability.

If any provision of this Contract is declared or found to be illegal, unenforceable or void, the Parties shall negotiate in good faith to agree upon a substitute provision that is legal and enforceable and is as nearly as possible consistent with the intentions underlying the original provision. If the remainder of this Contract is not materially affected by such declaration or finding and is capable of substantial performance, then the remainder shall be enforced to the extent permitted by law.

37.4 Waiver of Breach of Contract.

A waiver of any provision or any breach of a provision of this Contract shall not be binding upon either Party unless the waiver is in writing, signed by a duly authorized representative of the Party to be bound, as applicable, and such waiver shall not affect the rights of the Party not in breach with respect to any other or future breach. No course of conduct by a Party shall constitute a waiver of any provision or any breach of a provision of this Contract unless a written waiver is executed in accordance with the provisions of this Article 37.4.

37.5 Amendments.

This Contract, including any and all its Schedules, Attachments, Annexes, Exhibits and Appendices, may not be modified except by written instrument of subsequent date signed by an officer of Contractor, or another person designated in writing by any such officer to sign such an instrument and a senior vice president of Purchaser, or another person designated in writing by any such Purchaser senior vice president to sign such an instrument.

37.6 Captions.

The captions contained herein are for purposes of convenience only and shall not affect the construction of this Contract.

37.7 Relationships of the Parties.

It is expressly understood that Contractor and Purchaser intend by this Contract to establish the relationship of independent contractors only, and do not intend to undertake the relationship of principal and agent or to create a joint venture or partnership or any other relationship, other than that of independent contractors, between them or their respective successors in interests. Neither Contractor nor Purchaser shall have any authority to create or assume, in the name or on behalf of the other Party, any obligation, expressed or implied, or to act or purport to act as the agent or the legally empowered representative of the other Party, for any purpose whatsoever.

37.8 Entire Agreement.

This Contract, including all its Schedules, Attachments, Annexes, Exhibits and Appendices, represents the entire understanding and agreement between the Parties hereto with respect to the subject matter hereof, and supersedes all prior negotiations and agreements with respect to the subject matter hereof.

37.9 Standard of Conduct.

Both Parties agree that all their actions in carrying out the provisions of this Contract shall be in compliance with applicable laws and regulations and neither Party will pay or accept bribes, kickbacks or other illegal payments, or engage in unlawful conduct.

37.10 Construction.

This Contract, including all its Schedules, Attachments, Annexes, Exhibits and the Appendices have been drafted jointly by the Parties and in the event of any ambiguities in the language hereof, there shall be no inference drawn in favor of or against either Party.

37.11 Counterparts.

This Contract may be signed in any number of counterparts with the same effect as if the signature(s) on each counterpart were upon the same instrument.

37.12 Applicable Law.

This Contract shall be interpreted, construed and governed, and the rights of the Parties shall be determined, in all respects, according to the laws of the State of California, without reference to its conflicts of laws rules.

37.13 Survival.

Termination or expiration of this Contract for any reason shall not release either Party from any liabilities or obligations set forth in this Contract that (i) the Parties have expressly agreed shall survive any such termination or expiration or (ii) remain to be performed or by their nature would be intended to be applicable following any such termination or expiration. Without limitation, the following provisions shall survive termination or expiration of the Contract: 19, 20, 30, 31, 32 and 37.

37.14 U.N. Convention on the International Sales of Goods.

The U.N. Convention on the International Sales of Goods shall not apply or otherwise have any legal effect with respect to this Contract.

37.15 No Third-Party Beneficiaries.

This Contract is entered into solely between, and may be enforced only by, Purchaser and Contractor and their permitted assigns, and this Contract shall not be deemed to create any rights in third parties, including suppliers, customers and owners of a Party, or to create any obligations of a Party to any such third parties.

37.16 Consents and Approvals.

Except where expressly provided as being in the sole discretion of a Party, where agreement, approval, acceptance, consent, or similar action by either Party is required under this Contract, such action shall not be unreasonably delayed or withheld. An approval or consent given by a Party under this Contract shall not relieve the other Party from responsibility for complying with the requirements of this Contract, nor shall it be construed as a waiver of any rights under this Contract, except as and to the extent otherwise expressly provided in such approval or consent.

[*] CONFIDENTIAL TREATMENT REQUESTED BY ICO GLOBAL COMMUNICATIONS (HOLDINGS) LIMITED.

37.17 Mitigation and Good Faith.

Each Party shall have a duty to mitigate damages and to act in good faith.

37.18 Record Retention and Audit Rights.

Contractor and its subcontractors shall maintain books, records, documents and other evidence which sufficiently and properly reflects the accuracy of the costs assessed or charged directly or indirectly to Purchaser. In the event that Contractor's costs shall be subject to bona fide inquiry (for example, under Articles 16, 21 and 23 hereof), Purchaser shall have the right to audit Contractor's records provided that: (a) the scope of the audit shall be limited to matters relevant to the bona fide inquiry; (b) the audit shall be performed by a third party auditor reasonably satisfactory to Contractor; (c) all costs of the audit shall be borne by Purchaser; and (d) Purchaser shall not be provided with confidential information (such as labor rates), which in any report to Purchaser shall be excluded or otherwise redacted.

37.19 <u>Time is of the Essence</u>.

Subject to the applicable remedies and procedures set forth in this Contract, the Parties agree that time is of the essence in the performance of this Contract.

37.20

37.21 Representations.

Each Party hereby represents and warrants to the other:

- (a) Such Party is an entity duly formed, validly existing and in good standing in its jurisdiction of formation, with full power and authority to conduct its business as it is now being conducted, to own or use the properties and assets that it purports to own or use, and to perform all its obligations under Contract.
 - (b) This Contract constitutes the legal, valid, and binding obligation of such Party, enforceable against it in accordance with its terms.
 - (c) This Contract has been duly executed and delivered by such Party, and
- (d) Neither the execution and delivery of this Contract nor the performance by the Party of its obligations hereunder: (i) contravene, conflict with, or result in a violation of (A) any provision of the organizational documents of the Party, (B) any resolution adopted by the board of directors or the stockholders or its equivalent of a Party; or (C) any contract or obligation of the Party; or (ii) require, other than in the case of licenses and permits referred to in Article 7, the consent of approval of any Person including, in the case of Contractor, the approval of the bankruptcy court.

IN WITNESS WHEREOF, the Parties have executed this Contract by their duly authorized officers as of the date set forth in the Preamble.

Space Systems/Loral, Inc.

ICO Satellite Management LLC

By:

By:

Name: Ron Haley

Name: Dennis Schmitt

Title: CFO

Title: Director/CFO

ATTACHMENT A

FORM OF INVOICE

[Date]]

ICO Satellite Management LLC 2300 Carillon Point Kirkland, Washington, 98033

Attention:

RE:

Terms and conditions of the Amended and Restated Space Segment Contract, dated as of (as amended, supplemented or modified from time to time, the "ICO Satellite Contract"), between ICO SATELLITE MANAGEMENT LLC ("PURCHASER" OR "ICO") and SPACE SYSTEMS/LORAL, INC. ("CONTRACTOR")

Ladies and Gentlemen:

This Invoice is delivered to ICO pursuant to Article 5 of the ICO Space Segment Contract and constitutes Contractor's request for payment in the amount of \$\frac{1}{2}\$ for Milestone No. / Time Payment No.

Very truly yours,		
SPACE SYSTEMS/LORAL, INC.		
By: Title:	**************************************	

Annex I to Attachment A

Form of Contractor Certificate

Reference:

Milestone No.

/Time Payment No.

[Date]

ICO Satellite Management LLC 2300 Carillon Point Kirkland, Washington, 98033

Attention: Treasurer

RE:

Terms and conditions of the Amended and Restated Space Segment Contract, dated as of MANAGEMENT LLC ("PURCHASER" OR "ICO") and SPACE SYSTEMS/LORAL, INC. ("CONTRACTOR") (as amended, supplemented or modified from time to time, the "ICO Space Segment Contract")

Gentlemen:

This Certificate is delivered to you pursuant to Article 5 of the terms and conditions of the ICO Space Segment Contract. Each capitalized term used herein and not otherwise defined shall have the meaning assigned thereto in the terms and conditions of the ICO Space Segment Contract.

We hereby certify, based on information and belief, that, as of the date hereof:

- The ICO Space Segment Contract is in full force and effect and except as set forth in Schedule I hereto, has not been amended, supplemented or otherwise modified
- Except as set forth in Schedule I hereto, no event or condition is known to exist that permits or requires us to cancel, suspend, or terminate our performance under the ICO Space Segment Contract or that could excuse us from liability for non-performance thereunder.
- Except with respect to amounts that are the subject of a dispute or are overdue (such overdue amounts and such disputed amounts, if any, being described in reasonable detail in an attachment hereto), all amounts due and owing to us have been paid in full through the date of the immediately preceding Contractor Certificate or are not overdue.

The amount referred to in paragra	da (a) ak	ove was c	omputed i	in accor	dance v	with the term	ns of the I	CO Spac	e Segment	Contract.			
The Milestone to which Milestone	No.	relates h	as been c	omplet	ed in ac	cordance w	ith the IC(Space !	Segment Co	ontract.			
Very truly yours,		**,						,			-		
SPACE SYSTEMS/LORAL, I	NC.							٠ .					
By: Title:				-									
						A-I-2							

di.

SCHEDULE I TO ANNEX I TO ATTACHMENT A

List	of	Exce	ptions:

Amendments to ICO Space Segment Contract:

Exceptions Affecting Final Acceptance Date:

Exceptions Affecting Contractor's Performance:

A-I-1

[*] CONFIDENTIAL TREATMENT REQUESTED BY ICO GLOBAL COMMUNICATIONS (HOLDINGS) LIMITED. ATTACHMENT B KEY PERSONNEL

Key Personnel Position		Individual
Program Manager	transfer to the second section of the second section of the second section of the second section of the second	and the second control of the second
Spacecraft Engineering Ma	nager [*]	
Payload Program Manager		
Assembly, Integration and Space Segment Manager	lesting Manager	No college of the second second second second
Space Segment Technical L	ead [*]	
	A-I-2	

[*] CONFIDENTIAL TREATMENT REQUESTED BY ICO GLOBAL COMMUNICATIONS (HOLDINGS) LIMITED. ATTACHMENT C

GBBF – UNIQUE COMPONENTS AND FUNCTIONS ON BOARD THE SATELLITE

[*]

[*] CONFIDENTIAL TREATMENT REQUESTED BY ICO GLOBAL COMMUNICATIONS (HOLDINGS) LIMITED.

ATTACHMENT E

MAJOR SUBCONTRACTS LIST

[*]

Item Subcontractor

Suppliers on this list approved for use as Major Subcontractors or as replacement Subcontractors for the ICO Program for the respective items contained in this list.

A-I-8

EXHIBIT A

ICO 2 GHz GEO SATELLITE STATEMENT OF WORK

Confidential Treatment of This Entire Exhibit Has Been Requested, and the Text Has Been Submitted to the Commission.

EXHIBIT B

ICO 2-GHz GEO SATELLITE PERFORMANCE SPECIFICATION

Confidential treatment of this entire Exhibit has been requested, and the text has been submitted to the Commission.

EXHIBIT C COMMERCIAL PROGRAMS PRODUCT ASSURANCE PLAN

Entire Exhibit Redacted pursuant to U.S. International Traffic in Arms Regulation (ITAR), 22 CFR 120-130

EXHIBIT D

ICO 2-GHz GEO SATELLITE TEST PLAN

Entire Exhibit Redacted pursuant to U.S. International Traffic in Arms Regulation (ITAR), 22 CFR 120-130



EXHIBIT E

ICO SPACE SEGMENT PERFORMANCE SPECIFICATION

Confidential Treatment of This Entire Exhibit Has Been Requested, and the Text Has Been Submitted to the Commission

EXHIBIT F

ICO SATELLITE PAYMENT PLAN AND TERMINATION LIABILITY AMOUNTS

[*]

Prepared for: ICO Satellite Management LLC 2300 Carillon Point Kirkland, Washington 98033

ICO Satellite Management LLC Approval Signature & Date

Prepared by: Space Systems/Loral 3825 Fabian Way Palo Alto, California 94303-4201

Space Systems/Loral Approval Signature & Date

This document contains data and information proprietary to Space Systems/Lord, Inc., and ICO Satellite Management LLC. This data is being furnished pursuant to the provisions of the ICO Satellite Purchase Contract. The ICO Satellite Management LLC customer shall have the right to duplicate, use or disclose the data to the extent specified in the Purchase Contract.

Exhibit F: Payment Plan and Termination Liability Amounts ICO 2 GHz GEO Satellite Payment Plan

Confidential Treatment of This Entire Exhibit Has Been Requested, and the Text Has Been Submitted to the Commission

EXHIBIT G

ICO Space Segment STATEMENT OF WORK

Confidential Treatment of This Entire Exhibit Has Been Requested, and the Text Has Been Submitted to the Commission

EXHIBIT H

ICO Space Segment Test Plan

Entire Exhibit Redacted pursuant to U.S. International Traffic in Arms Regulation (ITAR), 22 CFR 120-130

EXHIBIT J

ICO DSS SOW and FUNCTIONAL REQUIREMENTS

Entire Exhibit Redacted pursuant to U.S. International Traffic in Arms Regulation (ITAR), 22 CFR 120-130

CERTIFICATE OF SERVICE

I, Jeffrey A. Marks, hereby certify that on this 18th day of December, 2006, I caused to be served a true copy of the foregoing by first class mail, postage prepaid, upon the following:

Cheryl A. Tritt Morrison & Foerster LLP 2000 Pennsylvania Ave., NW Suite 5500 Washington DC 20006 (202) 887-1500 Suzanne Hutchings Malloy ICO Global Communications 815 Connecticut Avenue, NW Suite 610 Washington, DC 20006 (202) 330-4005

Je frey A. Marks